

## Neural Networks for Natural Language Processing Mohamed Morchid

## ▶ To cite this version:

Mohamed Morchid. Neural Networks for Natural Language Processing. Artificial Intelligence [cs.AI]. Avignon Université, 2019. tel-02398814v1

## HAL Id: tel-02398814 https://theses.hal.science/tel-02398814v1

Submitted on 8 Dec 2019 (v1), last revised 5 Jan 2020 (v2)

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



ACADÉMIE D'AIX-MARSEILLE AVIGNON UNIVERSITÉ

## HDR

Defended at Avignon University to obtain the Habilitation À Diriger des Recherches diploma

**DOMAIN:** Computer Science

Laboratoire Informatique d'Avignon (EA 4128)

## Neural Networks for Natural Language Processing

by Mohamed Morchid

## Publicly defended the November 26<sup>th</sup> 2019 front of the jury composed with:

Mrs.	Dilek Z. HAKKANI-TÜR	Senior Principal Scientist, Alexa AI, USA	Reviewer
Mr.	Patrice BELLOT	Full Professor, AMU Polytech', LIS, Marseille	Reviewer
Mr.	Frédéric ALEXANDRE	First class Research Director, INRIA, Bordeaux	Reviewer
Mr.	Yannick ESTÈVE	Full Professor, AU, LIA, Avignon	Examiner
Mr.	Frédéric BÉCHET	Full Professor, AMU, LIS, Marseille	Examiner



Laboratoire Informatique d'Avignon



ACADÉMIE D'AIX-MARSEILLE AVIGNON UNIVERSITÉ

# HABILITATION À DIRIGER DES RECHERCHES

présentée à l'Avignon Université pour obtenir le diplôme d'Habilitation À Diriger des Recherches

**SPÉCIALITÉ : Informatique** 

Laboratoire Informatique d'Avignon (EA 4128)

## *Réseaux de Neurones pour le Traitement Automatique du Langage*

par Mohamed Morchid

## Soutenue publiquement le 26 Novembre 2019 devant un jury composé de :

Dilek Z. HAKKANI-TÜR	Senior Principal Scientist, Alexa AI, USA	Rapporteur
Patrice BELLOT	Professeur, AMU Polytech', LIS, Marseille	Rapporteur
Frédéric ALEXANDRE	Directeur de Recherche INRIA, Bordeaux	Rapporteur
Yannick ESTÈVE	Professeur, AU, LIA, Avignon	Examinateur
Frédéric BÉCHET	Professeur, AMU, LIS, Marseille	Examinateur
]	Dilek Z. HAKKANI-TÜR Patrice BELLOT Frédéric ALEXANDRE Yannick ESTÈVE Frédéric BÉCHET	Dilek Z. HAKKANI-TÜRSenior Principal Scientist, Alexa AI, USAPatrice BELLOTProfesseur, AMU Polytech', LIS, MarseilleFrédéric ALEXANDREDirecteur de Recherche INRIA, BordeauxYannick ESTÈVEProfesseur, AU, LIA, AvignonFrédéric BÉCHETProfesseur, AMU, LIS, Marseille



Laboratoire Informatique d'Avignon

To my family and Claire and in memory of my beloved father, Belfakih Morchid ...

... you will be eternally present here and everywhere, now and always ...

## Summary

This dissertation presents my recents efforts on both neural networks based methods and algorithms, and their applications on real-world tasks for Natural Language Processing (NLP). This document is organized following my different points of interest from hitherto proposed real-valued neural networks applied to challenging NLP related tasks, to novel complex-valued neural based architectures that seek to address the recent issues related to the learning process. The document presents my main achievements and is organized as follow:

Real-valued Neural Networks. Nowadays, NLP tasks employ massively Neural Networks (NN) based algorithms to solve challenging real-world tasks and the document:
introduces novel NNs for considering multiple input streams of information during the learning process to better predict the desired output,

• studies the impact of noisy documents from an Automatic Speech Recognition (ASR) system on an encoder-decoder that extracts robust features from spoken documents.

**Quaternion-valued Neural Networks.** Real-valued NN hardly handle inter- and intra-dependencies between input features during the learning. We propose novel Quaternion-valued neural networks (QNN) that consider inter-dependencies between the input features as well as the dependencies between basic elements composing a given feature. In this document we:

• present a Quaternion-valued Convolutional Neural Network (QCNN) that consider alongside the convolution process (inter-dependencies) and the latent relations between basic elements of each feature (inter-dependencies),

• introduce the Quaternion-valued Recurrent Neural Network (QRNN) to encode multi-dimensional input sequences from speech signal.

**Ongoing works and futur directions.** Real-valued and Quaternion-valued neural networks are difficult to learn since the amount of data required for learning is potentially large, even huge; moreover, these models are also employed to generate unseen information such as images or documents. Among these models, generative adversarial networks (GAN) have encountered a large success in different natural language processing (NLP) related tasks. Nonetheless, these models are unstable during the learning process. This part:

• introduces novel parsimonious neural networks that better handle large corpora and consider the long-term dependencies,

• stuies the impact of manual transcripts on automatically transcripts of spoken documents during a NLP task with GAN mapping approach.

# Contents

## Introduction

Ι	Rea	l-valued Neural Networks for Natural Language Processing	25
1	Para	allel Recurrent Neural Networks	27
	1.1	Introduction	28
	1.2	Parallel Long Short-Term Memory (PLSTM)	29
	1.3	Experiments	32
	1.4	Results and Discussion	33
2	Enc	oder-Decoder Neural Networks	37
	2.1	Introduction	38
	2.2	Autoencoder neural networks based systems	39
	2.3	Experiments	46
		2.3.1 DECODA framework	46
		2.3.2 Automatic Speech Recognition System	47
	2.4	Results and Discussion	48
	Qi	laternion Neural Networks for Natural Language Processing	5.
3	Qua	Iternion Convolutional Neural Networks	5
	3.1		54
	3.2	Motivations	55
	3.3	Quaternion algebra	5
	3.4	Quaternion Convolutional Neural Networks	52
	3.5	Experiments	58
	3.6	Results and Discussion	60
4	Qua	ternion Recurrent Neural Networks	63
	4.1	Introduction	64
	4.2	Quaternion Recurrent Neural Networks	64
	4.3	Experiments	64
	4.4	Results and Discussion	6

11

III	0	ngoin	g Research, Future Directions & General Perspectives	69
5	<b>Pars</b> 5.1 5.2 5.3 5.4	imonic Introd Parsin Exper Result 5.4.1 5.4.2 5.4.3	ous Neural Networks         uction         nonious Memory Unit (PMU)         iments         is and Discussion         Gates activity of GRU and PMU         Short-term dependencies from spoken dialogues         Long-term dependencies from 20-Newsgroups documents	71 72 76 76 76 76 79 80
6	Gen 6.1 6.2 6.3 6.4	erative Introd Gener Exper Result	Adversarial Networks uction	<b>83</b> 84 84 86 88
7	Gen	eral Pe	rspectives	91
IV		ppend	ix	93
Lis	st of i	i <b>llustra</b>	tions	103
Lis	st of (	tables		105
Bi	bliog	raphy		107
cι	J <b>RRI</b>	CULUI	M VITÆ (In French)	123

# Introduction

## Context

#### Few words on my research interests

My initial researches have been mostly based on topic models (Blei et al., 2003) applied to Natural Language Processing (NLP) tasks. Among these high-level representations, the latent Dirichlet allocation (LDA) (Blei et al., 2003) based model allows NLP systems to represent relevant content of documents in a abstract latent topics model. We have employed LDA representations to extract robust features from noisy spoken dialogues. After my Ph.D. thesis, we have been interested to evaluate the benefit of neural networks initialization with latent based features from an LDA model (Morchid et al., 2015b). It was my first investigation of neural networks for real-world NLP applications. Since 2015, my efforts have been focused on applications of hitherto proposed neural networks for real-life problems such as theme identification of spoken dialogues and on complex-valued neural networks for different pattern recognition related tasks. These interests are at the edge of the Natural Language Processing (NLP) and machine learning (ML) domains composing an important part of the Artificial Intelligence (AI) broad spectrum. Indeed, AI covers a large area of NLP throughout different real-world applications. Among AI based paradigms, connectionist models or Parallel Distributed Processing (PDP) (McClelland et al., 1986; Rumelhart et al., 1986) have been introduced in a large number of domains, ranging from different aspects of language processing to cognitive control from perception to memory. Connectionist models take inspiration from the manner in which information processing occurs in the brain. Processing involves the propagation of activation among simple units (artificial neurons) organized in networks, that is, linked to each other through weighted connections representing synapses or groups thereof. Each unit then transmits its activation level to other units in the network by means of its connections to those units. The activation function describes how each unit computes its activation based on its inputs, may be a simple linear function, but is more typically non-linear (for instance, a sigmoid function). These neural networks have encountered a wide success due to their capabilities to code complex abstract structures and patterns in latent sub-spaces. In the context of NLP applications, these neural networks models such as the Multi Layer Perceptron (MLP) have to process large even huge amount of data due to the rapid growing of sharing platforms on Internet as well as the number of available data-sets. Moreover, the datas available are even more composed with different signals (speech segments) or dimensions (images

or multi-channels sequences). Therefore, the models and paradigms proposed to process large amount of multidimensional data have to both be efficient to process quickly this large amount of information and consider multiple dimensions of the document content. Since our work are mostly focused on neural networks based models with applications on NLP, in the last years, we have been interested on other specific applications such as image processing (Parcollet et al., 2019). Nonetheless the last results are still too recent to be mentioned in this document. This thesis reports part of the work done in collaboration with Ph.D. students and colleagues on field of deep learning. In regards to the large number of neural based models and paradigms this chapter focuses on related works and the reader can refer to papers from main AI related conferences and journals for further details.

#### Brief history of neural networks with some key dates

Figure 1 presents some main steps throughout the last decades of the neural networks history. The contemporary period started with the advent of GPU cards at 1999 from Nvidea and the CUDA librairies in 2007 to develop the algorithms. These cards allows neural networks based models to drastically reduce the processing time required for learning by 70 (Raina et al., 2009). With the large number of academics and industrials that then employ CUDA to develop machine learning systems, NVidea have released a dedicated set of librairies called cuDNN in 2014.



Figure 1: Neural Networks in some of the main dates.

Deep learning is originated from the availability of large-scale training data-sets and the increased computation power from GPU cards and groups feature learning techniques exploiting artificial neural networks (LeCun et al., 2015) among other algorithms. Human being also conducts learning via neural networks based on biological neural networks inside the brain. Neuroscientists have studied how information is processed in human being brain in order to understand it capability of processing information effectively. Neural networks based models and paradigms are able to process data and natural information much better than human being. Deep learning becomes an international strategic issue due to their impressive performances on different domains such as human communications, medicine, law, political analytics or military <sup>1 2</sup> with unlimited uses. France will be part of these general leaders with recent political efforts <sup>3 4</sup>. Among AI models, neural networks have encountered a wide success in different real-world applications. These neural networks are composed with basic artificial neurones. This document is focusing on neural networks and their application on different NLP related tasks.

#### Neural Networks

The first definition of artificial neurons or formal neurons has been introduced in (Mc-Culloch et Pitts, 1943). In this definition a neuron has a variable number of weighted inputs simulating the dendrites of biological neurons. All inputs are summed and an activation function is applied to this sum to produce an output value. The first application of the formal neuron is the perceptron (Rosenblatt, 1958) composed of a single neuron and a heavy-side activation function and allows a binary classification. In addition to the neuron a supervised learning method of perceptron is introduced to estimate its weight representing an hyperplane by an iterative process. This hyperplane is optimized to separate two classes. On the other hand, a perceptron is unable to solve a problem that is not linearly separable. This weakness was highlighted by the problem of "Xor": Perceptron cannot model the "exclusive or" function. This limitation has slowed down the development of artificial neural networks for several decades. (Werbos, 1974) propose the gradient backpropagation method for learning artificial neural networks. Since 1982, artificial neural networks are able to extract descriptors through unsupervised learning (Kohonen, 1982). In these networks, the neurons are arranged in a grid. Each area of this neural grid learns to respond to a particular type of data. Once trained, an adaptive map can be used to either discretize or perform a vector quantization or to reduce the dimensionality of input data (Chihi, 1998; Nasrabadi et Feng, 1988). Artificial neural networks whose interneuronal connections form an acyclic graph are called "feedforward neural networks". The perceptron is an example of a feedforward

<sup>&</sup>lt;sup>1</sup>https://www.forbes.com/sites/stratfor/2018/02/07/the-coming-tech-war-withchina/#3a1b6a641cd4

<sup>&</sup>lt;sup>2</sup>Russia's leader, Vladimir Poutine, have claimed that the nations that will develop effective AI models and applications will lead the world: https://www.cnbc.com/2017/09/04/putin-leader-in-artificialintelligence-will-rule-world.html

<sup>&</sup>lt;sup>3</sup>https://www.economie.gouv.fr/France-IA-intelligence-artificielle

<sup>&</sup>lt;sup>4</sup>https://www.forbes.com/sites/forbestechcouncil/2017/12/05/these-seven-countries-are-in-a-race-to-rule-the-world-with-ai/

neural network.

Multilayer Perceptrons (MLPs) (Rumelhart et al., 1985) are supervised networks that can be used for classification, regression or dimensionality reduction. In this architecture the neurons are grouped in layers. There are an input layer that receives the information that the network have to process and an output layer that provides the neural network output prediction. The activation function of the output layer can be chosen according to the task that the network has to perform: the linear activation function is used to learn linear regression models; the sigmoid activation function is employed for logistic regressions or binary classifications; the softmax activation function (Bishop et al., 1995) performs multiclass classifications. A cost function calculates the error between the predicted output and the expected output. A variable number (greater than or equal to 1) of hidden layers with a nonlinear activation function is inserted between input and output layers. Each neuron in a layer is connected to all the neurons of the previous and next layers. On the other hand, there are no connections between neurons from the same layer. The weights of these connections between two successive layers are learned by backpropagating the gradient between the predicted and the expected output. More details on different neural networks throughout the next dedicated chapters to improve the readability and the understanding of each model in a suitable context.

## **Research directions**

We relate in this section different research directions based on ongoing works on different neural networks related models for real world NLP applications. Each of the following sections describes the paradigms and the perspectives of each directions. All these researches are different and close in some regards. Indeed, these directions have a common ground composed of both theoretical (neural networks) and practical (NLP real life applications) components.

### Multidimensional models

Available multidimensional data from different mediums are even more large and, in the last decades, scientists from different AI related fields such as language processing or machine learning have given an important efforts to propose efficient algorithms and paradigms to handle large amount of information. Indeed most of the documents available nowadays are composed with different information from different mediums or streams. This is particularly the case for videos that are composed with images, audio and potentially textual content. Sharing platforms such as Youtube <sup>5</sup> collect large even huge amount of videos per days. Indeed this year statistics on Youtube show that 400 hours of video are uploaded to every minute <sup>6</sup> and the number of available videos exceed 7 billion. Different approaches have been studied to process such as multi-streams data (Dai et al., 2004; Babcock et al., 2002) such as the Clustering on Demand (CoD) framework (Dai et al., 2004) or the Data Stream Management System (DSMS) (Babcock

<sup>&</sup>lt;sup>5</sup>www.youtube.com

<sup>&</sup>lt;sup>6</sup>https://www.brandwatch.com/blog/youtube-stats/

et al., 2002). These models reach good results but are mostly based on optimization problems that not retain an internal state to memorize sequences. Artificial Intelligence based algorithms are efficient to learn from sequence's basic element states such as neural networks.

Deep Neural Networks (DNN) allow systems to efficiently process these big data in terms of systems performances and processing time required during the learning process. Nonetheless, spoken and textual documents processed in Natural Language Processing (NLP) systems have to take into account sequences made of the words composing a sentence or a paragraph. The Recurrent Neural Networks (RNN) (Rumelhart et MacClelland, 1986; Pearlmutter, 1989) is a DNN and is well adapted to sequence data for spoken or textual documents processing in NLP-based systems such as statistical parametric speech synthesis (Fernandez et al., 2014; Fan et al., 2014), speech emotion recognition (Chao et al., 2015; Chen et Jin, 2015), facial landmark detection (Chen et al., 2017) and speech recognition (Gajecki, 2014; Graves, 2012; Siniscalchi et al., 2013). RNNs can be applied to almost any problem with sequential structure. Nevertheless RNNs fail to learn long-term dependencies due to the vanishing gradient problem (Bengio et al., 1994) and gates based units have been introduced to address this drawback like Long Short-Term Memory (Hochreiter et Schmidhuber, 1997) or Gated Recurrent units (Chung et al., 2014). These RNN-based models learn the internal state from a sequence of basic elements and process only uni-dimentionnal data.

Convolutional Neural Networks (CNN) (LeCun et al., 1989) can process multidimensional data, such as images (Red, Green and Blue (RGB) pixels) or documents (word/sentence matrix) (Zhang et al., 2015). Nonetheless CNN are limited to 2- or 3-D dimensional feature spaces and the number of available streams for a given information may be larger than 2 or 3. Another drawback of both CNN- and RNN-based models is the manner to compute latent states of multiple parallel inputs during the learning. Indeed recent researches have proposed to learn the latent states in all gates forming the gated RNN (Graves et al., 2007; Graves et Schmidhuber, 2009; Stollenga et al., 2015) and applications such as pixel-wise brain image segmentation (Stollenga et al., 2015). The different inputs corresponding to different streams are added within each gate to code latent dependencies of these streams during the learning process of gates as well as the exposed outputs.

Nonetheless all parts of the model (gates for LSTM/GRU) process all streams to better predict the optimal output while each stream has a particular content. In the case of label prediction of a given stream **s**, all parallel streams do not contain enough information on **s** to propose an efficient candidat as an output of **s**. We have proposed during the Ph.D. thesis of Mohamed Bouaziz to learn from individual streams an efficient process called Parallel Long Short-Term Memory (see chapter 1) to predict the next label of a given stream knowing the other ones with LSTM-RNN models (Bouaziz et al., 2016). This model allows the system to take into account different streams during the learning process. The experiments show that the proposed RNN-based model reach better performances than n-gram models. Nonetheless, all streams are taken into account in the same level for all gates and all streams. The streams can be split throughout the gates based on the goal off each gate <sup>7</sup>. Therefore, an interesting perspective for this work is to find paradigms to better split the information from different streams between gates of RNN based models in regards to the application. In the case of label prediction of a given stream **s** the forget and output gates have to process all streams but the input gate have to be considered only with the stream **s**. Indeed, in this particular case, only the stream **s** can predict an efficient candidat for the next label.

#### Autoencoder neural networks

Encoder-decoder neural networks or Autoencoders are a special architecture of MLP. In regards to the large, even huge, bibliography on neural networks, this section focuses only on autoencoders and their sub models. One can refer to (LeCun et al., 2015; Goodfellow et al., 2016) for more information. The term feedforward refers to a set of neural networks with connections between neurons form an acyclic graph. MLP (Rumelhart et al., 1985) are supervised neural networks that can be used for classification, regression or dimensionality reduction. In this architecture the neurons are grouped in layers with an input layer that receives the information as a set of features that the network process; an hidden layer that represents the input vector in a hidden space; and an output layer that contains the network prediction. The activation function of the output layer can be chosen according to the task that the network has to perform. The linear activation function is used to learn linear regression models. The sigmoid activation function is used for logistic regressions or binary classifications. The Softmax activation function (Bishop et al., 1995) is employed for multi-labels classification related tasks. A cost function computes the error between the output produced by the MLP and the expected output. Each neuron contained in a layer is connected to all the neurons of the next layer with no connections between this neuron and the neurons from the same layer. The weights of these connections are learned by back-propagating the gradient of the error between the predicted output and the expected one throughout the layers. An autoencoder is a MLP with the same features for the input layer and the expected output.

The work presented in Chapter 2 describes the first efforts to process documents on autoencoders to learn the mapping from automatically transcribed spoken documents from an Automatic Speech Recognition (ASR) system to manually transcribed documents. The aim of these studies is to evaluated the potential to neural based systems to learn efficient latent relations between "noisy" documents (from ASR) and "clean" ones (from human transcriptions). These approaches are based on autoencoders neural networks on either the generative, discriminative or both sub-processes. Our work depicted in Chapter 2, reflect my strong interest on latent variables from autoencoders and we plan to follow the study during the next years in regards to the promising results obtained by the proposed autoencoders approaches alongside with the enthusiasm generated by GANs. Chapter 6 reports our first effort on GAN for NLP.

<sup>&</sup>lt;sup>7</sup>the input gate of the LSTM evaluate how much from the hidden state candidate or memory has to be considered; the forget gate retain or forget part of the previous hidden state and the output gate decides the proportion of the new hidden state that has to be exposed to the output

One can refer to (Creswell et al., 2018) for more information on autoencoders.

### Quaternion-valued neural networks

During my Ph.D. thesis, my work has been focused on the extraction process of robust features from latent spaces such as Latent Dirichlet Allocation (LDA) (Blei et al., 2003), from *i*-vectors (Morchid et al., 2015a) or from hyper-complex numbers called Quaternions (Morchid et al., 2013). Quaternions are an extension of complex-numbers and are defined as hypercomplex numbers that contain a real and three separate imaginary components perfectly fitting to three and four dimensional feature vectors such as for image processing, robot kinematics or computer graphics (Sangwine, 1996; Pei et Cheng, 1999; Aspragathos et Dimitros, 1998). Contrary to traditional homogeneous representations, quaternion networks bundle sets of features together. Thereby guaternion numbers allow neural network based models to code latent inter-dependencies between groups of input features during the learning process with fewer parameters than real-valued neural networks by taking advantage of the Hamilton product as the equivalent of the ordinary product but between quaternions. Early applications of quaternion-valued back-propagation algorithms (Arena et al., 1994, 1997) have efficiently solved quaternion functions approximation tasks. More recently neural networks of complex and hypercomplex numbers have received an increasing attention (Hirose et Yoshida, 2012; Tygert et al., 2016; Danihelka et al., 2016; Wisdom et al., 2016) and some efforts have shown promising results in different applications. In particular a deep quaternion network (Parcollet et al., 2016; Titouan et al., 2017; Parcollet et al., 2017), a deep quaternion convolutional network (Chase Gaudet, 2017; Titouan et al., 2018) and a quaternion recurrent neural network (Parcollet et al., 2018c) have been developed and employed by out team from the LIA for different challenging tasks such as images classification, compression and reconstruction or speech recognition.

Motivations to use quaternion neural networks over real-valued ones for quaternionvalued tasks are numerous and well documented (Parcollet et al., 2018c; Isokawa et al., 2009). The major advantage of any QNN is the more efficient and natural representation of the input information compared to traditional real-valued NNs. Indeed a good neural network model has to represent, encode and learn the nature of the input data. Many realistic tasks involve multi-dimensional features with input elements composed with multiple internally related components. These applications are image processing with three channels describing a single pixel or speech recognition with three values (Mel filter banks and deltas) for a unique time frequency, or human motion recognition with outputs and inputs to be represented in the 3D space with 3D coordinates. To succeed in such multidimensional tasks the employed model needs a specific representation and processing of the features. Indeed traditional real-valued NNs consider internal relations at the same level than contextual and global dependencies. As an example, relations between different pixels are considered equivalently to the relation that links the three channels composing each pixel, and it is not clear if both relations are well learned. More precisely, composed entities are processed by traditional real-valued NNs as a bag of independent and smaller

components, while a natural solution is to process such entities as multidimensional and internally related elements, such as with quaternion neural networks (QNNs). Consequently QNNs have been shown to suit particularly well to these representations and it has been demonstrated that they are perfectly able to learn internal relations describing the multidimensional features, with way less neural parameters (Matsui et al., 2004; Parcollet et al., 2018d). Indeed a 4-number quaternion weight linking two 4-number quaternion units only has 4 degrees of freedom, whereas a standard neural net parametrization has  $4 \times 4 = 16$ , i.e., a 4-fold saving in memory.

#### Parsimonious neural networks

Neural architectures are huge and Quaternions allow the models to require less parameters to learn intra- and inter-dependencies between input features but require a 4-dimensional representation to process Hamilton dot product. Most of the NLP related tasks are based on bag-of-words representations, even the use of embeddings are nowadays the usual input representation. Therefore, real-valued neural networks are well suited to process basic document features such as words occurrences or word embeddings. Moreover, real-valued neural networks require a large amount of data to learn latent dependencies in the features space to represent effectively these basic elements throughout enough related contexts. Therefore, the processing time as well as the large number of documents required for learning efficient document representations is closely related too the network architecture; moreover, these representations have to be robust enough to be effective since unseen documents are processed by the NLP systems by learning different hidden states for different related contexts of sequences for both short- and long-term dependencies. Recurrent Neural Networks (RNN) are a powerful set of architectures that code and store internal hidden states of sequences to learn basic element latent dependencies.

Nonetheless, RNN-based models (Zaremba, 2015) require a large set of patterns to code most of the possible contexts of a given event and are time consuming. It is mostly the case since the task is to reveal latent dependencies between documents event (word occurrences) during a natural language processing (NLP) task. Indeed, a given word has to be encountered in different sentences (contexts) to better understand the latent relations between these basic elements. Nevertheless RNNs fail to learn long-term dependencies due to the vanishing gradient problem (Hochreiter, 1998). RNNs with gates such as Long Short-Term Memory (Wu et al., 2017b; Sundermeyer et al., 2012; Greff et al., 2017) (LSTM) avoid the vanishing gradient problem of the RNN (Hochreiter, 1998) along by taking into account both short- and long-term dependencies of the word properly in different contexts but need to process the information in different memory blocks composed with a set of cell-related gates. Even if the results of LSTM-RNNs are promising the processing time required to treat large documents data sets is quite huge due to the different gates activation sub-processes. Indeed, LSTM has to decide from a previous state and a candidat state to process dedicated decision processes via gates to update the memory cell. A recent addition to the RNN set of models called the Gated Recurrent Unit (GRU) has been proposed by (Bahdanau et al., 2014; Wu et al., 2017b) to address this issue and has shown good performances in several tasks such as speech recognition (Graves et al., 2013b), machine translation (Sutskever et al., 2014; Cho et al., 2014). Deep internal structures have been proposed to introduce a depth gate (Yao et al., 2015) to better expose internal hidden relations between memories. GRU is very similar to LSTM in that it uses a combination of gates to adjust exposure from input to the hidden states. It does however not use a memory cell instead opting to fully expose its state to the output thereby also doing away with the output gate. Switching between full retention, mixed and forget mode is implemented using a reset gate *r* and an update gate *z*. This is the main drawback of the GRU-RNN model in that the role and the management of the GRU gates is not based on the relation between short- and long-term dependencies. For example, the reset of the hidden state is provided by both update and reset gates of the GRU. Moreover, the role of the reset gate of the GRU is to avoid the hidden state candidate but the update gate z can remove this part of the hidden state information without the reset gate r. The minimal gated unit (MGU) (Zhou et al., 2016) is composed with a unique gate to update and reset the cell state. They adapt a GRU with a gate that merges r and z gate of the GRU (r = z) onto a unique gate called f. The authors "prefer an RNN architecture that has the smallest number of gates without losing LSTM's accuracy benefits" and "propose the Minimal Gated Unit (MGU), which has the smallest possible number of gates in any gated unit. MGU only has 1 gate, which is the forget gate. MGU is based on GRU, and it further couples the input (reset) gate to the forget (update) gate". The motivation to merge the update and reset gates is not based on an theoretical study of term dependencies. A unique gate f will both reduce the importance of the candidate hidden state and the memory of long-terms dependencies during the processing of the hidden state candidate. This behaviour is close to the leaky unit (Bengio et al., 2013a) that reduces the hidden state candidate influence when the previous hidden states are considered with a single gate *u*. The LU is equivalent to the GRU with a single gate and requires less processing time for learning than the GRU and the LSTM. Nonetheless, the LU codes mostly long-term dependencies but reveals little in the way of short-term dependencies. Other papers have studied single gate units such as the MinimalRNN (Chen, 2017). The MinimalRNN is a GRU with a single gate and differs from LU by processing the candidate hidden state with the input vector  $s_{h,m}$  only  $(r_i(t) = 0$  in the case of the GRU). Therefore, the authors reduce the number of gate but do not study the behaviour of the gate during the learning process.

These efforts to propose efficient neural based architectures requiring even less processing time is crucial. Indeed, since the large number of documents are available with the grow of Internet and the number of applications are mostly employed on mobile devices such as telephones or tablets, the needs on parsimonious models to process this large amount of information are important with less parameters (for mobile devices for example). Therefore, the neural based models have to process these documents effectively with a small number of parameters and with less processing time than hitherto proposed neural networks.

Our work started with quaternion neural networks (see chapter 3 and 4) allow NLP systems to require less neural parameters to process effectively big corpus. We

have also proposed (see chapter 5) the "Parsimonious Memory Units" (PMU) for RNN that differs from GRU with a reduced number of gates by considering first the behaviour of the gates and the role of the activity of the gates in the learning process of the term dependencies in the hidden state. Moreover, the PMU architecture is based on the assumption that short- and long-term dependencies are related. (Yao et al., 2015) present the depth GRU that allows the internal state to consider latent relations between the memories from different layers and differs from our work which is interested in the best way to manage both short- and long-term dependencies during the learning process. We plan to continue the study of memory behaviour within neural process and provide novel neural based architectures to process large data-sets with less parameters alongside with efficient performances during the AISSPER project (ANR 2019) for wich I am the coordinator that will start on January 2020 in collaboration with LIUM and Orkis. The aim of AISSPER is to provide efficient end-to-end neural based models for Spoken Language Understanding (SLU).

### Generative adversarial networks

The obtained results with QNN and autoencoders are promising and generative models will be a main research topic for my future directions. Recently, two promising architectures have been introduced to provide powerful generative models: variational autoencoders (VAE) and Generative Adversarial Networks (GAN). Variational autoencoders are introduced as a link between variational inference (Fox et Roberts, 2012) and deep neural networks (Kingma et Welling, 2013; Gal et Ghahramani, 2015). They consider a set of latent random variables *z*, to capture the variations of the observed variables x. Their joined distribution are defined by:

$$p(x,z) = p(x|z)p(z)$$
(1)

Where the distribution of prior probabilities of p(z) follows a normal distribution and p(x|z) is an observed model whose parameters are calculated by the neural network as a function of z. The non-linear projection of z to x makes it impossible to infer the posterior distribution of p(z|x). For this purpose the VAE uses a variational approximation q(z|x) whose distribution follows a normal distribution. The parameters of this normal distribution (mean and variance) are the output of a non-linear function also learned by the neural network. The generative model p(x|z) and the inference model q(z|x) are learned by backpropagation in order to maximize the lower bound of the likelihood of p(x). They have been used successfully to generate sentences (Bowman et al., 2015), images (Gregor et al., 2015) or to apply rotations on 3D images (Kulkarni et al., 2015).

Generative adversarial networks (Goodfellow et al., 2014) (GAN) combine two neural networks. The first network G is a generative network that generates a vector x from a latent representation from documents of the training corpus. The second network D is a discriminant network that predicts the probability that the input sample is generated by G instead of coming from the training corpus. Both networks optimize their weight to find a balance. They have been used mainly to generate images, from

constraints of shapes or colors (Zhu et al., 2016) or from textual descriptions (Reed et al., 2016).

Nowadays Generative Adversarial Networks (GAN) became one of the dominant approaches for learning latent features from generative models in recent machine learning researches. These models provide a flexible algorithm for learning latent variables describing the data generating processes in which noise is transformed into "clean" data samples on large datasets. A large set of GAN variants are based on autoencoder GANs (AE-GANs). The autoencoders improve GAN training (Nguyen et al., 2017) by combining an autoencoder loss, a GAN loss, and a classification loss defined using a pre-trained classifier. AE-GANs can be broadly classified into three approaches:

- those using an autoencoder as the discriminator, such as energy-based GANs and boundary-equilibrium GANs (Berthelot et al., 2017),
- those using a denoising autoencoder to derive an auxiliary loss for the generator, such as denoising feature matching GANs (Warde-Farley et Bengio, 2016),
- and those combining ideas from VAEs and GANs such as the variational autoencoder GAN (VAE-GAN) (Larsen et al., 2015) that adds an adversarial loss to the variational evidence lower bound objective, the more recent mode-regularized GANs (MRGAN) (Che et al., 2016) or the adversarial generator encoders (AGE) (Ulyanov et al., 2018).

Recently, these generative adversarial networks (GANs) (Goodfellow et al., 2014) received an astonishing interest due to the remarkable results obtained in computer vision (Zhu et al., 2017; Radford et al., 2015; Kim et al., 2017; Berthelot et al., 2017). The ability of GANs to generate samples that are closely-related to targets ones has also been extended to natural language processing (NLP) tasks, such as text and dialogue generation (Rajeswar et al., 2017; Donahue et Rumshisky, 2018; Li et al., 2017), or neural machine translation (Yang et al., 2017; Wu et al., 2017a). To the best of our knowledge, (Wu et al., 2017a) is the most related work to the problem addressed in this paper. Indeed, (Wu et al., 2017a) proposed a model aiming to generate sentences which are hard to be discriminated from human-translated sentences. Consequently, the GAN model is expected to learn a mapping from one language to another one based on human manual translations. GANs became a specific domain in AI and researchers produce a broad spectrum of related models. an conditional version of the GAN has been proposed in (Mirza et Osindero, 2014) to generate unseen elements (not included in the data set) from unlabeled data. (Makhzani et al., 2015) propose a general approach, called an adversarial autoencoder (AAE) that employ hidden states of an autoencoder and random distributions into a generative model for different tasks. The autoencoder is trained with dual objectives: a traditional reconstruction error criterion, and an adversarial training criterion (Goodfellow et al., 2014) that matches the aggregated posterior distribution of the latent representation of the autoencoder to an arbitrary prior distribution. This model is close to the CycleGan (Zhu et al., 2017) that also employ an intermediate representation from an autoencoder to learn a GAN model. In regards to rapid grow of this sub-domain of GAN and the large number of papers related to GAN, one can easily find out more details and models in the main conferences and journals related to AI<sup>8</sup>. Nonetheless, GAN does not take into consideration the target classes when generating the samples, since *golden-targets* (i.e manual transcriptions) are not available at testing time.

We are interested in such as models since they allow NLP based systems to both generate unseen representations and extract robust features from the document content. We have starting to follow this way and propose novel architectures to process noisy spoken documents (Titouan et al., 2019a). The task considered in this thesis (see chapter 6) replaces the initial language by an automatic transcription of a conversation, and the target language by its manual transcription. Furthermore, we propose to perform classification on top of the generation, as driven by the semi-supervised GAN (SGAN) approach (Odena, 2016).

## Structure of the thesis

The document relates different research directions followed since my Ph.D. thesis defended at 2014 November. The different chapters thereafter are initiated with Ph.D. students that I have supervised from 2015 to 2019. The first direction followed was related to my interest on machine learning methods for NLP real-world applications. We have started by evaluating hitherto proposed neural networks based models for Natural Language Processing (NLP) tasks. These models have then been extended with original algorithms to improve the efficiency during the NLP related tasks. Then, we have proposed novel neural networks architectures based on hyper-complex algebra to better handle multidimensional information. These approaches have encountered a wide success and interest from different institute and researchers from heterogeneous AI related fields. Alongside to these works, I have started new directions to allow neural network algorithms to process large amount of documents with less processing time and better performances than mere ones. All these research directions will be followed throughout the ANR AISSPER project that will start at January 2020 in collaboration with the LIUM (Le Mans University) and Orkis (Aix-en-Provence company). Indeed, AISSPER will first employ already proposed neural based architectures and then novel ones for end-to-end SLU tasks.

Part I of the document describes the first works on machine learning for NLP. The aim of this period was to evaluate hitherto proposed neural networks based algorithms for real-world NLP applications. We present a research done during the Mohamed Bouaziz's Ph.D. thesis, in CIFRE convention with the company EDD and co-supervised by Georges Linarès (LIA), Richard Dufour (LIA) and myself between September 2013 and December 2017. Therefore, chapter 1 presents an original Long Short-Term Memory-based (LSTM) architecture, called Parallel LSTM (PLSTM), that carries out multiple parallel synchronized input sequences in order to predict an

<sup>&</sup>lt;sup>8</sup>For example, this website gives both papers and codes for a large number of GAN models: https://github.com/eriklindernoren/PyTorch-GAN

output. The proposed PLSTM could be used for parallel sequence classification purposes. The PLSTM approach is evaluated during an automatic telecast genre sequences classification task. The results show that the proposed PLSTM method outperforms the baseline and the LSTM model. This work correspond to the first step and direction on machine learning for NLP. This effort is followed by a study of encoder-decoder neural networks for NLP tasks. Chapter 2 corresponds to part of the work done during the Killian Janod's Ph.D. thesis in CIFRE convention with the company Orkis and co-supervised with Georges Lianrès (LIA), Richard Dufour (LIA) from September 2013 to December 2017. This chapter studies different set of features from encoder-decoder neural networks based architectures for the Spoken Language Understanding (SLU) task of DECODA theme identification task of spoken dialogues. These dialogues are transcribed from an Automatic Speech Recognition (ASR) system. These automatic transcripts are affected by errors from automatic transcription process that are especially frequent since speech signal in recorded in noisy conditions. An ASR system induces linguistic errors in features employed for different NLP related applications. This chapter deals with the recovery of corrupted linguistic features in spoken documents by introducing robust features from encoder-decoder neural networks based models evaluated in the real-world application of the DECODA spoken conversation analysis task.

Part II details novel neural networks based architectures partially realized in collaboration with Montréal Institute of Learning Algorithme (MILA) researchers. Among them, Y. Bengio, M. Ravanelli and others have collaborated on Quaternion valued neural networks for speech processing. We have proposed novel research directions to introduce complex-valued neural networks to better manage internal dependencies between features of a basic element. Therefore, chapter 3 presents these promising models based on convolutional neural networks and the Quaternion hypercomplex numbers called "Quaternion Convolutional Neural Network" (QCNN). We report in this thesis the QCNN evaluated on an NLP related task (Theme identification task of spoken dialogues). Chapter 4 introduces the Quaternion Recurrent Neural Networks (QRNN) and an extension of the LSTM to Quaternion algebra called QLSTM. These models are evaluated during the speech recognition task of TIMIT. This work have been realized during the Ph.D. thesis of Titouan Parcollet co-supervised with Georges Linarès (LIA) and during the internship of Titouan Parcollet at the MILA (Canada) under the co-supervision of Yoshua Bengio.

Part III of the thesis corresponds to the ongoing and futur research directions. We first present an ongoing research that will lead to part of my futur research. The previous chapters chart the path of my research throughout different neural networks based topics and language processing applications. These models compose the first step of my research in the area of machine learning for NLP and require a large even huge processing time during the learning process. Chapter 5 introduces novel promising research perspectives for faster and more efficient neural networks based models and exposes a research direction based on a ongoing work for parsimonious recurrent neural network. Nowadays even more real-life NLP related applications are

based on deep learning algorithms. These models require a long processing time for the learning phase. The proposed model allow NLP based systems to achieve better results than other RNN architectures with less processing time. The work started on studying encoder-decoder neural networks is extended in chapter 6 with the first efforts on proposing novel generative models to extract from noisy spoken documents robust features for SLU. Finally chapter 7 presents future short-term directions on both real and quaternion-valued neural networks, and long-term perspectives on parsimonious neural networks and GAN. Overall, this thesis is an overview of my futur directions in terms of neural based architectures (RNN, GAN, ...) and real-world applications (find out more efficient algorithms for NLP tasks) in collaboration with different Ph.D. students, interns and colleagues.

# Part I

# Real-valued Neural Networks for Natural Language Processing

## **Chapter 1**

# **Parallel Recurrent Neural Networks**

### Contents

1.1	Introduction	28
1.2	Parallel Long Short-Term Memory (PLSTM)	29
1.3	Experiments	32
1.4	Results and Discussion	33

### Abstract

We present in this chapter an original LSTM-based architecture, named Parallel LSTM (PLSTM), that carries out multiple parallel synchronized input sequences in order to predict a common output. The proposed PLSTM could be used for parallel sequence classification purposes. The PLSTM approach is evaluated on the automatic telecast genre sequences classification task. Results show that the proposed PLSTM method outperforms the baseline n-gram models as well as the state-of-the-art LSTM approach. This chapter corresponds to part of the work done during Mohamed Bouaziz's Ph.D. thesis, in CIFRE convention with the company EDD and co-supervised by Georges LianrÃ's (LIA), Richard Dufour (LIA) and myself between September 2013 and December 2017.

## 1.1 Introduction

The work presented in this chapter is from the collaboration with Mohamed Bouaziz during his Ph.D. thesis in CIFRE<sup>1</sup> with the EDD (L'Europèenne de Donnèes) company from Paris (France). EDD processes large amount of multimedia documents and streams to provide for the customers a set of speech and natural language related services such as documents retrieval and indexation. In this context, the Ph.D. thesis has focused on the prediction of the genre label of telecasts on Electronic Program Guide (EPG) multi-streams of different channels. Nonetheless, the proposed architecture can easily process different multidimensional data or streams. The context of the Ph.D thesis of EPG prediction, has many concerns related to the difficulty to express an event (telecast genre) given a sequence of previous telecasts that are not taking place at the same time each day. Therefore, the use of different parallel channels gives a more robust context to predict the genre label for a given channel. Moreover, one can find, on the internet, an impressive list of programs available on demand at any time of the day. This forces the original TV channels to diversify their offers and to propose programs that best meet the expectations of viewers regarding the period of the day. However, this large amount of streams needs to be analyzed and organized to allow the consumers and industries to process effectively the mass media content. Scientists have been given a certain level of interest to automatic sequence classification of TV channels' speech in a multi-channel context (Bredin et al., 2014; Bouchekif et al., 2013). Indeed, recently, automatic sequence classification became an ubiquitous problem, having then encountered a high research interest (Gers et al., 2001; Severyn et Moschitti, 2015; Huang et al., 2016). This is due to the need to structure knowledge as a set of dependent localized information alongside with the new computer capabilities to efficiently process large amount of data. Among the recent methods employed to structure these sequences, the machine learning domain provides a set of high-level representations well adapted to automatic sequence classification based on Deep Neural Networks (DNN) such as Convolutional Neural Networks (CNN) (LeCun et al., 1998) or Recurrent Neural Networks (RNN) (Elman, 1990). RNN architectures such as Long Short-Term Memory (LSTM) (Hochreiter et Schmidhuber, 1997) and Bidirectional LSTM (BLSTM) (Graves et Schmidhuber, 2005) have gained a particular attention in different domains and tasks including sentence (Sundermeyer et al., 2012) or successive images (Vinyals et al., 2015) processing. Nonetheless, RNNs or RNN-HMM could not be directly employed for sequence classification using multiple inputs from synchronous streams such as TV shows coming from different channels. Indeed, RNNs can only be trained to make a set of elements labeled in a single stream of input information.

We have introduced an original multi-stream neural network architecture, called Parallel LSTM (PLSTM), that simultaneously takes into account different synchronous streams in order to automatically classify this multi-stream sequence. To evaluate the effectiveness of the proposed PLSTM multi-stream neural network architecture, exper-

<sup>&</sup>lt;sup>1</sup>The CIFRE convention (Conventions Industrielles de Formation par la REcherche) is a partnership between academics and industrials for granting a Ph.D. thesis with mainly an real world application purpose. More information on the CIFRE convention are available at http://www.anrt.asso.fr/fr/cifre-7843

iments are carried out on the LIA's Electronic Program Guide (EPG) dataset containing 3 years of TV programs from 4 different channels. The PLSTM performance is compared with the LSTM as well as a classic n-gram models considered as the baseline. Our PLSTM approach is an important step further for sequence classification since it can be applied to any set of synchronous sequences. The annexe proposes an overview of a couple of RNN architectures.

## 1.2 Parallel Long Short-Term Memory (PLSTM)

Sequence models, such as LSTM based RNNs, are well known for their capability to classify sequential data. On the other hand, these models can only take into account homogeneous sequential data from a single stream. In the case of additional sources, these models can not be employed. In order to take into account parallel sequences, a solution is to combine these streams viewed as a single sequence. This combination is carried out by two models with a certain number of limits.



*Figure 1.1:* Combination of parallel sequences: an illustrative example with 3 streams on the last 3 events :  $E_t$  coressponds to the  $t^{th}$  event.

#### Multi-streams input features of the PLSTM

The first method of combining the parallel sequences consists of a kind of alignment between the different sequences in order to merge the events of the same order within a single event (see Figure 1.1-b). It is a question of combining the event terms  $(1 \le t \le T)$  of each of the *N* sequences to form the compound  $t^{\text{th}}$  event. The sequence

of *T* compound events is then used as a conventional sequence where each event *t* is represented by a vector of N events. The limitations of this method ere related to, for example, an LSTM layer will consider a recurrent link between the  $t^{\text{th}}$  and the  $(t + 1)^{\text{th}}$ composed event. As a consequence, a dependence between the streams will be learned, wrongly, between the  $t^{\text{th}}$  basic event of the stream n' and the  $(t+1)^{\text{th}}$  basic event of another stream n'' ( $n' \neq n''$ ,  $1 \le n' \le N$  and  $1 \le n'' \le N$ ). In Figure 1.1-b, an example of this potentially wrong dependency is between the basic event at t = 1 of stream 2 and the elementary event at t = 2 of stream 1. The first event appends after the second one The second method concatenates the N parallel sequences, each containing T events, to obtain a single unidimensional "super-vector" of length  $T \times N$  (see Figure 1.1-c). the drawback of this method of concatenation is that the vector produced does not represent a sequence but rather a sequence of sequences. Models adapted to sequential data consider any input data as a homogeneous sequence of events. The combination of multi-stream sequential data and their inclusion in sequence models is, in most of the cases, irrelevant. We nevertheless retain the use of the first method of combining the parallel sequences since the latter is more prone to practical order issues.

### PLSTM model

The Bidirectionnal-RNN (BRNN) neural architecture, presented in the annexe, uses the same sequence  $\mathbf{x}$  as an input for both forward and backward directions, which is useful for information from a single stream. We propose in this chapter an original neural recurrent neural network, called Parallel RNN (PRNN) presented in Figure 1.2, that takes advantage from the BRNN structure in a multistream context. The original PLSTM architecture corresponds to the PRNN description by replacing the  $\mathcal{H}$  function with the LSTM composite function. PLSTM differs from the classical BLSTM by feeding forward, not a shared sequence, but different input vectors through a dedicated hidden layer  $\mathbf{h}^n$  for each input vector  $\mathbf{x}^n$ . Moreover, BLSTM employs only 2 hidden layers due to its bidirectional architecture while PLSTM is potentially composed with multiple hidden layers. The input sequences are considered independent and require to be mapped in homogeneous separate subspaces (W matrix from input x to hidden h spaces). Therefore, a single LSTM containing concatenated inputs from different independent sequences is not theoretically suitable for finding out a common homogeneous subspace to map heterogeneous input representation of parallel sequences. Thus, for each  $n^{th}$  stream  $(1 \le n \le N)$ , the PLSTM takes the input sequence  $\mathbf{x}^n = (x_1^n, x_2^n, \dots, x_T^N)$ and computes the hidden sequence  $\mathbf{h}^n = (h_1^n, h_2^n, \dots, h_T^N)$  and the output vector **y** by iterating from t = 1 to T.

$$h_t^N = \mathcal{H}(\mathbf{W}_{x^N h^N} x_t^N + \mathbf{W}_{h^N h^N} h_{t-1}^N + b_h^N)$$
(1.1)

$$h_t^2 = \mathcal{H}(\mathbf{W}_{x^2h^2}x_t^2 + \mathbf{W}_{h^2h^2}h_{t-1}^2 + b_h^2)$$
(1.3)

$$h_t^1 = \mathcal{H}(\mathbf{W}_{x^1h^1} x_t^1 + \mathbf{W}_{h^1h^1} h_{t-1}^1 + b_h^1)$$
(1.4)

$$y_t = \sum_{n=1}^{N} \mathbf{W}_{h^n y} h_t^n + b_y \tag{1.5}$$

where N is the number of streams. In our experiments, the output vector y takes



Figure 1.2: Parallel Long Short-Term (PLSTM) neural network.

advantage of the *N* channels to predict the telecast's genre for one given channel *n*  $(1 \le n \le N)$ .

Therefore, PLSTM feeds forward separate sequences in order to predict a label and codes internal hidden structures between the parallel hidden sequences. (Graves et Schmidhuber, 2005) introduces the BLSTM with Back Propagation Trough Time (BPTT) algorithm (Schuster, 1999) for training. For our proposed PLSTM architecture, the training takes place over *N* input sequences:

<u>Forward Pass</u>: feeds all input data for the sequences into the PLSTM and determines the predicted outputs.

- For 1 to *T*: Forward pass for each of the *N* hidden layers.
- For all *t* ∈ [1...*T*]: perform the forward pass for the output layer using the activation state of the *N* hidden layers.

<u>Backward Pass</u>: processes the error function derivative for the sequences used in the forward pass.

- For all  $t \in [1 \dots T]$ : perform the backward pass for the output neurons.
- For *t* from *T* to 1: perform the backward pass for each hidden layers using the error observed on the output layer.

<u>Updating Weights</u> The update weight for all matrices from each hidden state related to each stream is computed in the same manner than for the mere LSTM (Hochreiter et Schmidhuber, 1997).

## 1.3 Experiments

Multistream sequence classification is evaluated with the proposed PLSTM architecture (2 and 4 parallel sequences) as well as the classic LSTM network on an automatic TV show genre labeling task. Two n-gram based models (baseline) are also considered for fair comparison. Next sections describe the dataset, the genre sequence classification as well as the neural networks settings.

### Multichannel EPG dataset

The Electronic Program Guide (EPG) dataset is extracted from 4 French TV channels (M6, TF1, France 5 and TV5 Monde) for 3 years, from January 2013 to December 2015. M6 channel is used in our experiments as the output stream. Data from 2013 and 2014 are merged and split into the *training* (70%) and *validation* (30%) datasets using a *stratified shuffle split* (Pedregosa et al., 2011) in order to preserve the same percentage of samples of each class in the output of both folds, while the 2015 dataset is kept for testing. In order to guarantee a clean experimental environment, labels (*i.e.* genres) that are absent at least in one of the three folds were removed. Doing so allows us to have equivalent datasets in terms of labels vocabulary. Table 1.1 shows the genres distribution for M6, the chosen output channel.

Genres	Training	Validation	Test
Weather	2,691	1,153	1,712
Fiction	1,890	810	1,478
News	913	392	679
Other magazine	981	421	466
Music	461	197	340
Teleshopping	421	180	308
TV game show	476	204	287
Cartoon	361	155	205
Other	277	119	133
Reality TV	83	36	76
Documentary	29	13	14
Total	8,107	3,680	5,698

Table 1.1: Genres Distribution for train, validation and test sets in M6 channel output.

#### **Genre Prediction Experiments**

For a given input history sequence (composed of the *n* previous telecast genres), a genre label representing the next M6's telecast is output. The size of the genre sequences (*n*) varies from 1 to 4. Then, three input configurations are employed. **Mono-channel input:** only M6 history sequences for a baseline *n-gram* experiment (with a *statistical language model* from the SRILM toolkit (Stolcke et al., 2002)) and a straightforward *LSTM* model. **Bi-channel input:** both M6 and TF1 channel histories are employed as input for *P2LSTM* (PLSTM with two parallel streams as a BLSTM

with forward-forward directions and separate inputs). The aim of this experiment is to move up the context's information using a similar and rival channel, the two being generalist channels. **Multichannel input:** History of each of the 4 streams (*i.e.* channels) is used as input for *4n-gram* and P4LSTM experiments (PLSTM with 4 parallel streams).

#### Neural Networks Setup

The mere LSTM, and the proposed P2LSTM and P4LSTM, are composed with 3 layers: input layer **x** of size varying from 1 to 4, a hidden layer **h** of size 80 for all LSTM-based models and an output layer **y** with a size equals to the number of different possible TV genres (11). The Keras library (Chollet, 2015), based on Theano (Bastien et al., 2012) for fast tensor manipulation and CUDA-based GPU acceleration, has been employed to train neural networks on an Nvidia GeForce GTX TITAN X GPU card. The training times, detailed in Table 1.2 for all models, match with the sequence size of all models. Indeed, even with the most time-consuming configuration, namely P4LSTM with 4 elements history, the training does not last more than 25 minutes.

Sequence size	1	2	3	4
n-gram	1	1	1	1
4n-gram	2	5	17	51
LSTM	51	146	319	362
P2LSTM	259	473	485	439
P4LSTM	536	923	844	1,386

**Table 1.2:** Training times (in seconds) of models employed during the experiments for different telecast genres sequence sizes.

## 1.4 **Results and Discussion**

Table 1.3 shows the overall results, in terms of the standard F1 metric related to the genre prediction task outputs, using each method and for different stream sequence sizes from 1 to 4.

Seq. size	n-gram	4n-gram	LSTM	P2LSTM	P4LSTM
1	19.07	59.39	11.60	47.46	47.24
2	51.38	58.35	34.64	54.17	59.54
3	57.41	57.10	50.74	58.69	59.92
4	56.87	57.26	56.47	58.67	60.81

Table 1.3: F1-score (%) of each n-gram and LSTM models.

#### N-gram based models

The multi-channel 4n-gram model outperforms the simple n-gram one for each of the different 4 genre sequence configurations except for 3 sized history. *4n-gram* method reaches around 59% of F-score using 1 sized sequences against near 57%

for mono-channel n-gram using its best history configuration. The observed results confirm the interest of using multiple streams to predict the next telecast's genre for a specified channel.

### LSTM and PLSTM

One can figure out from Table 1.3 that mono-channel LSTM does not even outperform the mono-channel n-gram experiment. P4LSTM obtains the best result with an F1 score close to 61% using a sequence of size 4. In order to analyze these results, the Error Rates (ER) are also presented in Table 1.4. The overall F1 scores are different from those related to the ER. For example, at its best configuration of a 4 sized sequence, P4LSTM error rate reaches about 23.5%, which corresponds to a correct rate of 76.5% against an F1-measure of only 61%. Moreover, although the range between the lowest and the highest values is almost 12 points for ER, it is only near 4 points for F1-measure. The reason of this is that the F1-metric may be not suitable for the task due to the labels imbalance with different numbers of genre occurrences varying from 14 to 1,712 in the test set.

Seq. size	n-gram	4n-gram	LSTM	P2LSTM	P4LSTM
1	51.36	30.29	60.13	36.68	34.35
2	39.06	28.99	48.99	29.38	25.08
3	31.71	29.64	35.73	28.85	24.76
4	35.43	30.5	29.59	28.27	23.52

Table 1.4: Error rates (ER) observed for each n-gram and LSTM models for different sequence sizes.

### Discussion

Confusion matrices of 4n-gram and P4LSTM methods are shown in Tables 1.5 and 1.6 to point out benefits of the proposed PLSTM model.

1282	65	0	323	11	4	17	0	5	2	3
292	904	0	177	35	9	15	26	16	0	4
2	12	636	1	27	0	0	0	1	0	0
69	40	3	296	10	2	6	0	29	6	5
16	35	17	5	219	0	3	2	43	0	0
45	1	0	1	0	246	0	0	15	0	0
2	7	0	19	0	0	245	0	6	8	0
0	32	6	0	9	<u>102</u>	0	56	0	0	0
10	10	0	25	1	2	15	0	66	2	2
20	14	0	9	0	0	4	0	19	9	1
4	3	0	4	0	0	0	0	2	0	1

*Table 1.5:* Confusion matrix for the 4n-gram output: labels are shown according to their decreasing frequency as in Table 1.1.

It is worth emphasizing that most of the missed instances in all systems are wrongly

1624	38	0	46	0	2	0	0	2	0	0
375	861	0	188	18	0	18	16	2	0	0
4	0	669	1	5	0	0	0	0	0	0
114	34	0	306	6	0	3	0	3	0	0
15	18	0	1	297	0	7	2	0	0	0
49	15	0	0	0	244	0	0	0	0	0
49	14	0	14	11	0	199	0	0	0	0
25	28	1	0	0	4	0	147	0	0	0
70	3	0	48	0	1	0	0	11	0	0
35	0	0	23	0	0	14	0	4	0	0
8	0	0	5	0	0	1	0	0	0	0

*Table 1.6:* Confusion matrix for the P4LSTM output: labels are shown according to their decreasing frequency as in Table 1.1.

labelled as one of the two most frequent classes, *Weather* and *Fiction*, as well as the *Other Magazine* genre (some examples are in green cells). *False positives* are more recurrent in some small classes such as *Other Magazine* than in the relatively more frequent class *News*. The reason is that *News* is a well defined genre occurring mostly at the same time each day unlike *Other Magazine* genre that encompasses various telecasts that are broadcast at several and irregular daytimes. *Teleshopping* shows are often broadcast at more nearly the same time of morning than *Cartoons* and affect dramatically the performance of the 4n-gram model in this context (cf. underlined italic cell in Table 1.5). Finally, the confusion matrix of P4LSTM experiment shows also that this system fails to predict the least frequent genres, *Reality TV* and *Documentary*. For example, for the two least frequent genres, *Reality TV* and *Documentary*, respectively none of the 76 and the 14 instances were correctly found. This leads to a *precision* of 0 which penalizes the average precision and then the overall F-score. In order to evaluate the impact of the

Seq. size	n-gram	4n-gram	LSTM	P2LSTM	P4LSTM
1	23.30	70.32	14.18	58.00	57.74
2	59.47	68.04	42.34	66.21	72.77
3	66.71	66.55	62.01	71.74	73.23
4	65.67	66.77	69.01	71.71	74.32

*Table 1.7:* F1 score (%) of n-gram and LSTM models, the two least frequent genres Reality TV and Documentary not being included.

least frequent genres on the F1 measure, especially on the three LSTM based systems, we also reported on Table 1.7 the F1 results on the same outputs of the experiments of Table 1.3 by excluding the two least frequent genres from the averages of precision and recall (*Reality TV* and *Documentary*). The state-of-the-art mono-channel LSTM performances gradually become closer and closer to the multichannel n-gram model ones (cf. Figure 1.3) when the size of sequences moves up with an F1 score of 69%. Therefore, LSTM-based models require longer sequences to learn long term dependencies than the 4n-gram methods. Overall, the results of the PLSTM detailed in Table 1.4


*Figure 1.3:* F1 score for n-gram and LSTM models, the two least frequent genres Reality TV and Documentary not being included.

and Figure 1.3, demonstrate the benefits obtained at least for history sequences longer than 2 genres with an F1 score greater than 71%. Regarding multichannel P4LSTM approach, the highest performance reaches an F1-measure of about 74% using 4 sized sequences with a gain of about 3 and 4 points compared respectively to P2LSTM and 4n-gram model best performances. With the rapid grow of multimedia documents such as videos, the results observed with the PLSTM are promising and will allow NLP systems to process multidimensional data. Indeed, researchers across pattern recognition domains propose more and more neural networks based systems to process effectively multi streams for image processing (Singh et al., 2016), video classification (Wu et al., 2016) or speech processing (Ma et al., 2019). This work will be followed to propose novel neural based architectures to better consider each stream in an appropriate condition. For example, LSTM cells are composed with gates that manage in different level the knowledge stream. Indeed, some gates are devoted to expose the output or forget the hidden state. Nonetheless, the input gate has to propose a suitable hidden state candidate for a given stream but all streams have not enough information on this given stream to effectively provide part of the hidden state candidate.

## **Chapter 2**

## **Encoder-Decoder Neural Networks**

#### Contents

2.1	Introduction		
2.2	Autoencoder neural networks based systems		
2.3	<b>Experiments</b>		
	2.3.1 DECODA framework 46		
	2.3.2 Automatic Speech Recognition System 47		
2.4	Results and Discussion 48		

#### Abstract

This chapter corresponds to part of the work done during the Killian Janod's Ph.D. thesis, in CIFRE (Orkis) where we study different set of features from different encoder-decoder neural networks for the DECODA theme identification task of spoken dialogues. These dialogues are transcribed with an Automatic Speech Recognition (ASR) system and are affected by errors that are especially frequent when speech signal is recorded in noisy conditions.

#### 2.1 Introduction

The other research direction related to hitherto proposed real-valued neural networks for language processing is presented in this chapter. This work has been made during the Ph.D. thesis of Killian Janod. The aim of this work is to find robust representations of spoken documents automatically transcribed by an Automatic Speech Recognition System (ASR). These transcripts are error prone and NLP applications require robust features representing the document content to process effectively these documents. These features for characterizing a document or a conversation can be words, their relations in sentences or discourse structures. Selecting and extracting relevant features of these types depend on the analysis task and is, in general, a very difficult and not completely solved research problem. The problem is even more complex for spoken conversations since words are hypothesized by an ASR component with transcription errors partly due to speech signal variability caused by speaker characteristics, environment and channel noises, or acquisition devices, as well as a highly conversational language register (hesitations, repetitions, grammatical errors, disfluencies...). Nevertheless, important and applicable results have been obtained on specific tasks as speech analytics (Melamed et Gilbert, 2011), topic identification (Lagus et Kuusisto, 2002; Hazen, 2011) and segmentation (Eisenstein et Barzilay, 2008; Purver, 2011), and other spoken conversation analysis tasks reviewed in (Tur et De Mori, 2011). We have proposed a set of neural networks based models to reduce the impact of ASR errors on the identification of a predetermined set of topics assuming that a spoken document can be classified as belonging to one and only one topic as discussed in (Hazen, 2011). This assumption is acceptable if the task is to classify a single topic corresponding to the theme of a conversation. An example of conversation theme is the type of customer concern/request/complaint expressed during a customer care service (CCS) conversation. A fragment of a conversation of this type in the application domain is shown in Figure 2.1. The basic problem is to reduce the effect of ASR errors. For this purpose, two different objectives have been identified, inspired by two deep learning approaches: to consider ASR errors as noise to be reduced with denoising autoencoders (DAE); to make abstractions of ASR features with stacked autoencoders to obtain distributed representations that tend to capture features of semantic contents characterizing conversation themes. These features are expected to produce an effective generalization useful for training classifiers in specific domains (Hinton et Salakhutdinov, 2006; Bengio et al., 2007). The conversation is represented (input features) by a product of word term frequency (TF), inverse document frequency (IDF) as suggested in (Hazen, 2011), and posterior probabilities of a theme given a term. The input is then processed by a stack of autoencoders and a DAE to obtain robust latent features sets to be used for hypothesizing conversation themes with a multi-layer perceptron (MLP) classifier. When DAEs are used, the input variability caused by ASR errors is considered and reduced in the attempt to recover the same type of features obtained with manual transcripts. When stacked autoencoders (SAE) are used the objective is to reduce variability of ASR features without imposing the reconstruction of clean features. The performances obtained with DAEs and SAEs features are compared in a theme identification task. Moreover, effective encoder/decoder architectures are proposed for



*Figure 2.1: Example of a dialogue from the DECODA corpus labelled by the agent as a transportation card issue but also containing the infraction theme.* 

reconstructing bottleneck features obtained with manual transcripts using bottleneck features obtained from ASR transcripts. These architectures are inspired by the consideration that bottleneck features can be further processed to encode a level of abstraction eliciting relevant semantic contents encoded in the input features. A further investigation is proposed for adapting the reconstruction process to a specific task. Experimental results confirm that the proposed novel autoencoders that integrate corrupted and clean feature abstractions provide better evidence of semantic contents relevant for a topic identification task.

### 2.2 Autoencoder neural networks based systems

We have proposed to consider ASR transcripts as noisy data and regards transcription correcting as denoising. A discussion in (Turian et al., 2010) suggests that neural word representations may capture word semantic information. These representations can be obtained with unsupervised learning of autoencoder parameters. As a first step, the possibility is investigated that hidden features obtained with simple and stacked autoencoders embed sufficient semantic information for characterizing themes of noisy conversations. As a second step, hidden features obtained with denoising autoecoders are considered to evaluate the impact of training for reconstructing an input artificially perturbed with a noise model. As manual transcriptions are available for a train set, in a third step, features are extracted with semi-supervised reconstruction of manual transcription features from ASR noisy transcription features. Notice that manual transcriptions features can be perturbed by a variability induced by the fact that CCS users may mix information relevant for a theme with personal or factual irrelevant information. As a fourth step, the possibility of stacking hidden layers of features estimated with the previously mentioned autoencoders is considered with or without global fine-tuning.

#### **Document representation**

The task considered in this chapter is the reconstruction of a feature distribution corrupted by the imprecision of the feature extraction component. Features are obtained from a set of content words exhibiting high mutual information with the application domain themes. Given a document of a corpus  $\mathbb{D}$ , an input feature  $x_d \in \mathbb{R}^n$  called TF-IDF-Gini is defined with its *i*-th element computed as follows:

$$x_i = tf(i) \times df_{\mathbb{D}}(i) \times G(i) \tag{2.1}$$

where tf(i) the number of occurrences of terms *i* in the document,  $df_{\mathbb{D}}(i)$  is the inverse document frequency (IDF) and G(i) the word purity. The purity of a feature *f* is scored with the Gini criterion defined as follows:

$$G(f) = \sum_{t \in \mathbb{T}} \mathbb{P}^2(t|f) = \sum_{t \in \mathbb{T}} \left( \frac{df_t(f)}{df_{\mathbb{D}}(f)} \right)^2$$
(2.2)

where  $df_t(f)$  is the number of document containing the term f annotated with label t.



Figure 2.2: Autoencoder model. Biases are omitted for the sake of simplicity.

#### **Basic Autoencoder Concepts**

An Autoencoder (AE) is a three-layer feed-forward neural network made of an encoder and a decoder as shown in Figure 2.2. The encoder computes a hidden representation of x made of a vector h of size m (number of hidden units) as follows:

$$h = \sigma(W^{(1)}x + b^{(1)}), \qquad (2.3)$$

where  $W^1$  is a  $m \times n$  weight matrix and  $b^{(1)}$  is a *m*-dimensional bias vector.  $\sigma(.)$  is a non-linear activation function. In the approach proposed in this chapter the function is the hyperbolic tangent defined as follows:

$$\sigma(y) = \frac{e^y - e^{-y}}{e^y + e^{-y}}$$
(2.4)

The decoder attempts to reconstruct the input vector x from the hidden vector h to obtain the output vector  $\tilde{x}$ :

$$\tilde{x} = \sigma(W^{(2)}h + b^{(2)}), \qquad (2.5)$$

where the reconstructed vector  $\tilde{x}$  is a *n*-dimensional vector,  $W^{(2)}$  is a  $n \times m$  weight matrix and  $b^{(2)}$  is a *n*-dimensional bias vector. AE parameter estimation is performed by minimizing the following reconstruction Mean Square Error (MSE)  $L_{\text{MSE}}$  with respect to the set of parameters  $\theta = \{W^{(1)}, b^{(1)}, W^{(2)}, b^{(2)}\}$  using the MSE (Bengio, 2009) :

$$L_{\text{MSE}}(\theta) = \frac{1}{d} \sum_{x \in D} l_{\text{MSE}}(x, \tilde{x}) = \frac{1}{d} \sum_{x \in D} ||x - \tilde{x}||^2$$
(2.6)

Two AEs are trained for the task considered in this chapter: one for reconstructing features ( $x^{(ASR)}$ ) of automatic transcriptions from an ASR system, and the other one to recount features ( $x^{(TRS)}$ ) of manually transcribed documents. The parameters estimated for these autoencoders ( $W^{(ASR)}$  and  $W^{(TRS)}$ ) are pre-trained for further use to estimate the bottleneck features (hidden layer) of the Supervised deep autoencoder.



*Figure 2.3:* Stacked autoencoder (SAE) architecture (Janod et al., 2016). The parameters of each layer are estimated as for a shallow autoencoder (left). After a pre-training step, the hidden layers are stacked to form a denoising autoencoder (right) whose parameters are further fine-tuned by global error back propagation.

#### Stacked Autoencoders (SAE)

It has been argued in (Bengio et al., 2007; Hinton et al., 2006) that DNNs may encode input data at progressively deeper levels of abstraction in successive hidden layers

of stacked autoencoders (SAE). In this type of DNN with *k* hidden layers, the latent features at the *i*-th intermediate hidden layer, for an input vector *x*, are computed as with equation 2.3 by setting  $h^{(0)} = x$ , and  $h^{(i)} = \sigma(W^{(i)}h^{(i-1)} + b^{(i)}) \forall i \in \{1, 2, ..., k\}$ . Therefore, each layer is pre-trained as a shallow autoencoder for a fixed number of iterations. The learnt hidden layer vector  $h^{(i)}$  is stored and used to learn the next layer  $h^{(i+1)}$ . Greedy pre-training is progressively performed in this way starting with  $h^{(i+1)}$  as shown by Figure 2.3. After pre-training the last layer, a fine-tuning is performed on the entire stack of hidden layers to obtain a generative model providing different levels of abstractions for the input vector *x*.

#### **Denoising Autoencoder (DAE)**

The aim of an autoencoder is to build a robust generative model to encode and decode a given input vector x in a latent space h. During the learning process, the autoencoder fails to completely separate relevant information from residual noise for a given input distribution (Vincent et al., 2010). For this reason, (Vincent et al., 2010) propose to corrupt the inputs before the encoding process and then decode this noisy representation to a clean one with a Denoising Autoencoder (DAE). In this way, the DAE is expected to recover a clean representation from a noisy input by learning a robust generative model. In this model, the input vectors x are considered as "clean" representations. The aim of this DAE is to obtain a robust reconstruction from an input vector to a clean output one. Therefore, x is artificially corrupted via a function that can be:

- An Additive isotropic *Gaussian Noise* (GS)  $x^{(\text{corrupted})}|x \sim \mathcal{N}(x, \sigma^2 I)$
- A *Masking Noise* (MN) which forces a fraction of the element of *x* to 0.
- A *Salt-and-pepper Noise* (SN) which forces the values of some elements of *x* to their minimum or maximum possible value (0 or 1) according to a fair coin flip (Vincent et al., 2010).

This corrupted input  $x^{(\text{corrupted})}$  is then mapped to a hidden layer with the (3) by replacing x with  $x^{(\text{corrupted})}$ . During the learning process, the denoising autoencoder learns the parameters  $\theta = \{W^{(1)}, b^{(1)}, W^{(2)}, b^{(2)}\}$  to minimize the reconstruction error  $L(x, \tilde{x})$ . The motivation for using this type of DAE is that a good representation *h* of a corrupted or partially destroyed representation of an input vector *x* is informative of *x* and invariant to a perturbation  $x^{(\text{corrupted})}$  of *x* due to noise. The conditional distribution used to generate the corrupted version  $x^{(\text{corrupted})}$  is artificially induced. The problem considered in this chapter is different since input features are extracted from already noisy real-life data. In fact, input features originate from conversations recorded with real-life background noise. Such a noise is unpredictable and a noise model cannot be identified as the corruption of a clean input. This motivates the proposal of a new approach based on hidden bottleneck features and the incorporation of both noisy and clean documents in the learning process.

#### **Basics Concepts on Supervised Bottleneck Features**

Multi-Layer Perceptron (MLP) are feed-forward neural network made of an input

layer, an output layer and at least one hidden layer. MLP networks are known to be a universal function approximator (Haykin et Network, 2004). In the work described in this chapter, the MLPs are used for document classification. For each hidden layer, a latent representation of the input is computed using the (3) with the same setting as for the stacked autoencoders. Usually the class is predicted by output layers using a non-linear Softmax function defined by:

$$\sigma(y) = \frac{e^{y_j}}{\sum_{k=0}^{K} e^{y_k}} \text{ for } j \in 1, \dots, K$$
(2.7)

Where K is the number of elements in y. The learning process attempts to reduce the parameters  $\theta = W_0, ..., W_j, b_0, ..., b_j$  to minimize the prediction error  $L(x, \tilde{x})$ . The state-of-the-art loss function used in an MLP is the Cross Entropy Error (Nasr et al., 2002). If the MLP parameters are estimated with output supervision, the latent representation encoded in each hidden layer is more abstract and more meaningful for the task than the one computed just for reconstructing the previous layer (Haykin et Network, 2004). Bottleneck features are those of the hidden layer that provides the more meaningful, abstract and if possible compressed latent representation preserving at the same time enough information of the original distribution useful for higher level processing. Bottleneck features have been intensively used in neural acoustic models (Grèzl et al., 2007) (Dong Yu, 2011). The MLP based classification is used in this chapter for comparing the performance a variety of autoencoder latent features. This choice is considered to be acceptable, at least for the considered application type, since the best classification results obtained in this way are superior to those reported in (Esteve et al., 2015) using a deep neural network and the addition of semantic and LDA input features.

#### **Proposed Features from a Supervised Deep Autoencoder**

The goal of this chapter is to obtain a robust representation of a malformed and noisy document transcribed by an ASR system. This robust representation is based on latent features of a Supervised Deep Autoencoder (SDAE), using both automatic and manual transcriptions. For this reason, a new architecture shown in Figure 2.4-(c) is proposed in this chapter. The parameter estimation of some components of this architecture is initialized using the estimations obtained with the networks shown in Figures 2.4-(a) and 2.4-(b). The architecture in Figure 2.4-(c) uses, as input, a feature vector from imperfect automatic transcripts (ASR) (considered as a corrupted input), and an output feature vector from clean manual transcripts (TRS).

In this architecture, ASR bottleneck features represented by vector  $h^{ASR}$  are mapped into TRS bottleneck features represented by vector  $h^{TRS}$  by introducing a mapping layer of hidden features represented by vector h.

**Initialization:** As emphasized in (Hinton et al., 2006), the parameters initialization in the architecture of Figure 2.4-(c) is critical. For this reason, the estimation should include appropriately pre-trained weight matrices to reach optimal performance. For this purpose, the SDAE weight matrices  $W^{(ASR)}$  and  $W^{(TRS)}$  are initialized with straightforward autoencoders as shown in Figures 2.4-(a) and 2.4-(b).

Learning process: A shallow autoencoder (dashed square in Figure 2.4 (c)) is then



*Figure 2.4:* Illustration of the proposed Supervised Deep Autoencoder (c) (SDAE) initialized with weight matrices from autoencoders (ASR (a) and TRS (b)).

learnt to define the non-linear transformation from the corrupted or noisy hidden space  $h^{(ASR)}$  (green) to the clean one  $h^{(TRS)}$  (blue) through a bottleneck hidden layer h that represents an encoding of the corrupted input  $x^{(ASR)}$ . The total reconstruction error  $L_{MSE}$  defined in eq. 2.6 is computed with an error  $l_{MSE}$  evaluated between the output vector  $\tilde{x}^{(TRS)}$  and the "clean" corresponding TRS vector  $x^{(TRS)}$ :

$$l_{\rm MSE}(x^{\rm (TRS)}, \tilde{x}^{\rm (TRS)}) = ||x^{\rm (TRS)} - \tilde{x}^{\rm (TRS)}||^2$$
(2.8)

The Heterogeneous bottleneck features extraction process for a given noisy document  $x^{(ASR)}$  from the architecture presented in Figure 2.4-(c) with an encoding and a decoding phase is described as follows. **Encoding phase:** an input ASR vector  $x^{(ASR)}$  is encoded to obtain a vector h in the bottleneck hidden layer through the hidden space  $h^{(ASR)}$ ; **Decoding phase:** the vector h is then mapped in the "clean" latent space to obtain the vector  $h^{(TRS)}$  to reconstruct  $\tilde{x}^{(TRS)}$  close to the clean vector representation of the document manually transcribed  $x^{(TRS)}$ . The Supervised Deep Autoencoder (SDAE) is not fine-tuned to preserve both transformations by keeping unchanged  $W^{(ASR)}$  and  $W^{(TRS)}$ . This is consistent with the initial intuition that reconstruction error between the observed output  $\tilde{x}^{(TRS)}$  and the desired one  $x^{(TRS)}$ , should not modify the meaningful

weight matrices  $W^{(ASR)}$  and  $W^{(TRS)}$  from the dedicated Autoencoders. For the sake of comparison and to evaluate the relevance of this assumption an equivalent fine-tuned supervised autoencoder (FSDAE) which learnt  $W^{(1)}$ ,  $W^{(2)}$ ,  $W^{(ASR)}$  and  $W^{(TRS)}$  during the last training step, is proposed.

#### Task-specific Denoising Autoencoder

The second new architecture proposed has the purpose of obtaining a denoised representation of a document which is best suited for the classification task. As TRS and ASR embeddings are obtained with a theme classifier, these components are estimated with the train data of the application domain corpus. This representation



*Figure 2.5:* Illustration of the proposed Task-specific Denoising Autoencoder (TDAE) with bottleneck features from MLP trained on ASR (a) and TRS (b).

is close to the SDAE using both features from manual documents and features from noisy documents. This structure of the proposed architecture is shown in the Figure 2.5-(c). Its input is a noisy representation and its output is a clean one for the same document. Instead of relying on the denoising capacity of two DAEs, this network uses latent task meaningful features estimated by training the task classification. The architecture in Figure 2.5-(c) uses as input both the corrupted transcriptions (ASR) and the clean manual transcriptions (TRS).

**First learning step:** The TDAE is based on task specific features learnt with supervision. These features are computed by two MLPs one trained in noisy condition and one trained in clean condition. Bottleneck features are extracted from both MLPs.

Second learning step: During this step, the first and last layers are locked using

the representation extracted from the two previously trained MLPs. A shallow autoencoder is then trained to obtain the non-linear transformation from the corrupted supervised hidden space to the clean supervised hidden space through a hidden layer that represents the robust encoding of the corrupted input  $x^{ASR}$ . The total reconstruction error  $L_{MSE}$  defined in equation (2.6) is evaluated between the shallow AE output  $h^{TRS}$  and the MLP encoded version of the clean document  $h^{TRS}$ .

### 2.3 Experiments

The effectiveness of the semantic content encoded in the proposed bottleneck features is evaluated for the theme identification task of the DECODA corpus (Bechet et al., 2012). Theme identification is casted as a classification task performed by an MLP. The MLP has one hidden layer with 256 neurons. The parameters are estimated by Adam gradient descent (Kingma et Ba, 2014) with an initial learning rate of  $10^{-3}$ . The number of iterations is determined by early stopping on a development corpus.

#### 2.3.1 DECODA framework

The DECODA corpus, labeled by human annotators and made available as described in (Bechet et al., 2012), is a set of human-human telephone conversations between agents and customers of the customer care service (CCS) of the Paris transportation system RATP. This corpus is composed of 1,242 telephone conversations, corresponding to about 74 hours of signal, split as described in Table 2.1. Conversations have been manually transcribed and labelled with one theme corresponding to the principal concern mentioned by the customer. The train set is used to compose the subset of discriminative words with the TF-IDF-Gini method. A "stop list" of 126 words<sup>1</sup> is removed since they are considered as unnecessary words for the task. The train corpus contains 7,920 words, while the test corpus contains 3,806 words, only 70.8% of which occurring in the train corpus. This is due to the fact that users may mix the expression of their concerns with application domain irrelevant facts while the agents follow a protocol using words expressing domain facts. Following the protocol, agents tend to influence the users to mention, among other things, application domain related facts. This suggests focusing of words that express domain semantic facts for estimating probability distributions of pertinent conversation features. Relevant word selection using the train set is discussed in (Morchid et al., 2014b) showing that there is no advantage in theme identification with manual transcriptions of the development set if more than 800 words are used for the DECODA corpus. Based on the above findings, a set of 100 theme specific words is selected for the experiments described in the following. These words are then merged to form a vocabulary of 707 words with which the classifier input features are obtained. For each word, an input TF-IDF-Gini features is computed. Hidden topic LDA features, computed starting with probabilities of 800 words, have been used in (Morchid et al., 2014b). As large variations in classification accuracies were observed for different topic space dimensions in the development set, the topic space for which

<sup>&</sup>lt;sup>1</sup>http://code.google.com/p/stop-words/

the best classification results were observed with the development set was also used for the test set. It was also observed that the optimal hidden space for the development set was not optimal for the test set. This suggested considering in (Morchid et al., 2015a) a large number (300) of hidden LDA spaces and integrate the features obtained with all these spaces into a single *c*-vector of reduced dimension. Results obtained with the development set showed that only 116 theme specific words should be considered to obtain the best results. In (Esteve et al., 2015) the same set of 116 words was used with the same ASR system together with 58 semantic features automatically extracted to express domain relevant facts. These features were used for classification with a simple MLP. For each theme, a set of 100 specific words is then selected. These words are then merged to form a vocabulary of 707 words with which the classifier input features are obtained.

Class	Number of samples		
label	training	dev.	text
problems of itinerary	145	44	67
lost and found	143	33	63
time schedules	47	7	18
transportation cards	106	24	47
state of the traffic	202	45	90
fares	19	9	11
infractions	47	4	18
special offers	31	9	13
Total	740	175	327

Table 2.1: Composition of the DECODA corpus.

#### 2.3.2 Automatic Speech Recognition System

Experiments are performed with a large vocabulary continuous speech recognition (LVCSR) system with 230,000 Gaussians used in triphone acoustic models. Model parameters are estimated with maximum a-posteriori probability (MAP) adaptation from 150 hours of speech in telephone condition. The vocabulary contains 5,782 words. A 3-gram language model (LM) was obtained by adapting a basic LM with the transcriptions of the DECODA train set. The ASR word error rate (WER) is 33.8% on the train, 45.2% on the development, and 49.5% on the test subset of the corpus. These high error rates are mainly due to speech disfluencies and to adverse acoustic environments for some dialogues (calls from train stations, noisy and crowded streets with mobile phones...). The same corpus was recently processed with a novel ASR system (Rousseau et al., 2014) with advanced neural components that performed very well in international competitions. WER of 33.8% for the development and 34,5% for the test sets were obtained, still showing rather frequent errors due to above mentioned unpredictable real-life environments and situations. Nevertheless, the impact on the selected topic specific words does not seem to prevent obtaining classification results close with the ones observed with manual transcriptions. In fact, as reported in (Morchid et al.,

2014b), a maximum 89.7% accuracy was obtained with the same word set and ASR system as in this chapter. Such an accuracy is close to the reported best 92.2% accuracy obtained with manual transcriptions. Even if the optimal choice on the development set does not lead to the best accuracy with the test set, it is evident that the features used in this chapter are adequate to evaluate and compare the effects of the different feature denoising methods considered in this chapter. The manual transcriptions (TRS) are also used to show the maximum classification performance that could be achieved (*i.e.* best condition if no transcription error has been made by an ASR system). It is worth noticing that a similar WER has been reported on a similar corpus type (conversations from call-center) (Garnier-Rizet et al., 2008).

### 2.4 **Results and Discussion**

Table 2.2 compares the best accuracies of the different architectures and features proposed in this chapter. Although AE<sub>ASR</sub> and SAE are simpler compared to others and are trained in a completely unsupervised way, they are among the more robust evaluated approaches. This can be explained by the fact that these neural networks are both able to remove an important portion of the erratic noise contained in the automatically transcribed documents. However, both methods cannot achieve the performance obtained with the original clean corpus. This means that there is still another kind of noise introduced by the automatic transcription step that cannot be compensated with noisy examples only. The smaller relative improvement obtained by an AE on clean data (0.83%) compared to the improvement on the data from ASR (3.76%), as shown in Table 2.2, pointed out that there is multiple factors for the noise present in ASR data and one of them is the automatic transcription process. The accuracy of the SDAE approaching 83.2% suggests the following observations. Table ?? shows that SDAE accuracy is just 0.9 point under the accuracy obtained with clean documents (TRS). This result shows that reconstructed feature vectors are still perturbed by a residual noise resulting in a decrease of classification performance with respect to using TRS features in  $AE_{TRS}$ .

The second is that introducing a form of supervision with manual transcriptions in the denoising process improves the quality of the learnt representation and lets the model remove a large portion of the type of noise that is the only one affecting manual transcriptions. Thirdly, the relatively poor results reported in Table **??**) for DDAE, DAE, FSDAE show that it is difficult to remove general- and task-dependent ASR noise at the same time. In the proposed SDAE, the first and last layers capitalize on the capacity of  $AE_{ASR}$  and  $AE_{TRS}$  to remove residual noise by reducing the more complex or abstract form of noise thanks to a cleaner latent representation. The fixed matrix during the second learning phase forces the network to denoise only the latent representations.

Overall, the best results are obtained with TDAE. With this solution, most of the noise affecting manual and ASR features is removed from the latent representations. In this way, the accuracy obtained with the ASR features equals the one achieved with TRS features, *i.e.* 84.1%. This is due to a pre-training supervision that makes it possible to remove most of the noise harmful for the classification task.

The latent representations from different environments let both TDAE and SDAE be-

ing able to construct a robust mapping representation for two reasons. First, the latent information carried by a document is related to the semantics of this document. Both clean and noisy versions of documents should share the same semantic independently of the actual words used. Secondly, the latent representation compresses the documents and force them to keep only the relevant information. In these latent spaces, noisy and clean versions of the same document have more in common than their visible form which makes them easier to remove the noise. The improvement obtained by the these networks compared to the very similar networks (in an unsupervised way) SAE and MLP-AE respectively shows that the supervision brought by the manual transcription is favorable for the posterior denoising process. We plan to continue the study of robust

Method	Feature	Test
employed	vector	Accuracy
MLP bottleneck	$h^2$	71.3
DDAE	$h^{(1)}$	72.5
DAE	h	74.3
FSDAE	h	76.5
TF.IDF	-	77.1
AE <sub>ASR</sub>	h	81
MLP-AE	$h^2$	81.3
SAE	$h^{(2)}$	82.0
Proposed SDAE	h	83.2
Proposed TDAE	$h^2$	84.1

Table 2.2: Best theme classification accuracy (%) observed for each set of features from ASR.

features from neural networks of spoken documents with novel generative adversarial networks (GAN), and a first investigation is reported in chapter 6. GAN models will be a main part of my future direction due to their capability to learn complex mapping from random inputs and require few supervision.

## Part II

# Quaternion Neural Networks for Natural Language Processing

## **Chapter 3**

# **Quaternion Convolutional Neural Networks**

#### Contents

3.1	Introduction	54
3.2	Motivations	55
3.3	Quaternion algebra	57
3.4	Quaternion Convolutional Neural Networks	57
3.5	Experiments	58
3.6	Results and Discussion	60

#### Abstract

This chapter presents a novel model based on convolutional neural networks and the Quaternion hyper-complex numbers called "Quaternion Convolutional Neural Network" (QCNN). The proposed QCNN is evaluated on NLP related tasks (Theme identification task of spoken dialogues). This work has been realized during the Ph.D. thesis of Titouan Parcollet co-supervised with Georges LinarÃ's (LIA) while Titouan Parcollet was an intern at the MILA (Canada) under the co-supervision of Yoshua Bengio.

#### 3.1 Introduction

The previous chapters have presented the first research directions on Machine Learning (ML) models for NLP related applications. However, ML models, such as neural networks, employ document or signal representations based on basic low level features. Therefore, these basic representations reveal little in way of document statistical structure by only considering these features as a "bag-of-words", ignoring relations between them; moreover, the number of parameters are often large, even huge. During the last 5 years, we have proposed to remedying this weaknesses by extending the complex features based on Quaternion algebra presented in (Morchid et al., 2013) to neural networks. The effectiveness of such as algebra to represent multidimensional objects have been studied for speech recognition (Parcollet et al., 2018c,d,a, 2019; Titouan et al., 2019b), spoken language understanding (Parcollet et al., 2017; Titouan et al., 2017; Parcollet et al., 2016) and image processing (Parcollet et al., 2019); moreover, different quaternion based neural networks have been studied such as multi-layer perceptron (Parcollet et al., 2016, 2017), encoder-decoders (Titouan et al., 2017), recurrent neural networks (Parcollet et al., 2018c, 2019) and convolutional neural networks called QCNN (Parcollet et al., 2018d, 2019). One can refer to these papers for further details. The two newt chapters focus on the studies evaluating the effectiveness and efficiency of QCNN and QRNN architectures. QCNN are compared in this chapter to the realvalued CNN in a speech recognition (TIMIT) task. CNNs employ document representations based on features basic level. Therefore, these basic representations reveal little in way of document statistical structure by only considering signal, words or topics contained in the document ignoring relations between them. We propose in this chapter, to remedy this weakness by extending the complex features based on Quaternion algebra presented in (Morchid et al., 2013) to neural networks called QCNN. This original QCNN approach is based on hyper-complex algebra to take into account features dependencies in documents. Moreover, such models rely on unidimensional representations of the input information based on real numbers. Many realistic tasks require an adapted representation to fit the multidimensionality of the input features, such as pixels of an image, acoustic features, 3D models, or the different speech turns in a conversation. Therefore, traditional NNs process each component independently while a more natural way is to process each group of components as a single entity to learn both internal and contextual dependencies. Indeed, it is known that human-human conversations about specific items contain contextual relations between mentions of different speakers. In order to capture a part of these relations, it has been proposed to model a conversation with hyper-complex numbers (Parcollet et al., 2016; Morchid et al., 2013), the Quaternions, that integrate specific features for each speaker.

Quaternions are hypercomplex numbers that contain a real and three separate imaginary components, fitting perfectly to 3 and 4 dimensional input feature vectors, such as for image processing and robot kinematics (Sangwine, 1996; Pei et Cheng, 1999; Aspragathos et Dimitros, 1998). The idea of bundling groups of numbers into separate entities is also exploited by the recent capsule network (Sabour et al., 2017). Conversely to traditional homogeneous representations, capsule and quaternion networks bundle sets of features together. Thereby, quaternion neural network based models are able to code latent inter-dependencies between groups of input features during the learning process with less parameters than traditional NNs, by taking advantage of the *Hamilton product* as the equivalent of the ordinary product, but between quaternions. Quaternion neural networks (Isokawa et al., 2003; Arena et al., 1994, 1997) have been proposed to solve different tasks that involve composed entities as input features (Arena et al., 1994, 1997). More precisely, good results have been obtained in the past for theme identification of telephone conversations (Parcollet et al., 2016) using a quaternion-based multilayer perceptron (QMLP) with adapted features for each speaker. However, the QMLP used as a solution to this task does not take into consideration the external and contextual informations that can exist between different turns of a dialogue.

#### 3.2 Motivations

A major challenge of current machine learning models is to obtain efficient representations of relevant information for solving a specific task. Consequently, a good model has to efficiently code both the relations that occur at the feature level, such as between the Mel filter energies, the first, and second order derivatives values of a single time-frame, and at a global level, such as a phonemes or words described by a group of time-frames. Moreover, to avoid overfitting, better generalize, and to be more efficient, such models also have to be as small as possible. Nonetheless, real-valued neural networks usually require a huge set of parameters to well-perform on speech recognition tasks, and hardly code internal dependencies within the features, since they are considered at the same level as global dependencies during the learning. In the following, we detail the motivations to employ quaternion-valued neural networks instead of real-valued ones to code inter and intra features dependencies with less parameters.

First, a better representation of multidimensional data has to be explored to naturally capture internal relations within the input features. For example, an efficient way to represent the information composing an acoustic signal sequence is to consider each time-frame as being a whole entity of three strongly related elements, instead of a group of unidimensional elements that *could* be related to each others, as in traditional real-valued neural networks. Indeed, with a real-valued NN, the latent relations between the Mel filter banks energies, and the first and second order derivatives of a given time-frame are hardly coded in the latent space since the weight has to find out these relations among all the time-frames composing the sequence. Quaternions are fourth dimensional entities and allow one to build and process elements made of up to four elements, mitigating the above described problem. Indeed, the quaternion algebra and more precisely the Hamilton product allows quaternion neural network to capture these internal latent relations within the features of a quaternion. It has been shown that QNNs are able to restore the spatial relations within 3D coordinates (Matsui et al., 2004), and within color pixels (Isokawa et al., 2003), while real-valued NNs failed. In fact, the quaternion-weight components are shared through multiple quaternion input parts during the Hamilton product, creating relations within the elements. Indeed, Figure 3.1 shows that the multiple weights required to code latent relations within



**Figure 3.1:** Illustration of the input features  $(Q_{in})$  latent relations learning ability of a quaternionvalued layer (right) due to the quaternion weight sharing of the Hamilton product (Eq. 3.5), compared to a standard real-valued layer (left).

a feature are considered at the same level as for learning global relations between different features, while the quaternion weight w codes these internal relations within a unique quaternion  $Q_{out}$  during the *Hamilton product* (right).

Second, quaternion neural networks make it possible to deal with the same signal dimension than real-valued NN, but with four times less neural parameters. Indeed, a 4-number quaternion weight linking two 4-number quaternion units only has 4 degrees of freedom, whereas a standard neural net parametrization have  $4 \times 4 = 16$ , i.e., a 4-fold saving in memory. Therefore, the natural multidimensional representation of quaternions alongside with their ability to drastically reduce the number of parameters indicate that hyper-complex numbers are a better fit than real numbers to create more efficient models in multidimensional spaces such as speech recognition. Indeed, modern automatic speech recognition systems usually employ input sequences composed of multidimensional acoustic features, such as log Mel features, that are often enriched with their first, second and third time derivatives (Davis et Mermelstein, 1990; Furui, 1986), to integrate contextual information. In standard NNs, static features are simply concatenated with their derivatives to form a large input vector, without effectively considering that signal derivatives represent different views of the same input. Nonetheless, it is crucial to consider that these three descriptors represent a special state of a time-frame, and are thus correlated. Following the above motivations and the results observed on previous works about quaternion neural networks, we hypothesize that for acoustic data, quaternion NNs naturally provide a more suitable representation of the input sequence, since these multiple views can be directly embedded in the multiple dimensions space of the quaternion, leading to smaller and more accurate models.

### 3.3 Quaternion algebra

The quaternion algebra **H** defines operations between quaternion numbers. A quaternion Q is an extension of a complex number to the hyper-complex plane defined in a four dimensional space as:

$$Q = r\mathbf{1} + x\mathbf{i} + y\mathbf{j} + z\mathbf{k},\tag{3.1}$$

where *r*, *x*, *y*, and *z* are real numbers, and 1, **i**, **j**, and **k** are the quaternion unit basis. In a quaternion, *r* is the real part, while  $x\mathbf{i} + y\mathbf{j} + z\mathbf{k}$  with  $\mathbf{i}^2 = \mathbf{j}^2 = \mathbf{k}^2 = \mathbf{i}\mathbf{j}\mathbf{k} = -1$  is the imaginary part, or the vector part. Such a definition can be used to describe spatial rotations. A quaternion *Q* can also be summarized into the following matrix of real numbers, that turns out to be more suitable for computations:

$$Q_{mat} = \begin{bmatrix} r & -x & -y & -z \\ x & r & -z & y \\ y & z & r & -x \\ z & -y & x & r \end{bmatrix}.$$
 (3.2)

The conjugate  $Q^*$  of Q is defined as:

$$Q^* = r\mathbf{1} - x\mathbf{i} - y\mathbf{j} - z\mathbf{k}.$$
(3.3)

Then, a normalized or unit quaternion  $Q^{\triangleleft}$  is expressed as:

$$Q^{\triangleleft} = \frac{Q}{\sqrt{r^2 + x^2 + y^2 + z^2}}.$$
(3.4)

Finally, the Hamilton product  $\otimes$  between two quaternions  $Q_1$  and  $Q_2$  is computed as follows:

$$Q_{1} \otimes Q_{2} = (r_{1}r_{2} - x_{1}x_{2} - y_{1}y_{2} - z_{1}z_{2}) + (r_{1}x_{2} + x_{1}r_{2} + y_{1}z_{2} - z_{1}y_{2})\mathbf{i} + (r_{1}y_{2} - x_{1}z_{2} + y_{1}r_{2} + z_{1}x_{2})\mathbf{j} + (r_{1}z_{2} + x_{1}y_{2} - y_{1}x_{2} + z_{1}r_{2})\mathbf{k}.$$
(3.5)

The Hamilton product is used in QCNNs to perform transformations of vectors representing quaternions, as well as scaling and interpolation between two rotations following a geodesic over a sphere in the  $\mathbf{R}^3$  space as shown in (Minemoto et al., 2017).

### 3.4 Quaternion Convolutional Neural Networks

This section describes the quaternion features employed as well as the convolution process.

**Quaternion internal representation.** The QCNN is a quaternion extension of well-known real-valued and complex-valued convolutional neural networks (CNN)

(He et al., 2016; Trabelsi et al., 2017). The quaternion algebra is ensured by manipulating matrices of real numbers. Consequently, a traditional 2*D* convolutional layer, with a kernel that contains *N* feature maps, is split into 4 parts: the first part equal to *r*, the second one to *x***i**, the third one to *y***j** and the last one to *z***k** of a quaternion  $Q = r1 + x\mathbf{i} + y\mathbf{j} + z\mathbf{k}$ . Nonetheless, an important condition to perform backpropagation in either real, complex or quaternion neural networks is to have cost and activation functions that are differentiable with respect to each part of the real, complex or quaternion functions for quaternion have been investigated (Xu et al., 2017) and a quaternion backpropagation algorithm have been proposed in (Nitta, 1995). Consequently, the split activation (Arena et al., 1994; Parcollet et al., 2016) function is applied to every layer and is defined as follows:

$$\alpha(Q) = \alpha(r) + \alpha(x)\mathbf{i} + \alpha(y)\mathbf{j} + \alpha(z)\mathbf{k}, \qquad (3.6)$$

with  $\alpha$  corresponding to any standard activation function.

**Quaternion-valued convolution.** Following a recent proposition for convolution of complex numbers(Trabelsi et al., 2017) and quaternions (Chase Gaudet, 2017), this chapter presents basic neural networks convolution operations using quaternion algebra. The convolution process is defined in the real-valued space by convolving a filter matrix with a vector. In a QCNN, the convolution of a quaternion filter matrix with a quaternion vector is performed. For this computation, the Hamilton product is computed using the real-valued matrices representation of quaternions. Let  $W = R + X\mathbf{i} + Y\mathbf{j} + Z\mathbf{k}$  be a quaternion weight filter matrix, and  $X_p = r + x\mathbf{i} + y\mathbf{j} + z\mathbf{k}$  the quaternion input vector. The quaternion convolution w.r.t the Hamilton product  $W \otimes X_p$  is defined as follows:

$$W \otimes X_p = (Rr - Xx - Yy - Zz) + (Rx + Xr + Yz - Zy)\mathbf{i} + (Ry - Xz + Yr + Zx)\mathbf{j} + (Rz + Xy - Yx + Zr)\mathbf{k},$$
(3.7)

and can thus be expressed in a matrix form:

$$W \otimes X_p = \begin{bmatrix} R & -X & -Y & -Z \\ X & R & -Z & Y \\ Y & Z & R & -X \\ Z & -Y & X & R \end{bmatrix} * \begin{bmatrix} r \\ x \\ y \\ z \end{bmatrix} = \begin{bmatrix} r' \\ x'\mathbf{i} \\ y'\mathbf{j} \\ z'\mathbf{k} \end{bmatrix}, \qquad (3.8)$$

An illustration of such operation is depicted in Figure 3.2.

#### 3.5 Experiments

The performance and efficiency of the QCNNs are evaluated on a phoneme recognition task on the TIMIT framework.



Figure 3.2: Illustration of the quaternion convolution

**Data-set and acoustic features of quaternions.** The TIMIT (Garofolo et al., 1993) dataset is composed of a standard 462-speaker training dataset, a 50-speakers development dataset and a core test dataset of 192 sentences. During the experiments, the SA records of the training set are removed and the development set is used for early stopping. The raw audio is transformed into 40-dimensional log mel-filter-bank coefficients with deltas, delta-deltas, and energy terms, resulting in a one dimensional vector of length 123. An acoustic quaternion Q(f, t) associated with a frequency f and a time frame t is defined as follows:

$$Q(f,t) = 0 + e(f,t)\mathbf{i} + \frac{\partial e(f,t)}{\partial t}\mathbf{j} + \frac{\partial^2 e(f,t)}{\partial^2 t}\mathbf{k}.$$
(3.9)

It represents multiple views of a frequency *f* at time frame *t*, consisting of the energy e(f,t) in the filter band corresponding to *f*, its first time derivative describing a slope view, and its second time derivative describing a concavity view. Finally, a unique quaternion is composed with the three corresponding energy terms. Thus, the quaternion input vector length is  $41(\frac{123}{3})$ .

**Connectionist Temporal Classification (CTC).** In the acoustic modeling part of ASR systems, the task of sequence-to-sequence mapping from an input acoustic signal  $X = [x_1, ..., x_n]$  to a sequence of symbols  $T = [t_1, ..., t_m]$  is complex due to:

- *X* and *T* could be in arbitrary length.
- The alignment between *X* and *T* is unknown in most cases.

Specially, *T* is usually shorter than *X* in terms of phoneme symbols.

To alleviate these problems, connectionist temporal classification (CTC) has been proposed (Graves et al., 2006). First, a softmax is applied at each timestep, or frame, providing a probability of emitting each symbol X at that timestep. This probability results in a symbol sequences representation P(O|X), with  $O = [o_1, ..., o_n]$  in the latent space O. A blank symbol '-' is introduced as an extra label to allow the classifier to deal with the unknown alignment. Then, O is transformed to the final output sequence with a many-to-one function g(O) defined as follows:

$$\begin{cases} g(z_1, z_2, -, z_3, -) \\ g(z_1, z_2, z_3, z_3, -) \\ g(z_1, -, z_2, z_3, z_3) \end{cases} = (z_1, z_2, z_3).$$
(3.10)

Consequently, the output sequence is a summation over the probability of all possible alignments between *X* and *T* after applying the function g(O). Accordingly to (Graves et al., 2006) the parameters of the models are learned, based on the cross entropy loss function. During the inference, a best path decoding algorithm is performed. Therefore, the latent sequence with the highest probability is obtained by performing argmax of the softmax output at each timestep. The final sequence is obtained by applying the function g(.) to the latent sequence.

Weight initialization is crucial to efficiently train neural networks. An appropriate initialization improves training speed and reduces the risk of exploding or vanishing gradient. A quaternion initialization is composed of two steps. First, for each weight to be initialized, a purely imaginary quaternion  $q_{imag}$  is generated following an uniform distribution in the interval [0, 1]. The imaginary unit is then normalized to obtain  $q_{imag}^{\triangleleft}$ following the quaternion normalization equation. The later is used alongside to other well known initializing criterion such as (Glorot et Bengio, 2010) or (He et al., 2015) to complete the initialization process of a given quaternion weight named *w*. More information on weight initialization are available in (Parcollet et al., 2018c).

#### 3.6 **Results and Discussion**

Results on the phoneme recognition task on the TIMIT dataset are reported in Table 3.1. It is worth noticing the important difference in terms of the number of learning parameters between real and quaternion valued CNNs. It is easily explained by the quaternion algebra. In the case of a dense layer with 1,024 input values and 1,024 hidden units, a real-valued model will have  $1,024^2 \approx 1$ M parameters, while to maintain equal input and output nodes (1,024) the quaternion equivalent has 256 quaternions inputs and 256 quaternion-valued hidden units. Therefore, the number of parameters for the quaternion model is  $256^2 \times 4 \approx 0.26$ M. Such a complexity reduction turns out to produce better results and may have other advantages such as a smallest memory footprint while saving NN models. Moreover, the reduction of the number of parameters does not result in poor performance in the QCNN. Indeed, the best PER reported is 19.64% from a QCNN with 256 feature maps and 10 layers, compared to a PER of 20.57% for a real-valued CNN with 64 feature maps and 10 layers. It is worth underlying that both model accuracies are increasing with the size and the depth of the neural network. However, bigger real-valued feature maps leads to overfitting. In fact, as shown in Table 3.1, the best PER for a real-valued model is reached with 64 (20.57) feature maps and decreasing at 128 (20.62%) and 256 (21.23). The QCNN does not suffer from such weaknesses due to the smaller density of the neural network and achieved a constant PER improvement alongside with the increasing number of feature maps. Furthermore, QCNNs always performed better than CNNs independently of the model topologies.

**Table 3.1:** Experiment results expressed in term of phoneme error rate (PER) percentage of both QCNN and CNN based models on the TIMIT phoneme recognition task. The results are from a 3 folds average. 'L' stands for number of Layers, 'FM' for number of feature maps, and 'Params' for number of learning parameters. The latter is expressed in order to be equivalent for both models. Therefore, 32FM is equal to 32FM for real numbers and 8 quaternion-valued FM

Models	Dev	Test	Params
	PER %	PER %	
ℝ-CNN-6L-32FM	22.18	23.54	3.3M
IH-QCNN-6L-32FM	22.16	23.20	0.87M
ℝ-CNN-10L-32FM	21.77	23.43	3.4M
IH-QCNN-10L-32FM	22.25	23.23	0.9M
ℝ-CNN-6L-64FM	21.19	22.12	4.8M
IH-QCNN-6L-64FM	21.44	21.99	1.2M
<b>ℝ-CNN-10L-64FM</b>	19.53	20,57	5.4M
IH-QCNN-10L-64FM	19.78	20.44	1.4M
ℝ-CNN-6L-128FM	20.33	22.14	9M
IH-QCNN-6L-128FM	20.12	21.33	2.3M
<b>ℝ-CNN-10L-128FM</b>	19.37	20.62	11.5M
IH-QCNN-10L-128FM	19.02	19.87	2.9M
ℝ-CNN-6L-256FM	20.43	22.25	22.3M
IH-QCNN-6L-256FM	19.94	20.54	5.6M
<b>ℝ-CNN-10L-256FM</b>	18.89	21.23	32.1M
<b>ℍ-QCNN-10L-256FM</b>	18.33	19.64	8.1M

With much fewer learning parameters for a given architecture, the QCNN performs always better than the real-valued one on the reported task. In terms of PER, an average relative gain of 3.25% (w.r.t CNNs result) is obtained on the testing set. It is also worth recalling that the best PER of 19.64% is obtained with just a QCNN without HMMs, RNNs, attention mechanisms, batch normalization, phoneme language model, acoustic data normalization or adaptation. Further improvements can be obtained with exactly the same QCNN by just introducing a new acoustic feature in the real part of the quaternions.

## Chapter 3. Quaternion Convolutional Neural Networks

## **Chapter 4**

# **Quaternion Recurrent Neural Networks**

#### Contents

4.1	Introduction	64
4.2	Quaternion Recurrent Neural Networks	64
4.3	Experiments	64
4.4	Results and Discussion	65

#### Abstract

This chapter introduces the Quaternion Recurrent Neural Networks (QRNN) and an extension of the Long Short-Term Memory to Quaternion algebra called QLSTM. These models are evaluated during the speech recognition task of TIMIT. This work has been realized during the Ph.D. thesis of Titouan Parcollet co-supervised with Georges Linar $\tilde{A}$ ''s (LIA).

### 4.1 Introduction

The work presented in this chapter is another extension to quaternion algebra to a realvalued neural network. The Quaternion recurrent neural network and different extensions have been proposed during the Ph.D. thesis of Titouan Parcollet and presented at different international conferences (Parcollet et al., 2018b, a, 2019). In this chapter we limit ourselves to the QRNN/QLSTM (Parcollet et al., 2018b) and an comparaison to RNN during an application of speech recognition (TIMIT). This chapter proposes then to integrate local features in a novel model called quaternion recurrent neural network (QRNN) and its gated extension called quaternion long-short term memory neural network (QLSTM). The model learns both inter- and intra-dependencies between multidimensional input features and the basic elements of a sequence with drastically fewer parameters, making the approach more suitable for low-resource applications. The effectiveness of the proposed QRNN and QLSTM is evaluated on the realistic TIMIT phoneme recognition task that shows that both QRNN and QLSTM obtain better performances than RNNs and LSTMs with a best observed phoneme error rate (PER) of 18.5% and 15.1% for QRNN and QLSTM, compared to 19.0% and 15.3% for RNN and LSTM. Moreover, these results are obtained alongside with a reduction of 3.3 times of the number of free parameters. Similar results are observed with the larger Wall Street Journal (WSJ) dataset.

### 4.2 Quaternion Recurrent Neural Networks

The QRNN is an extension of the real-valued (Medsker et Jain, 2001) and complexvalued (Hu et Wang, 2012; Song et Yam, 1998) recurrent neural networks to hypercomplex numbers. The quaternion internal representation is the same that defined in section **??**. The QRNN differs from the real-valued RNN in each learning sub-processes. Therefore, let  $x_t$  be the input vector at timestep t,  $h_t$  the hidden state,  $W_{hx}$ ,  $W_{hy}$  and  $W_{hh}$ the input, output and hidden states weight matrices respectively. The vector  $b_h$  is the bias of the hidden state and  $p_t$ ,  $y_t$  are the output and the expected target vectors. More details of the learning process and the parametrization are available on (Parcollet et al., 2018b).

### 4.3 Experiments

This Section details the acoustic features extraction and the experimental setups.

**Quaternion acoustic features.** The raw audio is first split every 10ms with a window of 25ms. Then 40-dimensional log Mel-filter-bank coefficients with first, second, and third order derivatives are extracted using the *pytorch-kaldi*<sup>1</sup> (Ravanelli et al., 2018b) toolkit and the Kaldi s5 recipes (Povey et al., 2011). An acoustic quaternion

<sup>&</sup>lt;sup>1</sup>pytorch-kaldi is available at https://github.com/mravanelli/pytorch-kaldi

Q(f, t) associated with a frequency *f* and a time-frame *t* is formed as follows:

$$Q(f,t) = e(f,t) + \frac{\partial e(f,t)}{\partial t}\mathbf{i} + \frac{\partial^2 e(f,t)}{\partial^2 t}\mathbf{j} + \frac{\partial^3 e(f,t)}{\partial^3 t}\mathbf{k}.$$
(4.1)

Q(f,t) represents multiple views of a frequency f at time frame t, consisting of the energy e(f,t) in the filter band at frequency f, its first time derivative describing a slope view, its second time derivative describing a concavity view and the third derivative describing the rate of change of the second derivative. Quaternions are used to learn the spatial relations that exist between the 3 described different views that characterize a same frequency (Tokuda et al., 2003). Thus, the quaternion input vector length is 160/4 = 40. Decoding is based on Kaldi (Povey et al., 2011) and weighted finite state transducers (WFST) (Mohri et al., 2002) that integrate acoustic, lexicon and language model probabilities into a single HMM-based search graph.

The TIMIT corpus and neural networks configurations. The training process is based on the standard 3,696 sentences uttered by 462 speakers, while testing is conducted on 192 sentences uttered by 24 speakers of the TIMIT (Garofolo et al., 1993) dataset. A validation set composed of 400 sentences uttered by 50 speakers is used for hyper-parameter tuning. The models are compared on a fixed number of layers M = 4 and by varying the number of neurons N from 256 to 2,048, and 64 to 512 for the RNN and QRNN respectively. Indeed, it is worth underlining that the number of hidden neurons in the quaternion and real spaces do not handle the same amount of real-number values. Indeed, 256 quaternion neurons output are  $256 \times 4 = 1,024$ real values. Tanh activations are used across all the layers except for the output layer that is based on a softmax function. Models are optimized with RMSPROP with vanilla hyper-parameters and an initial learning rate of  $8 \cdot 10^{-4}$ . The learning rate is progressively annealed using a halving factor of 0.5 that is applied when no performance improvement on the validation set is observed. The models are trained during 25 epochs. All the models converged to a minimum loss due to the annealed learning rate. A dropout rate of 0.2 is applied over all the hidden layers (Srivastava et al., 2014) except the output one. The negative log-likelihood loss function is used as an objective function. All the experiments are repeated 5 times (5-folds) with different seeds and are averaged to limit any variation due to the random initialization.

#### 4.4 **Results and Discussion**

We discuss in this section the results obtained with QRNNs, QLSTMs, RNNs and LSTMs on the TIMIT speech recognition tasks. The results reported in bold on tables are obtained with the best configurations of the neural networks observed with the validation set.

**Quaternion recurrent neural networks (QRNN).** The results on the TIMIT task are reported in Table 4.1. The best PER in realistic conditions (w.r.t to the best validation PER) is 18.5% and 19.0% on the test set for QRNN and RNN models respectively, highlighting an absolute improvement of 0.5% obtained with QRNN. These results

Models	Neurons	Dev.	Test	Params
	256	22.4	23.4	1M
DNINI	512	19.6	20.4	2.8M
NININ	1,024	17.9	19.0	9.4M
	2,048	20.0	20.7	33.4M
	64	23.6	23.9	0.6M
ODNINI	128	19.2	20.1	1.4M
QKININ	256	17.4	18.5	3.8M
	512	17.5	18.7	11.2M

*Table 4.1:* Phoneme error rate (PER%) of QRNN and RNN models on the development and test sets of the TIMIT dataset. "Params" stands for the total number of trainable parameters.

compare favourably with the best results obtained so far with architectures that do not integrate access control in multiple memory layers (Ravanelli et al., 2018a). In the latter, a PER of 18.3% is reported on the TIMIT test set with batch-normalized RNNs. Moreover, a remarkable advantage of QRNNs is a drastic reduction (with a factor of  $2.5\times$ ) of the parameters needed to achieve these results. Indeed, such PERs are obtained with models that employ the same internal dimensionality corresponding to 1,024 real-valued neurons and 256 quaternion-valued ones, resulting in a number of parameters of 3.8M for QRNN against the 9.4M used in the real-valued RNN. It is also worth noting that QRNNs consistently need fewer parameters than equivalently sized RNNs with an average reduction factor of 2.26 times. This is easily explained by considering the content of the quaternion algebra. Indeed, for a fully-connected layer with 2,048 input values and 2,048 hidden units, a real-valued RNN has 2,048<sup>2</sup>  $\approx$  4.2M parameters, while to maintain equal input and output dimensions the quaternion equivalent has 512 quaternions inputs and 512 quaternion hidden units. Therefore, the number of parameters for the quaternion-valued model is  $512^2 \times 4 \approx 1$ M. Such a complexity reduction turns out to produce better results and has other advantages such as a smaller memory footprint while saving models on budget memory systems. This characteristic makes our QRNN model particularly suitable for speech recognition conducted on low computational power devices like smartphones (Chen et al., 2014). QRNNs and RNNs accuracies vary accordingly to the architecture with better PER on bigger and wider topologies. Therefore, while good PER are observed with a higher number of parameters, smaller architectures performed at 23.9% and 23.4%, with 1M and 0.6M parameters for the RNN and the QRNN respectively. Such PER are due to a too small number of parameters to solve the task.

**Quaternion long-short term memory (QLSTM).** We propose to extend the QRNN to state-of-the-art models such as long-short term memory neural networks (LSTM), to support and improve the results already observed with the QRNN compared to the RNN in more realistic conditions. LSTM (Hochreiter et Schmidhuber, 1997) neural networks were introduced to solve the problems of long-term dependencies learning and vanishing or exploding gradient observed with long sequences. Based on the equations of the forward propagation and back propagation through time of QRNN

Models	Neurons	Dev.	Test	Params
	256	14.9	16.5	3.6M
Ιςτη	512	14.2	16.1	12.6M
LSIN	1,024	14.4	15.3	46.2M
	2,048	14.0	15.9	176.3M
	64	15.5	17.0	1.6M
OLCENT	128	14.1	16.0	4.6M
QL51M	256	14.0	15.1	14.4M
	512	14.2	15.1	49.9M

*Table 4.2:* Phoneme error rate (PER%) of QLSTM and LSTM models on the development and test sets of the TIMIT dataset. "Params" stands for the total number of trainable parameters.

described in Appendix (Parcollet et al., 2018b), one can easily derive the equations of a quaternion-valued LSTM. Gates are defined with quaternion numbers following the proposal of (Danihelka et al., 2016). Therefore, the gate action is characterized by an independent modification of each component of the quaternion-valued signal following a component-wise product with the quaternion-valued gate potential. Let  $f_t$ ,  $i_t$ ,  $o_t$ ,  $c_t$ , and  $h_t$  be the forget, input, output gates, cell states and the hidden state of a LSTM cell at time-step t:

$$f_t = \alpha(W_f \otimes x_t + R_f \otimes h_{t-1} + b_f), \tag{4.2}$$

$$i_t = \alpha (W_i \otimes x_t + R_i \otimes h_{t-1} + b_i), \tag{4.3}$$

$$c_t = f_t \times c_{t-1} + i_t \times tanh(W_c \otimes x_t + R_c \otimes h_{t-1} + b_c), \tag{4.4}$$

$$o_t = \alpha (W_o \otimes x_t + R_o \otimes h_{t-1} + b_o), \tag{4.5}$$

$$h_t = o_t \times tanh(c_t), \tag{4.6}$$

where *W* are rectangular input weight matrices, *R* are square recurrent weight matrices and *b* are bias vectors.  $\alpha$  is the split activation function and  $\times$  denotes a componentwise product between two quaternions. Both QLSTM and LSTM are bidirectional and trained on the same conditions than for the QRNN and RNN experiments.

The results on the TIMIT corpus reported on Table 4.2 support the initial intuitions and the previously established trends. We first point out that the best PER observed is 15.1% and 15.3% on the test set for QLSTMs and LSTM models respectively with an absolute improvement of 0.2% obtained with QLSTM using 3.3 times fewer parameters compared to LSTM. These results are among the top of the line results (Graves et al., 2013b; Ravanelli et al., 2018a) and prove that the proposed quaternion approach can be used in state-of-the-art models. We have also evaluated both QLSTMs and LSTMs with a larger and more realistic corpus to validate the scaling of the observed TIMIT results (Section 6.4). Acoustic input features are described in Section ??, and extracted on both the 14 hours subset "train-si84" and the full 81 hours dataset "train-si284" of the Wall Street Journal (WSJ) corpus. The "test-dev93" development set is employed for validation while "test-eval92" composes the testing set. Models architectures are fixed with respect to the best results observed with the TIMIT corpus (Section 6.4). Therefore both QLSTMs and LSTMs contain four bidirectional layers of internal dimension of size 1,024. Then an additional layer of internal size 1,024 is added before the output layer. The only modification in the training process compared to the TIMIT experiments reported in Section 4.4, concerns the model optimizer which is set to Adam (Kingma et Ba, 2014) instead of RMSPROP. Results are from a 3-folds average.

*Table 4.3:* Word error rates (WER %) for WSJ14h and WSJ81h. "test-dev93" and "test-eval92" are used as validation and testing sets respectively. L stands for the number of recurrent layers.

Models	WSJ14	WSJ14	WSJ81	WSJ81	Params
	Dev.	Test	Dev.	Test	
LSTM	11.2	7.2	7.4	4.5	53.7M
QLSTM	10.9	6.9	7.2	4.3	18.7M

It is worth noticing that the results on Table 4.3 are promising (Graves et al., 2013a) (WER of 11.7% on "test-dev93") and competitive with state-of-the-art but more complex models based on better engineered features (Chan et Lane, 2015)(WER of 3.8% with the 81h on "test-eval92"). Table 4.3 shows that QLSTMs outperform LSTMs in all training conditions (14h or 81h) and with respect to both the validation and the testing sets. Moreover, QLSTMs need 2.9 times less parameters than LSTMs to achieve equivalent performances. These experiments demonstrate that QLSTMs scale well to larger and more realistic speech datasets and are still more efficient than real-valued LSTMs.

## Part III

# **Ongoing Research, Future Directions & General Perspectives**

## **Chapter 5**

## **Parsimonious Neural Networks**

#### Contents

5.1	Intro	luction	72		
5.2	Parsimonious Memory Unit (PMU)				
5.3	Experiments				
5.4	Results and Discussion				
	5.4.1	Gates activity of GRU and PMU	76		
	5.4.2	Short-term dependencies from spoken dialogues	79		
	5.4.3	Long-term dependencies from 20-Newsgroups documents	80		

#### Abstract

The previous chapters chart the path of my research throughout different neural networks based topics and language processing tasks. These models compose the first step of my research in the area of machine learning for NLP applications and require a large, even huge processing time during the learning process. In this chapter I give novel promising research perspectives for faster and more efficient neural networks based models. This work is an overview of part of my futur direction on terms of domain (deep learning) and interest (find out more efficient algorithms).
### 5.1 Introduction

My activities in the field of recurrent neural networks started with the Ph.D. thesis of Mohamed Bouaziz and compose an important part of my ongoing research and this chapter presents a parsimonious RNN suitable to process small corpuses with less processing time. The processing time required to learn the models with large amount of data is huge; moreover, these applications are inclined to be executed on the client side on mobile devices for example. Therefore, ML based systems have to be efficient enough in terms of processing time and obtained performances to be acceptable for mobile devices users. This research direction aims to propose novel neural network architectures and paradigms to better handle this large amount of uncontrolled datasets alongside to maintain even improve the performances. The chapter proposes a recurrent neural network architecture that manages short- and long-term dependencies differently than an Long Short-Term Memory (Sundermeyer et al., 2012; Greff et al., 2017) (LSTM) or Gated Recurrent Unit (GRU) (Cho et al., 2014) with a single gate called "Parsimonious Memory Unit Recurrent Neural Network" (PMU). Indeed, LSTM takes into account both short- and long-term dependencies to the word properly in different contexts, but needs to process the information in different memory blocks composed of a set of cell-related gates. Even if the results of LSTM-RNNs are promising, the processing time required to treat large documents data sets is quite huge due to the different gates activation sub-processes. A recent addition to the RNN set of models called the GRU has been proposed by (Wu et al., 2017b) to address this issue and has shown good performances in several tasks such as speech recognition (Graves et al., 2013b) or machine translation (Sutskever et al., 2014). GRU is very similar to LSTM, in that it uses a combination of gates to adjust exposure from input to the hidden states. It does however not use a memory cell but opting to fully expose its state to the output and doing away with the output gate. Switching between full retention, mixed and forget mode is implemented using a reset gate r and an update gate z. This is the main drawback of the GRU in that the role and the management of the gates are not based on the relation between short- and long-term dependencies. For example, the reset of the hidden state is provided by both update and reset gates of the GRU; moreover, the role of the reset gate is to avoid the hidden state candidate but the update gate z can remove this part of the hidden state information without the reset gate r. Therefore, all hidden states share the same distribution between short- and long-term dependencies. The PMU in which every cell is an efficient integrator with a single gate *u* that makes a strong assumption to reduce the processing time and better manage the latent relations between short- and long-term dependencies: short and long-term dependencies are related, and the more the RNN-based model learns from short dependencies, the less it learns from long ones.

### 5.2 Parsimonious Memory Unit (PMU)

The proposed PMU is related to RNN and more precisely to the Gated Recurrent Unit (GRU) introduced by (Cho et al., 2014). GRU is a LSTM-like adaptive "reset" and "update" memory unit. More information about LSTM and GRU are available in the annexe. The LSTM input gate  $i_j(t)$  is replaced in the GRU by an update gate  $z_j(t)$ , and the

forget gate  $f_j(t)$  by a reset gate  $r_j(t)$  for the GRU. The main idea behind the GRU unit is that the GRU exposes the memory content at each time step and balances the previous and new memory content strictly using leaky integration. This is particularly shown during the update process of the weight matrices. For brevity and clarity we detail here only the update of the weight matrix  $w_{h_jm}$  of the hidden state  $h_j$  for the state  $x_{hj}$  (more details are available in the appendix:

$$\Delta w_{h_jm} = \alpha \ e_{h_j}(t) \ \frac{\partial h_j(t)}{\partial w_{h_jm}}$$
  
$$\frac{\partial h_j(t)}{\partial w_{h_jm}} = \frac{\partial h_j(t-1)}{\partial w_{h_jm}} \left(1 - z_j(t)\right)$$
  
$$+ z_j(t)r_j(t)\sigma'(x_{h_j}(t))h_m(t-1) .$$
(5.1)

#### Short and long-term dependencies for the GRU

One can note that the update of the weight matrix depends on both short-term dependencies  $\left(\frac{\partial h_j(t-1)}{\partial w_{h,m}}\right)$  controlled by the inverse of the update gate  $(1-z_j(t))$ , and long-term dependencies ( $\sigma'(x_{hj}) h_m(t-1)$ ) balanced by the reset  $r_i(t)$  and the update  $z_i(t)$  gates (if active). The derivative of the hidden state has a restricted behaviour. Indeed, for example, to learn from only long-term dependencies, equation (5.1) shows that the conditions  $z_i = 1$  and  $r_i = 1$  have to be jointly satisfied. The GRU makes the assumption that long and short-term dependencies are not related and the short-term dependencies are controlled only by the reset gate  $r_i(t)$ . Indeed, the reset gate  $r_i(t)$  controls directly the long-term dependencies and  $(1 - z_i(t))$  controls also the long-term dependencies throughout a non-linear transformation. Thus, the weight matrix update does not reflect this balance between  $r_i(t)$  and  $(1 - z_i(t))$ . This model is a variant of the leaky-integration unit (LIU) proposed by (Bengio et al., 2013b) and, in the case of  $r_i(t) = 1 \quad \forall t \in T$ , the GRU is a straightforward LIU. Therefore, the reset gate  $r_i(t)$  and  $(1 - z_i(t))$  play the same role since short and long-term dependencies are correlated: the more the memory unit has to learn from short-term dependencies, the less the longterm dependencies are required to evaluate a relevant current hidden state. Based on this assumption, we have:

$$z_{j}(t) \simeq r_{j}(t)$$

$$\frac{\partial h_{j}(t)}{\partial w_{h_{j}m}} = \frac{\partial h_{j}(t-1)}{\partial w_{h_{j}m}} \left(1 - z_{j}(t)\right)$$

$$+ z_{j}(t)^{2} \sigma'(x_{hj}(t)) h_{m}(t-1) .$$
(5.3)

Equation (5.3) allows the GRU to maintain a balance between short ( $r_j$  in  $\sigma'$ ) and long ( $z_j$  and  $(1 - z_j)$ ) term dependencies. We propose a novel memory unit called "Parsimonious Memory Unit" (PMU) that addresses these three main issues of the GRU alongside reducing the processing time (Parsimonious):

• controlling uniformly for forward and update phases, the short-term dependen-



**Figure 5.1:** The graphical illustration of the (a) GRU and the proposed (b) PMU. The u represents the self-balanced PMU gate and h and  $\tilde{h}$  are the activation and the new activation state for the PMU. Long and short-term dependencies are also represented in the graphics with a single gate u for the PMU and both r and z gates for the GRU.

cies by evaluating these term dependencies with a same gate ( $r_j$  for previous hidden state and  $z_j(t)$  for the hidden state candidate),

- taking into consideration the relation between short- and long-term dependencies  $(z_i(t) \simeq r_i(t))$ ,
- managing short and long-term dependencies with dedicated hidden neurons.

The proposed PMU depicted in Figure 5.1-(b) is based on the assumption that the more the memory unit learns from short-term dependencies  $r_j$  and the less it needs to learn from long-term dependencies  $z_j$ . Let us describe how the activation of the *j*-th hidden unit  $h_j$  of the PMU is computed. First, the PMU gate  $u_j$  is computed as follows:

$$u_j(t) = \sigma(x_{uj}(t))$$
  
$$x_{uj}(t) = \sum_m s_{u_jm} v_m(t) + w_{u_jm} h_m(t-1) .$$

The activation of the proposed unit  $h_i$  is defined as:

$$h_{j}(t) = (1 - u_{j}(t))h_{j}(t - 1) + \tilde{h}_{j}(t)$$

$$\tilde{h}_{i}(t) = \tanh(x_{i})$$
(5.4)

$$x_{hj} = \sum_{m} s_{hjm} v_m(t) + w_{hjm} h_m(t-1) u_j(t) .$$
(5.5)

Finally, the output unit  $y_k$  is computed by:

$$y_k(t) = \tanh(x_k(t))$$
  
$$x_k(t) = \sum_m w_{km} h_m(t-1)$$

The hidden state matrix  $w_{h_im}$  is update as follows:

$$\begin{split} \Delta w_{h_jm} &= \alpha \; e_{h_j}(t) \; \frac{\partial h_j(t)}{\partial w_{h_jm}} \\ \frac{\partial h_j(t)}{\partial w_{h_jm}} &= \frac{\partial h_j(t-1)}{\partial w_{h_jm}} \left(1 - u_j(t)\right) \\ &+ \tanh'(x_{h_j}(t))u_j(t)h_m(t-1) \;. \end{split}$$
(5.6)

More details about PMU are available in (Morchid, 2018).

#### Short and long-term dependencies for the PMU

Equations (5.4), (5.5) and (5.6) are homogeneous with regards to the representation of short and long-term dependencies and the update process described in equation (5.6) of the weight matrix of the hidden state is detailed thereafter:

$$u_j(t) \to 1 \Rightarrow \frac{\partial h_j(t)}{\partial w_{h_jm}} \to h_m(t-1)\sigma'(x_{hj})$$
 (long-term) (5.7)

$$u_j(t) \to 0 \Rightarrow \frac{\partial h_j(t)}{\partial w_{h_jm}} \to \frac{\partial h_j(t-1)}{\partial w_{h_jm}}$$
 (short-term) (5.8)

In a same manner, the hidden state of the proposed PMU described in equations (5.4), (5.5), behaves during the forward pass as follows:

$$u_j(t) \to 1 \Rightarrow h_j(t) \to \tilde{h}_j(t)$$
 (long-term) (5.9)

$$u_j(t) \to 0 \Rightarrow h_j(t) \to h_j(t-1) + \beta_j$$
 (short-term) (5.10)

Table 5.1 presents how the short or long-term dependencies are considered with respect to the *u* gate of the PMU ( $\beta_j = \sigma(\sum_m s_{h_jm} v_m(t))$ ). All intermediate-term dependencies (1 <  $u_j$  < 0) give a balance between short and long-term memory.

*Table 5.1: Hidden state* **derivative** *behaviour for exclusive short or long-term dependencies for the* **PMU***.* 

Dependencies	Pass	u <sub>j</sub>	Value
Long-term (5.4)-(5.5)	Forward	1	$\tilde{h}_j(t)$
Short-term (5.4)-(5.5)	Forward	0	$h_j(t-1) + \beta_j$
Long-term (5.6)	Backward	1	$\sigma'(x_{hj})h_m(t-1)$
Short-term (5.6)	Backward	0	$\frac{\partial \hat{h}_j(t-1)}{\partial w_{h_im}}$

The difference between the classical GRU and the proposed PMU is observed when short-term dependencies are required for the hidden state during the forward pass (second column in these tables) with an additional value ( $\beta_j$ ). This value is not related to the hidden state and is evaluated with the input vector ( $v_m(t)$ ) and the input-to-hidden matrix ( $s_{h_im}$ ). Therefore, the short-term dependencies are not affected by  $\beta_j$ .

The PMU requires only one gate viewed as a balance between short and long term dependencies. Moreover, the PMU manages short and long-term dependencies with dedicated neurons.

# 5.3 Experiments

**Data-sets and metrics.** The DECODA corpus as well as on the Automatic Speech Recognition (ASR) system employed are available in sections 2.3.1 and 2.3.2. The categorization task of the 20-Newsgroups (Xu et al., 2013) dataset is employed to exhibit long-term dependencies. This corpus is described in the official website<sup>1</sup>. The same metric is employed to make the comparison between the results obtained by the proposed PMU-RNN model in "long-term" (20Newsgroups) and "short-term" (DECODA) dependencies easier. To evaluate the effectiveness of the proposed method, the authors in (Albishre et al., 2015; Srivastava et al., 2013; Bouallegue et al., 2014; Morchid et al., 2014a) used the accuracy. More information are available in (Van Asch, 2013).

**Recurrent Neural Networks Setup.** The RNN, LSTM, GRU and the proposed PMU are composed of 3 layers: input layer *s* of the size of the discriminative vocabulary (Morchid et al., 2014c) (50 words for each class gives 166 words for DECODA and 908 for the 20-News group data set), a hidden layer *h* of size varying from 10 to 300 for all RNN and an output layer  $y_k$  with a size equal to the number of classes (8 for DECODA and 20 for the 20-Newsgroups respectively).

# 5.4 **Results and Discussion**

The short and long-term dependencies are measured in the gates activity (update and reset gates for GRU and the gate of PMU) and depicted in Section 5.4.1. Section 5.4.2 and Section 5.4.3 compare RNN, LSTM, GRU and the proposed PMU in terms of accuracy and time processing.

#### 5.4.1 Gates activity of GRU and PMU

This section presents both GRU and PMU gates activity to demonstrate that the role of GRU's gates is similar for all hidden neurons while the PMU gates behave differently across its hidden neurons. The optimal point corresponds to the number of hidden neurons required to reach the best performances for short-term (Table 5.2) and long-term (Table 5.3) dependencies.

#### Short-term gates activity from a small corpus of spoken dialogues

Figure 5.2 a) presents the gates activity of a GRU-RNN with 8 hidden states <sup>2</sup> in the hidden layer. The dashed curves represent the activity of the update gate z and the plain black curves the reset gate r of the GRU-RNN.

<sup>&</sup>lt;sup>1</sup>http://qwone.com/~jason/20Newsgroups/

<sup>&</sup>lt;sup>2</sup>The term "hidden state" is employed even if the GRU and PMU have a hidden state rather than a hidden state as the LSTM.



*Figure 5.2:* An example a) of the update (*z* black dashed curves) and reset (*r* black curves) gates activity and b) of the PMU gate u activity for short-term-dependencies for short-term dependencies from the Decoda corpus for each iteration (from 0 to 500 X-axis) for the 8 memory blocks (hidden units).

One can easily point out that the update gate *z* converges to 0 and then let the reset *r* gate to manage both long- and short-term dependencies. Therefore, the reset gate *r* controls the short-term dependencies and the long-term dependencies are avoided. This is also mainly due to the small size of the training corpus that contains 740 documents alongside with the small number of themes (or classes) 8. Thus the prior for a given sequence to be met is quite high. Figure 5.2 a) underlines that the update gate  $z \simeq 0$  after 500 iterations and  $z \simeq 0.2$  for the optimal point (real test accuracy). The reset gate  $r \simeq 0.5$  regardless of the number of iterations. Figure 5.2 b) shows the *u* gate activity of the proposed PMU-RNN for the 8 hidden neurons. For convenience and readability, part of the neurons (3) are in dashed curves and the others in plain curves. It is worth emphasizing that the role of short and long-term dependencies

is split between the 8 gates. Indeed, the Figure 5.2 b) shows that 3 dashed curves of the gates u of the PMU-RNN hidden layer move down and are devoted to short-term dependencies (see Table 5.1) while the 5 others move up and code the long-term dependencies. The proposed PMU allows the RNN to separate well the learning process of short- and long-term sequences in different distinct hidden states (3 for short-term and 5 for long-term).



*Figure 5.3:* An example a) of the update (*z* black dashed curves) and reset (*r* black curves) gates activity for long-term dependencies and b) of the PMU gate u activity for long-term-dependencies from the 20-Newsgroups corpus for each iteration (from 0 to 500 X-axis) for the 8 memory blocks (hidden units).

#### Long-term gates activity from a large microblogging data set

The activity for the 8 hidden states of the two z and r gates of a GRU-RNN during the learning process for a given (randomly selected) document is detailed in Figure 5.3 a). The dashed curves represent the activity of the update gate z and the plain black

curves the reset gate r of the GRU-RNN. We can first underline that the curves follow the same trend as those for short-term dependencies of spoken dialogues corpus. Nevertheless, the curves are steeper and z converges such as in the case of short-term dependencies from the DECODA corpus, to 0. The reset r gate also manages as in the case of DECODA corpus both long- and short term dependencies. Consequently, the reset gate r controls the short-term dependencies and the long term dependencies are avoided. Figure 5.3 b) shows the u gate activity of the proposed PMU-RNN for the 8 hidden neurons. For convenience and readability part of the neurons (4) is in dashed curves and the others in plain ones. It is worth emphasizing that the role of short and long-term dependencies are well-split between the 8 PMU gates. Indeed, figure 5.3 b) shows that 4 dashed gates u of the PMU-RNN hidden layer move down and are devoted to short-term dependencies (see Table 5.1) while the 4 others move up and code long-term dependencies. The PMU allows the RNN to split the learning process of short and long-term sequences in different distincts hidden states better (4 for short-term and 4 for long-term).

#### 5.4.2 Short-term dependencies from spoken dialogues

This section studies the performances of RNN-based models during a theme identification task of spoken dialogues from the small DECODA corpus with a small training data set that tends to mostly reveal short-term dependencies; moreover this corpus contains a small number of close themes (8) (Morchid et al., 2014c, 2016). We compare RNN, LSTM, GRU to the proposed PMU in terms of accuracy and processing time required for learning. Table 5.2 sums up the accuracies and processing times with the confidence interval (CI).

#### Accuracies on a theme identification of spoken dialogues

We underlines first that the best accuracies observed for RNN, LSTM, GRU and PMU are equal (88.6%) but the RNN reaches the lowest accuracy (84.6%) with the development data set. Nonetheless, Table 5.2 shows that the accuracy obtained in **real conditions**<sup>3</sup> on the test data set is obtained with the PMU (82.3%) which is better than the performance of the RNN (81.9%) and the LSTM (81.7%). The GRU which is the most related model to the proposed PMU-RNN, real accuracy on the test data set 80.4% is 1.9 point under the PMU one. The best configuration point is close to the maximum for the test set in the case of the RNN. Indeed, the gap for the PMU between the maximum and optimal configuration points is 1 (36 for the maximum and 37 for the optimal) and is lower than the gap for the RNN (gap=2).

#### PMU vs. other RNNs in terms of processing time

We have noticed during the experiments that the processing time observed for the RNNs (RNN, LSTM, GRU and PMU) follows the same trend as the number of iterations. Nevertheless, the high gap between the maximum of the RNN (500 iterations) and the others (roughly 250 iterations) is not observed for the processing time curves

<sup>&</sup>lt;sup>3</sup>The best operating point (number of neurones in the hidden layer) estimated on the development set is applied to the test set.

(about 200 seconds for RNN, LSTM and GRU). Moreover, the maximum PT required for the proposed PMU is 174 with a gain of more than 20% compared to the GRU maximum PT. An interesting remark regarding these curves is that the PMU is well adapted to small data sets (requiring a low number of neurons in the hidden layer and a small number of basic operations). These observations are confirmed in Table 5.2. Moreover, the best accuracy reached after the PMU one is 81.9% with the mere RNN. The PMU require only 23 sec. (for the RNN) to reach this accuracy, that represents only one quater of the PT of the RNN (83 sec.).

Neural	hidden	Test	Number of	Proc. time	
model	size	acc. %	iterations	in seconds	
RNN	38	81.9 (±4.16)	500	83	
LSTM	25	81.7 (±4.18)	162	69	
GRU	16	80.4 (±4.29)	211	40	
PMU	37	82.3 (±4.13)	123	37	

*Table 5.2:* Sum up of accuracies ( $\pm$  confidence interval), number of iterations and processing time (seconds) with the DECODA corpus.

#### 5.4.3 Long-term dependencies from 20-Newsgroups documents

The high number of documents (20,000) alongside with the small number of subjects (20 classes) discussed lead to long-term dependencies between words and topics in the 20-Newsgroups corpus. The next sections evaluate the RNNs by varying the number of neurons in the hidden layer from 10 to 300 with a step of 5 in terms of accuracy and time processing. Table 5.3 sums up the performances in terms of accuracy and processing time.

#### Classification accuracies on a large data set of documents

The curves of gated-RNNs are more stable with a gap between the lowest and highest accuracies equal to about 3 points for both LSTM, GRU and PMU. The number of hidden neurons required to reach the best accuracy on the development data set is small for the PMU (155) compared to those for the RNN (290), LSTM (230) and GRU (215). Nonetheless the real accuracy on the test data set is obtained with the LSTM and GRU (75.0) but is close to the one from the PMU (74.9%) which is better than the performance of the RNN (74.4%). The gain observed for the gated RNNs is more than 5 points compared to the RNN. The large data set allows the RNNs to not observe the drawback related to the gap between the real and maximum accuracies reached on the test data set. Table 5.3 sums up the real accuracies for each RNN. The second column of Table 5.3 depicts the number of required neurons from the hidden layer to reach the optimal configuration on the development data set. During the experiments, the number of neurons varies from 10 to 300 with a step of 5 and Table 5.3 shows that a large number of neurons are needed to reach this optimal point (higher of 200 neurons for RNN, LSTM and GRU). Nonetheless, the proposed PMU reaches its best accuracy on the development data set with fewer neurones (155) and shows its capability to

better code long-term dependencies in fewer hidden states.

# PMU vs. others RNNs in terms of processing time for the classification task of 20-Newsgroups

It is worth pointing out that the maximum processing time (PT) of the RNN, LSTM and GRU (more than 5,000 sec.) is 20% higher than the maximum PT of the PMU (3,980 sec.). We can remark that the minimum PT of the PMU is 156 seconds and quite low compared to the minimum of the RNN (272 sec.), GRU (410 sec.) and LSTM (545 sec.) with a gain of more than 43%. Moreover, the PMU has a flat curve compared to the other RNNS with a small gap between the minimum and maximum PT of 3,824 in regards with those of the RNN (4,815), LSTM (5,215) and the GRU (5,165). Therefore, the proposed PMU is more stable in relation to the variation of the hidden size, and thus, to the size of the corpus (more documents requires more neurons in the hidden layer to well-code the latent relations between the input features). The last column of Table 5.3 shows the PT needed to obtain the real test accuracies. We can easily remark that the best PT is observed in real conditions for the test data set since the PMU-RNN is employed (2, 282 sec.). The RNN, LSTM and GRU obtain worse performances (more than 5, 100 sec.) than the PMU. The proposed Parsimonious Memory Unit allows

*Table 5.3:* **Sum up** *of the classification task of* **20-Newsgroups** *corpus performances in terms of accuracies and processing time.* 

Neural	hidden	Test	Number of	Proc. time
model	size	acc. %	iterations	in seconds
RNN	290	74.4 (±0.98)	500	5,053
LSTM	230	<b>75.0</b> (±0.98)	480	5,554
GRU	215	<b>75.0</b> (±0.98)	415	5,127
PMU	155	74.9 (±0.98)	250	2,282

the RNN to reach equivalent even better accuracies than state-of-the-art GRU- or LSTM-RNN with less processing time and a smaller hidden layer size. Moreover, the PMU better-manage short and long-term dependencies. Indeed, each hidden neuron deals with a particular aspect of the term dependencies (short or long) which is not the case for the GRU (same gate activity for all hidden units). The experiments underline that the number of hidden units required for learning is correlated with the size of the corpus and a huge number of hidden units lead to model over-fitting for small datasets (DECODA). This phenomenon is not observed since the data-set is large enough to require a large number of hidden units to code internal statistical dependencies (20Newsgroups). Overall, the proposed PMU-RNN obtains better performances on the language modeling task than the other RNNs with less processing time; moreover, the PMU requires the same period for training than the straightforward RNN while the PMU manages better both short- and long-term dependencies.

# Chapter 6

# **Generative Adversarial Networks**

#### Contents

<b>6.1</b>	Introduction	84
6.2	Generative neural models	84
6.3	Experiments	86
6.4	Results and Discussion	88

#### Abstract

This chapter presents the first effort on studying Generative Adversarial Networks (GAN) for NLP. GAN is a dedicated sud domain of deep learning and attracts a wide range of AI researchers. We present in this chapter how GAN based models can allow the NLP system to extract from ASR transcripts robust features based on human transcribed documents (TRS) to better predict the most related theme of a given spoken dialogue.

### 6.1 Introduction

The work presented in this chapter reflect my recent interest in generative models to extract robust representations of noisy documents. The models proposed here is the logical continuation to autoencoders studied in chapter 2 and more precisely with the TSDAE et SDAE for NLP. But NLP neural based systems are powerful to express relevant content when the spoken documents are recorded in controlled conditions. These systems are also limited by the quality of transcripts from the automatic speech recognition system (ASR). In chapter 2, we have proposed to use the knowledge available at training time through the TRS transcriptions to enhance the input representation of the ASR versions. This enhancement is made possible with the use of stacked and deep stacked auto-encoders to learn a static mapping that projects the ASR representation to the TRS latent space. Based on the promising results observed with this approach we have proposed to further investigate the distillation of the TRS knowledge to the ASR representation with the recent generative adversarial networks (GAN). GANs are an very active field of research and offer an interesting approach that focuses on a gametheoretic method to train a generative model (Goodfellow et al., 2014). Numerous architectures have been proposed to address various tasks (Radford et al., 2015; Wu et al., 2017a). From a simplified perspective GANs are commonly used to learn a mapping from a random noise space to a target one, making it possible to generate new unseen samples. In NLP tasks the noise space is commonly replaced with a well defined input representation such as text written in a specific language for neural machine translation (Yang et al., 2017). Then GANs project this latent representation to a different target language. In the task of theme identification of telephone conversations investigated, we consider the latent ASR transcription as the noise space and the TRS versions as the target one. After training the model is expected to enhance the ASR latent representation with TRS knowledge to further improve the results when classifying the documents. We propose a task adapted model called Machine-to-Human GAN (M2H-GAN) by merging the GAN with a semi-supervised GAN (SGAN) to better represent and classify telephone conversations. This chapter Introduces first a new GAN architecture called M2H-GAN to efficiently map the automatically transcribed representation of conversations, to a latent representation of their manually transcribed version and then compares the classification accuracy obtained with this new representation to other methods on a theme identification of telephone conversations task (Section 6.3).

### 6.2 Generative neural models

We propose a GAN merged with a semi-supervised SGAN to allow a projection of an automatically transcribed document, to its manual transcription representation with the Machine-to-Human GAN (M2H-GAN).

#### **Generative Adversarial Networks**

In a generative adversarial network (Goodfellow et al., 2014), two neural networks are trained in opposition. First, a generator G outputs a fake object named  $\tilde{x}$  from an input

random noise vector *z*:

$$\tilde{x} = G(z) \tag{6.1}$$

Then a discriminator D receives alternatively a true sample x or a fake one  $\tilde{x}$  from G, and outputs a probability distribution of the input being a fake or not. During training, D tries to maximize the log-likelihood of the correctly assigned source:

$$L = \mathbb{E}[\log p(\operatorname{real}|x)] + \mathbb{E}[\log p(\operatorname{fake}|\tilde{x})]$$
(6.2)

In the same manner G is trained to fool D by minimizing the second term of Eq. 6.2. Indeed, reducing the probability of correct classification of fake inputs increases the generating capability of G.



*Figure 6.1:* Illustration of the M2H-GAN architecture at training (top) and testing (bottom) time. Red and blue lines show the ASR and TRS representation signal. Note that the output of the generator G goes from red to blue during the training phase.

Auxiliary and semi-supervised GANs (Mirza et Osindero, 2014; Odena, 2016) have been proposed to take into consideration the labels in both the generator and the discriminator to drive the generation process toward a specific class. In an SGAN, *D* is trained to determine if the input signal is fake and select the most suitable label. Consequently the output dimension of *D* is of size N + 1 with *N* being the number of classes and +1 representing the *fake* case. The loss function remains unchanged. SGANs use labels to add a condition on the generation process, making it possible to generate samples of a specific class, such as *car* or *bird* for image generation.

#### Machine-to-Human representation with generative models

We propose to merge the initial GAN with its semi-supervised version SGAN in a model named Machine-to-Human GAN(M2H-GAN). An overview of the M2H-GAN architecture is depicted in Figure 6.1. In M2H-GAN,  $\tilde{x}$  is the generated representation of an automatically transcribed document (ASR) from *G* and *x* is the "clean" TRS version of the same sample. *D* is trained to determine if the input has been generated, or belongs to a certain class (SGAN), and thus contains N + 1 output neurons. Consequently *G* is jointly trained to map the ASR transcripts to a latent TRS and "clean" space in order to fool the discriminator. Unlike for SGAN, the generator of M2H-GAN does not have access to the label due to the fact that conversations classes are unknown during the testing phase. This modification allows the discriminator to have more room to discover if an input is fake or not, making it more powerful. As a consequence the generator must create a more convincing representation of the ASR signal, and receives gradients according to the label, without any conditioning on the input. An overview of M2H-GAN is depicted in Figure 6.1.

### 6.3 Experiments

This section introduces the theme classification of telephone conversations task with the DECODA dataset along with the proposed representation of the document.

#### Spoken conversations dataset

The corpus of spoken conversations is a set of automatically transcribed and annotated human-human telephone conversations of the Paris transportation system CCS (RATP). This corpus comes from the first version of the DECODA project detailed in section 2.3.1 and is employed to evaluate the effectiveness of the proposed M2H-GAN on a conversation theme identification task. The DECODA corpus is composed of 1,242 telephone conversations recorded during high traffic days in the capital, which is equivalent to about 74 hours of signal. The dataset was split into 8 categories or dominant themes that are detailed in Table 2.1. An example of a manually transcribed conversation of DECODA.

#### Abstract document representation with LDA

The latent Dirichlet allocation or LDA is an effective method to represent documents in an unsupervised manner, as probability distributions of hidden topics (Blei et al., 2003) in a document, and have shown their efficiency in many previous related works (Morchid et al., 2013; Parcollet et al., 2016). For the experiments described in this section LDA models are trained over the training set of DECODA following the standard hyper-parameters heuristic (Blei et al., 2003). It is important to note that two LDA models are trained with either the ASR or TRS conversations from the training sub-set of the DECODA data-set. Consequently  $\alpha = \frac{50}{T}$ , with *T* the number of topics, and  $\beta = 0.01$ . The number *T* has been previously investigated for this task in (Parcollet et al., 2016, 2017), and is set to 25. More precisely, 10 runs of the *T* = 25 LDA model are concatenated to obtain a final vector of size  $25 \times 10 = 250$  to alleviate any variations. Then every conversation is projected into the corresponding LDA space,

and is embedded in a vector of size 250.

#### **Experimental protocol**

To evaluate the effectiveness of the M2H-GAN to generate TRS-like representations of ASR transcripts we compare M2H-GAN to a GAN model on the theme classification of telephone conversations. Deep feed-forward NNs trained on TRS and ASR transcripts are used as baselines. We also compare M2H-GAN to previously investigated generative models (Janod et al., 2016). Training and testing steps are detailed in Algorithm 1, and can be summarized as follows: 1) Train GAN or M2H-GAN models; 2) Freeze the generator and train a DNN classifier from the generated features. Finally, Figure 6.1 represents the global architecture of the model.

### Algorithm 1 Training procedures.

- 1: **procedure** TRAIN GANS( $X_{trs}, X_{asr}$ )
- 2: Project  $X_{trs}$ ,  $X_{asr}$  in LDA to obtain  $Z_{trs}$ ,  $Z_{asr}$ .
- 3: Generate  $\tilde{X}_{asr}$  with G from  $Z_{asr}$ .
- 4: Train *D* and *G* based on  $\tilde{X}_{asr}$ ,  $Z_{trs}$ . (Goodfellow et al., 2014).
- 5: **procedure** TRAIN DNNS(*Xasr*)
- 6: Project  $X_{asr}$  in LDA to obtain  $Z_{asr}$ .
- 7: Generate  $\tilde{X}_{asr}$  with frozen *G* from  $Z_{asr}$ .
- 8: Train a DNN to classify  $\tilde{X}_{asr}$ .

**DNNs.** Classifiers rely on 2 hidden layers of size 256 with tanh activations, and a final *softmax* layer corresponding to the 8 themes of the DECODA dataset (Bechet et al., 2012). They are trained during 40 epochs based on the Adam optimizer (Kingma et Ba, 2014) with vanilla hyper-parameters and no regularization techniques. After training the maximum accuracy obtained on the test, along with the best result *w.r.t* to the best validation performances are saved.

**GAN.** The generator is made of 2 hidden layers of size 512 and 250 (corresponding to the size of the LDA vector) with layer-wise normalization (Ba et al., 2016) and tanh activations, while the discriminator is composed of 2 layers of 128 and 8 neurons with tanh and *sigmoid* activation functions. The discriminating labels are smoothed by being sampled from a uniform distribution bounded by [0.0, 0.7] for the valid ones and by [0.7, 1.0] for the fake ones as proposed in (Salimans et al., 2016).

**M2H-GAN.** The generator is identical to the GAN baseline. The discriminator also includes a semi-supervised classification task. Consequently the output layer is made of 9 neurons for the 8 themes of the DECODA framework plus the *FAKE* label.

Both GAN and M2H-GAN generators are trained to minimize the binary cross-entropy loss observed with the discriminator predictions on their fake generated features while their discriminators maximize the binary and traditional cross-entropy loss functions of correctly classified sources. Finally, models are trained in an adversarial manner as proposed in (Goodfellow et al., 2014) during 25 epochs with SGD, no momentum, and with a learning rate set to 0.02.

### 6.4 **Results and Discussion**

Two baselines DNN classifiers are tested on both the ASR and TRS versions of the DE-CODA corpus. Then, GAN-based approaches are trained following Algorithm 1. All the experiments are performed 10 times and averaged to alleviate variations due to the random initialization of the parameters.

Table 6.1 reports the average accuracies observed with the GAN, and the more adapted M2H-GAN approaches compared to simple DNN classifiers on the DECODA task. It is firstly important to note the difference in term of accuracy between the two baselines during the theme identification on both ASR and TRS transcripts. Indeed, while the standard deviation remains almost equal both real (*w.r.t* to the validation set) and max test accuracies are different. More precisely, the ASR-based DNN obtains a real test accuracy of 83.4% compared to 88.0% for the TRS-based DNN, representing a drop of 4.6%. This is easily explained by the high WER observed on the ASR transcriptions that alter significantly the LDA representation and the final classification performances. These results support the initial intuition that a translation of ASR documents to TRS-like representations allow us to better identify the most related theme of a spoken dialogue.

As a first step to reduce this gap ASR transcripts inputs are mapped to the TRS ones with a GAN. This approach obtains a best test accuracy of 84.1% for ASR inputs, reducing the absolute difference with TRS performances to 3.9%. The standard deviation is also lowered to 0.012 resulting in a slightly more stable model. Validation performances are altered with an average accuracy of 87.0% compared to 89.5% and 92.5% for the DNNs trained on the ASR and TRS respectively.

**Table 6.1:** Accuracies obtained by various models on the DECODA corpus. "Real Test" stands for the performances observed on the test set w.r.t to the validation set, while "Max Test" are the best results obtained. The 'Data" column gives information on the data used for training. Results are averaged over 10 runs. The standard deviation is computed over these runs and concern the "Real Test" performances.

Models	Data	Dev.	Real	Max	Std.
			Test	Test	Dev.
DNN	TRS	92.5	88.0	88.5	0.016
DNN	ASR	89.5	83.4	84.6	0.017
GAN	ASR	87.0	84.1	85.2	0.012
M2H-GAN	ASR	90.0	85.5	85.8	0.007

The Machine-to-Human mapping is then performed with the M2H-GAN. The real test accuracy is increased to 85.5% representing a absolute gain of 1.4% and 2.1% compared to the simpler GAN and DNN classifier respectively. The gap between the ASR classification performances and the TRS ones is also reduced to 2.5%. It is also worth underlying that the standard deviation is halved (0.007) in comparison to all the other models, resulting in a more robust representation of the spoken document content.

Table 6.2 shows the results observed with GAN and M2H-GAN models compared

**Table 6.2:** Accuracies obtained by proposed generative models compared to previous works on the DECODA corpus. "Real Test" stands for the performances observed on the test set w.r.t to the validation set while "Max Test" are the best results obtained. The 'Data" column informs on the data used for training. Results are averaged over 10 runs. The standard deviation is computed over these runs and concerns the "Real Test".

Models	Data	Dev.	Real	Max	Std.
			Test	Test	Dev.
AE(Janod et al., 2016)	ASR	-	81	-	-
DAE(Janod et al., 2016)	ASR	-	-	74.3	-
DSAE(Janod et al., 2016)	ASR	88.0	82.0	83.0	-
QDAE(Titouan et al., 2017)	ASR	90.0	85.2	85.2	-
GAN	ASR	87.0	84.1	85.2	0.012
M2H-GAN	ASR	90.0	85.5	85.8	0.007

to previously investigated generative models. Both GAN and M2H outperform the auto-encoders (AE), denoising auto-encoders (DAE), and deep stacked auto-encoders (DSAE) proposed in (Janod et al., 2016, 2017). Indeed, encoder and decoder are trained jointly to minimize the reconstruction error while the generator and discriminator are trained on different objectives impacting on each other. M2H-GAN also give better results than the recent quaternion-valued denoising auto-encoder (QDAE) despite the fact that the QDAE is based on a better document representation and a specific segmentation with the quaternion algebra.

# Chapter 7

# **General Perspectives**

The previous chapters depict novel research directions. Nonetheless, the efforts for real-valued neural networks (Part I) and for complex-valued neural networks (Part II) will continu throughout different Ph.D. thesis and projects. All these research directions will also be followed during the AISSPER project (ANR AAPG 2019) in collaboration with the LIUM and Orkis (from 2020 to 2023). Indeed, AISSSPER aims to provide robust NLP systems that extract relevant information from spoken documents based on neural networks based models such as QNNs and autoencoders. As scientific coordinator of the project, my expertise will be at the hart of the proposed neural networks based models. The will lead me to explore all these promising researches with different M.Sc. and Ph.D. students and PostDocs.

### **Real-valued Neural Networks**

Chapter 1 presents the parallel long short-term memory recurrent neural network. This model aims to code latent dependencies between parallel sequences. This model consider that an information from a given sequence (or stream) **s** is able to be predicted with the effort of other sequences  $\mathbf{s}'$  and  $\mathbf{s}$ . Indeed, the LSTM takes the decision to retain or forget the previous hidden state and decides how much from the hidden state candidate is required to predict the next element of the considered sequence. Nonetheless, the process of forgetting the previous hidden state and the part of the hidden state exposed to the output can be made with the collaboration of all sequences but the decision to select the best hidden state candidate (input gate) have to be made by the concerned sequence (sequence to predict). Indeed, the other sequences have not enough information to select the best candidate for the next element composing the sequence. Therefore this process has to be made by  $\mathbf{s}$ .

The works on encoder-decoder in collaboration with Killian Janod during his Ph.D. thesis presented in chapter 2 are based on mere neural networks. Nowadays novel architectures allow the neural networks based systems to better handle latent relations between spoken/transcribed features from spoken documents. Among these models Generative Adversarial Networks (GAN) and conditional GAN learn the mapping from a document to another. Therefore a perspective of this work is to employ automatically transcribed from an ASR and human transcriptions of spoken docu-

ments as "true" and "fake" version of the spoken documents content. The important interest from researchers for GAN is mostly due to their capability to learn from noise documents how to build robust representations.

# **Quaternion-valued Neural Networks**

Chapters 3 and 4 presents the Quaternion-valued neural networks (QNN). QNNs have encountered a great interest from researchers. This study have been made in collaboration with Titouan Parcollet during his Ph.D. thesis. His application for an internship at the MILA <sup>1</sup> has been selected to start an collaboration to develop the quaternion neural networks for different NLP elated tasks. The promise of QNNs has been shown in different NLP domains (speech recognition, theme identification, image processing) and different configurations (MLP, DNN, denoising auto-encoders, recurrent and convolutional neural networks). I plan to continue these works to extend these models to state-of-the-art algorithms and to evaluate these models in other domains (medical, economics, etc.).

The main issue related to Quaternion-based neural networks, is the representation of documents. In chapter 3 and 4 we have proposed to representing the document content with LDA based features for the user, the agent and the whole document. Therefore we obtain three sub-features for each topic from LDA. This representation reveals little in the way of interdependence between these sub-features; moreover, the dependance between a given topic for the user and agent is not trivial and difficult to understand. For these reasons we plan to discover novel features of document content to better code the information in the quaternion. We can base this research on image features such as the Red, Green, Blue (RGB) that code a pixel.

# Long-term Perspectives for Efficient Neural Networks

My ongoing projects are centered on recurrent neural networks (RNN). The aim of this direction is to better understood how RNNs learn to learn and how the memory is retained throughout the time. The study presented in Chapter 5 proposes a novel RNN-based model that better manages short- and long-term dependencies with less processing time required during the learning phase than GRU. The results obtained show that the PMU model is able to compute models from huge data-sets with less parameters. Therefore such algorithms may have a strong impact on artificial intelligence on mobile devices. Since the number of documents available on Internet is exponentially growing the needs for such models and algorithms become crucial for academics and industrials. Therefore I plan to extend these works during different projects involving industry. Chapter 6 shows that GANs learn from manually transcribed spoken documents robust mapping during a SLU task. The expressive power of GAN will be studied during future investigations. For example this study has underlined the instability during the learning process of both generator and discriminator networks. We plan to propose neural based architectures that focus only on maintaining the optimal balance between generator and discriminator during the learning process of a GAN.

<sup>&</sup>lt;sup>1</sup>https://mila.quebec

Part IV

# Appendix

# **Recurrent Neural Networks**

## **Basic of Recurrent Neural Networks**

Figure 1 presents the conventional RNN folded (a) and unrolled (b). The hidden state  $h_t$  is processed with different cells.



*Figure 1:* Illustration of (a) an Recurrent Neural Network (RNN) with loops and (b) an unrolled RNN for input  $x_t$  and its associated output  $y_t$  presented as a chain-like that reveals the link between sequences throughout the time.

The Recurrent Neural Network (RNN) is an extension of a straightforward feedforward neural network that handles a variable-length sequence as input by receiving a recurrent hidden state. The activation of this hidden state depends on the previous hidden state at each time-step. Therefore the activation of the *j*-th hidden state  $h_j$  at time-step *t* is computed as a state-to-state transition function  $\sigma_h$  of the current input vector  $v_t$  and the previous hidden states:

$$h_j(t) = \sigma(x_{hj}(t)) \tag{1}$$

$$x_{hj}(t) = \sum_{m} s_{hjm} v_m(t) + w_{hjm} h_m(t-1)$$
(2)

where  $x_{hj}(t)$  is the network sum at time step t,  $w_{hjm}$  (from unit m to  $h_j$ ) and  $s_{lm}$  are the state-to-state recurrent and input-to-hidden weight connection respectively;  $\sigma_h$  is the logistic sigmoid function. For brevity the biases are not considered. RNNs are able to output a probability distribution over the next element of a sequence given its current hidden state  $h_t$ . The RNNs factorize the probability of a sequence of length T into:

$$p(x_1,...,x_T) = p(x_1)p(x_2|x_1)...p(x_T|x_1,...,x_{T-1})$$

where the last element is a special end-of-sequence value. We model each conditional probability distribution with

$$p(x_t|x_1,\ldots,x_{t-1})=\phi(h_i(t))$$

where  $h_j(t)$  is evaluated with Eq.(1). (Bengio et al., 2003; Mikolov, 2012) have already proposed neural networks to model a probabilistic distribution over sequences. (Bengio et al., 1994; Hochreiter, 1998) expose the difficulty for a RNN to capture long-term dependencies due to the vanishing and exploding gradient problems. The first attempt to address this drawback of RNNs was to introduce an affine transformation followed by a simple element-wise non-linearty from gates units called Long Short-Term Memory (Hochreiter et Schmidhuber, 1997). (Cho et al., 2014) propose a new recurrent unit called Gated Recurrent Unit close to the LSTM. The next sections detail the LSTM and GRU to better underline the benefits of the proposed Parsimonious Memory Unit.

### Long Short-Term Memory (LSTM)

The Long Short-Term Memory (LSTM) (Hochreiter et Schmidhuber, 1997) network depicted in Figure 2, is a special case of RNNs (Elman, 1990). The goal of this architecture is to create an internal cell state of the network which allows it to exhibit dynamic temporal behaviour and to process arbitrary sequences of inputs, such as sequences of words (Sundermeyer et al., 2012) for language modeling, time series (Gers et al., 2001), etc. The basic unit in the hidden layer of the LSTM is a memory block that replaces the hidden unit in the RNN. The memory block allows cells to share the same gates to reduce the number of adaptive parameters.



*Figure 2:* The graphical illustration from (*Chung et al., 2014*) of a LTSM. The *i*, *f*, and o are the input, forget and output gates respectively.

This memory cell has in its core a recurrently self-connected gate called forget gate that learns to forget the previous cell state c. The cell state c is updated based on both the three previously defined gates and its current state. For brevity, the biases are not considered. The input  $i_j(t)$ , output  $o_j(t)$  and forget  $f_j(t)$  gates depend on the current input  $(v_m)$  and the previous hidden state  $h_m(t-1)$  and are computed as follows for



Figure 3: Bidirectional Recurrent Neural Network (BRNN).

the *j*-th memory block:

$$i_j(t) = \sigma(x_{ij}(t)) \tag{3}$$

with 
$$x_{ij}(t) = \sum_{m} s_{ijm} v_m(t) + w_{ijm} h_m(t-1)$$
 (4)

$$o_j(t) = \sigma(x_{oj}(t)) \tag{5}$$

with 
$$x_{oj}(t) = \sum_{m} s_{o_j m} v_m(t) + w_{o_j m} h_m(t-1)$$
 (6)

$$f_j(t) = \sigma(x_{fj}(t)) \tag{7}$$

with 
$$x_{fj}(t) = \sum_{m} s_{fjm} v_m(t) + w_{fjm} h_m(t-1)$$
 (8)

where  $\sigma$  is the logistic function;  $w_{lm}$  is the weight connection from unit m to unit l ( $w_{i_jm}$  connect unit m to gate  $i_j$ );  $s_{l_jm}$  is the state-to-state recurrent weight connection and  $\sigma$  is a logistic sigmoid function and j indexes memory blocks and the cell c of the j-th memory block is:

$$c_{j}(t) = f_{j}(t)c_{j}(t-1) + i_{j}(t)\tilde{c}_{j}(t)$$
(9)

$$\tilde{c}_{j}(t) = \tanh(\sum_{m} s_{c_{j}m} v_{m}(t) + w_{c_{j}m} h_{m}(t-1)).$$
(10)

Finally the hidden state  $h_i$  is computed as follows:

$$h_j(t) = o_j(t) \tanh(c_j(t)) \tag{11}$$

The output gate  $o_j$  controls how the memory content ( $c_j$ ) is exposed. Finally, assuming that the neural network topology contains an input, a hidden and an output layers, the output units  $y_k$  of all gates of a given unit are computed by:

$$y_k(t) = \tanh(x_k(t)) \tag{12}$$

$$x_k(t) = \sum_m s_{y_k m} v_m(t) + w_{y_k m} h_m(t-1) .$$
(13)

The backward pahse is detailed in (Hochreiter et Schmidhuber, 1997).

# **Bidirectional Long Short-Term Memory (BLSTM)**

LSTM networks use only the previous context to predict the next segment for a given sequence. Bidirectional RNN (BRNN) (Schuster et Paliwal, 1997), presented in Figure 3 can process both directions with two separate hidden layers (one for each direction). This type of RNN feeds to a same output layer fed forwarded inputs through the two hidden layers. Therefore the BRNN computes both *forward* hidden sequence  $\overrightarrow{\mathbf{h}}$  and *backward* sequence  $\overleftarrow{\mathbf{h}}$  as well as the output vector  $\mathbf{y}$  by iterating  $\overrightarrow{\mathbf{h}}$  from t = 1 to T, and  $\overleftarrow{\mathbf{h}}$  from t = T to 1:

$$\vec{h}_{j}(t) = \mathcal{H}(s_{\vec{h}_{jm}}v_{m}(t) + w_{\vec{h}_{jm}}\vec{h}_{m}(t-1))$$
(14)

$$\overleftarrow{h}_{j}(t) = \mathcal{H}(s_{\overleftarrow{h}_{j}m}v_{m}(t) + w_{\overleftarrow{h}_{j}m}\overleftarrow{h}_{m}(t-1))$$
(15)

$$y_j(t) = s_{\overrightarrow{h}y} \,\overrightarrow{h}_j(t) + w_{\overleftarrow{h}y} \,\overleftarrow{h}_j(t) \tag{16}$$

By replacing the BRNN cells with LSTM cells the Bidirectionnal LSTM (BLSTM) (Graves et Schmidhuber, 2005) is obtained. The BLSTM allows to exhibit long range context dependencies and takes advantage from the two directions structure. The output vector **y** is processed by evaluating simultaneously the two directions hidden sequences by computing the composite function  $\mathcal{H}$  in the forward ( $\vec{h}$ ) and backward ( $\vec{h}$ ) directions.

# **Gated Recurrent Unit**



*Figure 4:* The graphical illustration from (*Chung et al., 2014*) of a GRU. *r* and *z* represent the reset and update gates; *h* and *h* are the activation and the new activation state for the GRU.

The basic unit in the GRU depicted in Figure 4 is a hidden state learned during the forward and backward processes.

**Forward pass:** Reset *r* and update *z* gates are computed as:

$$r_j(t) = \sigma_r(x_{rj}(t)) \tag{17}$$

$$x_{rj}(t) = \sum_{m} w_{r_j m} h_m(t-1)$$
 and (18)

$$z_j(t) = \sigma_z(x_{zj}(t)) \tag{19}$$

$$x_{zj}(t) = \sum_{m} w_{z_j m} h_m(t-1) .$$
(20)

The activation of the proposed unit  $h_i$  is defined as:

$$h_j(t) = (1 - z_j(t))h_j(t - 1) + z_j(t)\tilde{h}_j(t)$$
(21)

$$\tilde{h}_j(t) = \sigma_h(x_{hj}) \tag{22}$$

$$x_{hj} = \sum_{m} w_{h_j m} h_m (t-1) r_j(t).$$
(23)

Finally, the output units  $y_k$  are computed by:

$$y_k(t) = \sigma_k(x_k(t)) \tag{24}$$

$$x_k(t) = \sum_m w_{km} h_m(t-1).$$
 (25)

**Backward pass:** The standard error objective function based on the target output  $t_k$  is defined as for the PMU. In the same way as for the PMU unit, the weight changes  $\Delta w_{y_k m}$  for output units are processed as:

$$\Delta w_{y_k m} = \alpha \ \sigma'_k(x_k(t)) \ e_k(t) \ h_m(t-1) \ . \tag{26}$$

The weight changes for the the reset (l = r) and the update (l = z) gates and the hidden state (l = h) are:

$$\Delta w_{lm} = \alpha \ e_{h_j}(t) \ \frac{\partial h_j(t)}{\partial w_{lm}}$$
(27)

and its internal error state  $e_{h_i}(t)$  is:

$$e_{h_j}(t) = \sum_k \sigma'_k(x_k(t)) w_{h_j l} e_k(t) .$$
(28)

To calculate the partial  $\frac{\partial h_j(t)}{\partial w_{lm}}$  from Eq.(27), we differentiate Eq.(21) and obtain a sum of three terms:

$$\frac{\partial h_j(t)}{\partial w_{lm}} = \frac{\partial h_j(t-1)}{\partial w_{lm}} (1 - z_j(t)) + z_j(t) \frac{\partial \tilde{h}_j(t)}{\partial w_{lm}} + \frac{\partial z_j(t)}{\partial w_{lm}} (\tilde{h}_j(t) - h_j(t-1)).$$
(29)

Differentiating the forward pass equations (17), (18), (19), (20), (21) and (23) for r, z and h we can substitute the unresolved partials and split the expression on the right hand

of equation (29) into three separate equations for the reset (l = r), update (l = z) gates and for the hidden state (l = h):

$$\begin{aligned} \frac{\partial h_j(t)}{\partial w_{r_jm}} &= \frac{\partial h_j(t-1)}{\partial w_{r_jm}} (1-z_j(t)) + z_j(t) \frac{\partial \tilde{h}_j(t)}{\partial w_{r_jm}} \\ &+ \frac{\partial z_j(t)}{\partial w_{r_jm}} (\tilde{h}_j(t) - h_j(t-1)) \end{aligned}$$

with  $\frac{\partial z_j(t)}{\partial w_{r_jm}} = 0$  and  $\frac{\partial \tilde{h}_j(t)}{\partial w_{r_jm}} = \frac{\partial \sigma_h(x_{hj}(t))}{\partial w_{r_jm}}$ , Therefore  $\frac{\partial h_j(t)}{\partial w_{r_jm}} = \frac{\partial h_j(t-1)}{\partial w_{r_jm}} \left(1 - z_j(t)\right)$   $+ z_j(t) \frac{\partial \sigma_h(x_{hj}(t))}{\partial x_{hj}} \frac{\partial x_{hj}}{\partial r_j(t)} \frac{\partial r_j(t))}{\partial x_{rj}(t)} \frac{\partial x_{rj}(t))}{\partial w_{r_jm}}$ 

finally for the reset gate *r*:

$$rac{\partial h_j(t)}{\partial w_{r_jm}} = \left(1 - z_j(t)
ight) rac{\partial h_j(t-1)}{\partial w_{r_jm}} + z_j(t)\sigma'_h(x_{hj}(t))w_{h_jm}h_m(t-1)^2\sigma'_r(x_{rt}(t))$$

in the same way with  $\frac{\partial \tilde{h}_j(t)}{\partial w_{z_jm}} = 0$ , we obtain for gate *z*:

$$\frac{\partial h_j(t)}{\partial w_{z_jm}} = \frac{\partial h_j(t-1)}{\partial w_{z_jm}} \left(1 - z_j(t)\right) + \left(\sigma_h(x_{hj}(t)) - h_j(t-1)\right) \sigma'_z(x_{zj}(t)) h_m(t-1) .$$
(30)

In a same manner with  $\frac{\partial z_j(t)}{\partial w_{h_jm}} = 0$ , we calculate the partial for the hidden state  $(l = h_j)$ :

$$\frac{\partial h_j(t)}{\partial w_{h_jm}} = \frac{\partial h_j(t-1)}{\partial w_{h_jm}} \left(1 - z_j(t)\right) 
+ z_j(t)r_j(t)\sigma'_h(x_{hj})h_m(t-1).$$
(31)

Based on equation (27) and the partials in equations (30) and (31) we can calculate the corresponding weight updates with the internal state error  $e_{h_j}(t)$  defined in equation (28):

$$\Delta w_{r_jm} = \alpha \ e_{h_j}(t) \ \frac{\partial h_j(t)}{\partial w_{r_jm}}$$
$$\Delta w_{z_jm} = \alpha \ e_{h_j}(t) \ \frac{\partial h_j(t)}{\partial w_{z_jm}}$$
$$\Delta w_{h_jm} = \alpha \ e_{h_j}(t) \ \frac{\partial h_j(t)}{\partial w_{h_jm}}$$

finally the weight matrices are updated for the reset r, update z gates, the output unit  $y_k$  and for the hidden state h with:

$$\begin{split} \Delta w_{r_jm} &= \alpha \; e_{h_j}(t) \left( \frac{\partial h_j(t-1)}{\partial w_{r_jm}} \left( 1 - z_j(t) \right) \right. \\ &+ z_j(t) \sigma'_h(x_{hj}(t)) w_{h_jm} h_m(t-1)^2 \sigma'_r(x_{rt}(t)) \right) \\ \Delta w_{z_jm} &= \alpha \; e_{h_j}(t) \left( \frac{\partial h_j(t-1)}{\partial w_{z_jm}} \left( 1 - z_j(t) \right) \right. \\ &+ \left( \sigma_h(x_{hj}(t)) - h_j(t-1) \right) \sigma'_z(x_{zj}(t)) h_m(t-1) \right) \\ \Delta w_{h_jm} &= \alpha \; e_{h_j}(t) \left( \frac{\partial h_j(t-1)}{\partial w_{h_jm}} \left( 1 - z_j(t) \right) \right. \\ &+ z_j(t) r_j(t) \sigma'_h(x_{hj}(t)) h_m(t-1) \right) \\ \Delta w_{y_km} &= \alpha \; e_k(t) \; \sigma'_k(x_k(t)) h_m(t-1) \; . \end{split}$$

# List of Figures

1	Neural Networks in some of the main dates.	12
1.1	Combination of parallel sequences: an illustrative example with 3 streams on the last 3 events : $E_t$ coressponds to the $t^{\text{th}}$ event.	29
1.2	Parallel Long Short-Term (PLSTM) neural network.	31
1.3	F1 score for n-gram and LSTM models, the two least frequent genres <i>Reality TV</i> and <i>Documentary</i> not being included.	36
2.1	Example of a dialogue from the DECODA corpus labelled by the agent as a <i>transportation card</i> issue but also containing the <i>infraction</i> theme.	39
2.2	Autoencoder model. Biases are omitted for the sake of simplicity.	40
2.3	Stacked autoencoder (SAE) architecture (Janod et al., 2016). The parame- ters of each layer are estimated as for a shallow autoencoder (left). After a pre-training step, the hidden layers are stacked to form a denoising autoencoder (right) whose parameters are further fine-tuned by global error back propagation.	41
2.4	Illustration of the proposed Supervised Deep Autoencoder (c) (SDAE) initialized with weight matrices from autoencoders (ASR (a) and TRS (b)).	44
2.5	Illustration of the proposed Task-specific Denoising Autoencoder (TDAE) with bottleneck features from MLP trained on ASR (a) and TRS	
	(b)	45
3.1	Illustration of the input features ( $Q_{in}$ ) latent relations learning ability of a quaternion-valued layer (right) due to the quaternion weight sharing of the <i>Hamilton product</i> (Eq. 3.5), compared to a standard real-valued layer (loft)	56
3.2	Illustration of the quaternion convolution	56 59
5.1	The graphical illustration of the (a) GRU and the proposed (b) PMU. The $u$ represents the self-balanced PMU gate and $h$ and $\tilde{h}$ are the activation and the new activation state for the PMU. Long and short-term dependencies are also represented in the graphics with a single gate $u$ for the	
	PMU and both $r$ and $z$ gates for the GRU	74

5.2	An example a) of the update ( <i>z</i> black dashed curves) and reset ( <i>r</i> black curves) <b>gates activity</b> and b) of the <b>PMU gate</b> <i>u</i> <b>activity</b> for <b>short-term</b> -	
	<b>dependencies</b> for <b>short-term</b> dependencies from the <b>Decoda</b> corpus for each iteration (from 0 to 500 X-axis) for the 8 memory blocks (hidden units).	77
5.3	An example a) of the update ( <i>z</i> black dashed curves) and reset ( <i>r</i> black curves) <b>gates activity</b> for <b>long-term</b> dependencies and b) of the <b>PMU gate</b> <i>u</i> <b>activity</b> for <b>long-term-dependencies</b> from the <b>20-Newsgroups</b>	
	corpus for each iteration (from 0 to 500 X-axis) for the 8 memory blocks (hidden units).	78
6.1	Illustration of the M2H-GAN architecture at training (top) and testing (bottom) time. Red and blue lines show the ASR and TRS representation signal. Note that the output of the generator <i>G</i> goes from red to blue during the training phase.	85
1	Illustration of (a) an Recurrent Neural Network (RNN) with loops and (b) an unrolled RNN for input $x_t$ and its associated output $y_t$ presented as a chain-like that reveals the link between sequences throughout the	
	time	95
2	The graphical illustration from (Chung et al., 2014) of a LTSM. The $i, f$ , and $o$ are the input, forget and output gates respectively.	96
3	Bidirectional Recurrent Neural Network (BRNN).	97
4	The graphical illustration from (Chung et al., 2014) of a GRU. $r$ and $z$ represent the reset and update gates; $h$ and $\tilde{h}$ are the activation and the	
	new activation state for the GRU.	98

# **List of Tables**

1.1	Genres Distribution for train, validation and test sets in M6 channel output.	32
1.2	Training times (in seconds) of models employed during the experiments for different telecast genres sequence sizes.	33
1.3	F1-score (%) of each n-gram and LSTM models.	33
1.4	Error rates (ER) observed for each n-gram and LSTM models for different sequence sizes.	34
1.5	Confusion matrix for the 4n-gram output: labels are shown according to their decreasing frequency as in Table 1.1.	34
1.6	Confusion matrix for the P4LSTM output: labels are shown according to their decreasing frequency as in Table 1.1.	35
1.7	F1 score (%) of n-gram and LSTM models, the two least frequent genres <i>Reality TV</i> and <i>Documentary</i> not being included.	35
2.1	Composition of the DECODA corpus.	47
2.2	from ASR.	49
3.1	Experiment results expressed in term of phoneme error rate (PER) per- centage of both QCNN and CNN based models on the TIMIT phoneme recognition task. The results are from a 3 folds average. 'L' stands for number of Layers, 'FM' for number of feature maps, and 'Params' for number of learning parameters. The latter is expressed in order to be equivalent for both models. Therefore, 32FM is equal to 32FM for real numbers and 8 quaternion-valued FM	61
4.1	Phoneme error rate (PER%) of QRNN and RNN models on the develop- ment and test sets of the TIMIT dataset. "Params" stands for the total	
4.2	number of trainable parameters	66
4.3	number of trainable parameters	67
1.0	"test-eval92" are used as validation and testing sets respectively. <i>L</i> stands for the number of recurrent layers.	68

5.1	Hidden state <b>derivative</b> behaviour for exclusive short or long-term dependencies for the <b>PMU</b> .	75
5.2	<b>Sum up</b> of accuracies (± confidence interval), number of iterations and processing time (seconds) with the <b>DECODA</b> corpus.	80
5.3	<b>Sum up</b> of the classification task of <b>20-Newsgroups</b> corpus performances in terms of accuracies and processing time.	81
6.1	Accuracies obtained by various models on the DECODA corpus. "Real Test" stands for the performances observed on the test set w.r.t to the validation set, while "Max Test" are the best results obtained. The 'Data" column gives information on the data used for training. Results are aver- aged over 10 runs. The standard deviation is computed over these runs and concern the "Real Test" performances.	88
6.2	Accuracies obtained by proposed generative models compared to previ- ous works on the DECODA corpus. "Real Test" stands for the perfor- mances observed on the test set w.r.t to the validation set while "Max Test" are the best results obtained. The 'Data" column informs on the data used for training. Results are averaged over 10 runs. The standard	
	deviation is computed over these runs and concerns the "Real Test"	89

# Bibliography

- (Albishre et al., 2015) K. Albishre, M. Albathan, & Y. Li, 2015. Effective 20 newsgroups dataset cleaning. Dans les actes de 2015 IEEE/WIC/ACM International Conference on Web Intelligence and Intelligent Agent Technology (WI-IAT), Volume 3, 98–101. IEEE.
- (Arena et al., 1997) P. Arena, L. Fortuna, G. Muscato, & M. G. Xibilia, 1997. Multilayer perceptrons to approximate quaternion valued functions. *Neural Networks* 10(2), 335–342.
- (Arena et al., 1994) P. Arena, L. Fortuna, L. Occhipinti, & M. G. Xibilia, 1994. Neural networks for quaternion-valued function approximation. Dans les actes de *Circuits* and Systems, 1994. ISCAS'94., 1994 IEEE International Symposium on, Volume 6, 307– 310. IEEE.
- (Aspragathos et Dimitros, 1998) N. A. Aspragathos & J. K. Dimitros, 1998. A comparative study of three methods for robot kinematics. *Systems, Man, and Cybernetics, Part B: Cybernetics, IEEE Transactions on* 28(2), 135–145.
- (Ba et al., 2016) J. L. Ba, J. R. Kiros, & G. E. Hinton, 2016. Layer normalization. *arXiv* preprint arXiv:1607.06450.
- (Babcock et al., 2002) B. Babcock, S. Babu, M. Datar, R. Motwani, & J. Widom, 2002. Models and issues in data stream systems. Dans les actes de *Proceedings of the twenty-first ACM SIGMOD-SIGACT-SIGART symposium on Principles of database systems*, 1–16. ACM.
- (Bahdanau et al., 2014) D. Bahdanau, K. Cho, & Y. Bengio, 2014. Neural machine translation by jointly learning to align and translate. *arXiv preprint arXiv:1409.0473*.
- (Bastien et al., 2012) F. Bastien, P. Lamblin, R. Pascanu, J. Bergstra, I. J. Goodfellow, A. Bergeron, N. Bouchard, & Y. Bengio, 2012. Theano: new features and speed improvements. Deep Learning and Unsupervised Feature Learning NIPS 2012 Workshop.
- (Bechet et al., 2012) F. Bechet, B. Maza, N. Bigouroux, T. Bazillon, M. El-Beze, R. De Mori, & E. Arbillot, 2012. Decoda: a call-centre human-human spoken conversation corpus. LREC'12.
- (Bengio, 2009) Y. Bengio, 2009. Learning deep architectures for ai. *Foundations and trends*® *in Machine Learning* 2(1), 1–127.
- (Bengio et al., 2013a) Y. Bengio, N. Boulanger-Lewandowski, & R. Pascanu, 2013a. Advances in optimizing recurrent networks. Dans les actes de 2013 IEEE International Conference on Acoustics, Speech and Signal Processing, 8624–8628. IEEE.
- (Bengio et al., 2013b) Y. Bengio, N. Boulanger-Lewandowski, & R. Pascanu, 2013b. Advances in optimizing recurrent networks. Dans les actes de 2013 IEEE International Conference on Acoustics, Speech and Signal Processing, 8624–8628. IEEE.
- (Bengio et al., 2003) Y. Bengio, R. Ducharme, & P. Vincent, 2003. A neural probabilistic language model. *Journal of Machine Learning Research* 3, 1137–1155.
- (Bengio et al., 2007) Y. Bengio, Y. LeCun, et al., 2007. Scaling learning algorithms towards ai. *Large-scale kernel machines,L. Bottou, O. Chapelle, D. DeCoste, J. Weston (eds), MIT Press* 34(5).
- (Bengio et al., 1994) Y. Bengio, P. Simard, & P. Frasconi, 1994. Learning long-term dependencies with gradient descent is difficult. *IEEE transactions on neural networks* 5(2), 157–166.
- (Berthelot et al., 2017) D. Berthelot, T. Schumm, & L. Metz, 2017. Began: Boundary equilibrium generative adversarial networks. *arXiv preprint arXiv:*1703.10717.
- (Bishop et al., 1995) C. M. Bishop et al., 1995. *Neural networks for pattern recognition*. Oxford university press.
- (Blei et al., 2003) D. Blei, A. Ng, & M. Jordan, 2003. Latent dirichlet allocation. *The Journal of Machine Learning Research* 3, 993–1022.
- (Bouallegue et al., 2014) M. Bouallegue, M. Morchid, R. Dufour, M. Driss, G. Linarès, & R. De Mori, 2014. Subspace gaussian mixture models for dialogues classification. Dans les actes de *Conference of the International Speech Communication Association (IN-TERSPEECH) 2014.* ISCA.
- (Bouaziz et al., 2016) M. Bouaziz, M. Morchid, R. Dufour, G. Linarès, & R. De Mori, 2016. Parallel long short-term memory for multi-stream classification. Dans les actes de 2016 IEEE Spoken Language Technology Workshop (SLT), 218–223. IEEE.
- (Bouchekif et al., 2013) A. Bouchekif, G. Damnati, & D. Charlet, 2013. Complementarity of lexical cohesion and speaker role information for story segmentation of french tv broadcast news. Dans les actes de *Statistical Language and Speech Processing*, 51–61. Springer.
- (Bowman et al., 2015) S. R. Bowman, L. Vilnis, O. Vinyals, A. M. Dai, R. Jozefowicz, & S. Bengio, 2015. Generating sentences from a continuous space. *arXiv preprint* arXiv:1511.06349.
- (Bredin et al., 2014) H. Bredin, A. Laurent, A. Sarkar, V.-B. Le, S. Rosset, & C. Barras, 2014. Person instance graphs for named speaker identification in tv broadcast. Dans les actes de *Proceedings of Odyssey*.

- (Chan et Lane, 2015) W. Chan & I. Lane, 2015. Deep recurrent neural networks for acoustic modelling. *arXiv preprint arXiv:*1504.01482.
- (Chao et al., 2015) L. Chao, J. Tao, M. Yang, Y. Li, & Z. Wen, 2015. Long short term memory recurrent neural network based multimodal dimensional emotion recognition. Dans les actes de *Proceedings of the 5th International Workshop on Audio/Visual Emotion Challenge*, 65–72. ACM.
- (Chase Gaudet, 2017) A. M. Chase Gaudet, 2017. Deep quaternion networks. *arXiv* preprint arXiv:1712.04604v2.
- (Che et al., 2016) T. Che, Y. Li, A. P. Jacob, Y. Bengio, & W. Li, 2016. Mode regularized generative adversarial networks. *arXiv preprint arXiv:1612.02136*.
- (Chen et al., 2014) G. Chen, C. Parada, & G. Heigold, 2014. Small-footprint keyword spotting using deep neural networks. Dans les actes de 2014 IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), 4087–4091.
- (Chen, 2017) M. Chen, 2017. Minimalrnn: Toward more interpretable and trainable recurrent neural networks. *arXiv preprint arXiv:1711.06788*.
- (Chen et Jin, 2015) S. Chen & Q. Jin, 2015. Multi-modal dimensional emotion recognition using recurrent neural networks. Dans les actes de *Proceedings of the 5th International Workshop on Audio/Visual Emotion Challenge*, 49–56. ACM.
- (Chen et al., 2017) Y. Chen, J. Yang, & J. Qian, 2017. Recurrent neural network for facial landmark detection. *Neurocomputing* 219, 26–38.
- (Chihi, 1998) I. Chihi, 1998. Understanding kohonen networks. *ENSI, National School for Computer Sciences*, 1–10.
- (Cho et al., 2014) K. Cho, B. Van Merriënboer, D. Bahdanau, & Y. Bengio, 2014. On the properties of neural machine translation: Encoder-decoder approaches. *arXiv preprint arXiv*:1409.1259.
- (Chollet, 2015) F. Chollet, 2015. keras. https://github.com/fchollet/keras.
- (Chung et al., 2014) J. Chung, C. Gulcehre, K. Cho, & Y. Bengio, 2014. Empirical evaluation of gated recurrent neural networks on sequence modeling. *arXiv preprint arXiv:*1412.3555.
- (Creswell et al., 2018) A. Creswell, T. White, V. Dumoulin, K. Arulkumaran, B. Sengupta, & A. A. Bharath, 2018. Generative adversarial networks: An overview. *IEEE Signal Processing Magazine* 35(1), 53–65.
- (Dai et al., 2004) B.-R. Dai, J.-W. Huang, M.-Y. Yeh, & M.-S. Chen, 2004. Clustering on demand for multiple data streams. Dans les actes de *Data Mining*, 2004. ICDM'04. Fourth IEEE International Conference on, 367–370. IEEE.
- (Danihelka et al., 2016) I. Danihelka, G. Wayne, B. Uria, N. Kalchbrenner, & A. Graves, 2016. Associative long short-term memory. arXiv preprint arXiv:1602.03032.

- (Davis et Mermelstein, 1990) S. B. Davis & P. Mermelstein, 1990. Comparison of parametric representations for monosyllabic word recognition in continuously spoken sentences. Dans les actes de *Readings in speech recognition*, 65–74. Elsevier.
- (Donahue et Rumshisky, 2018) D. Donahue & A. Rumshisky, 2018. Adversarial text generation without reinforcement learning. *arXiv preprint arXiv:1810.06640*.
- (Dong Yu, 2011) M. S. Dong Yu, 2011. Improved bottleneck features using pretrained deep neural networks. Dans les actes de *Conference of the International Speech Communication Association (INTERSPEECH)*. International Speech Communication Association.
- (Eisenstein et Barzilay, 2008) J. Eisenstein & R. Barzilay, 2008. Bayesian unsupervised topic segmentation. Dans les actes de *Proceedings of the Conference on Empirical Methods in Natural Language Processing*, 334–343. ACL.
- (Elman, 1990) J. L. Elman, 1990. Finding structure in time. *Cognitive science* 14(2), 179–211.
- (Esteve et al., 2015) Y. Esteve, M. Bouallegue, C. Lailler, M. Morchid, R. Dufour, G. Linarès, D. Matrouf, & R. De Mori, 2015. Integration of word and semantic features for theme identification in telephone conversations. Dans les actes de *Natural Language Dialog Systems and Intelligent Assistants*, 223–231. Springer.
- (Fan et al., 2014) Y. Fan, Y. Qian, F.-L. Xie, & F. K. Soong, 2014. Tts synthesis with bidirectional lstm based recurrent neural networks. Dans les actes de *Interspeech*, 1964–1968.
- (Fernandez et al., 2014) R. Fernandez, A. Rendel, B. Ramabhadran, & R. Hoory, 2014. Prosody contour prediction with long short-term memory, bi-directional, deep recurrent neural networks. Dans les actes de *Interspeech*, 2268–2272.
- (Fox et Roberts, 2012) C. W. Fox & S. J. Roberts, 2012. A tutorial on variational bayesian inference. *Artificial intelligence review* 38(2), 85–95.
- (Furui, 1986) S. Furui, 1986. Speaker-independent isolated word recognition based on emphasized spectral dynamics. Dans les actes de Acoustics, Speech, and Signal Processing, IEEE International Conference on ICASSP'86., Volume 11, 1991–1994. IEEE.
- (Gajecki, 2014) L. Gajecki, 2014. Architectures of neural networks applied for lvcsr language modeling. *Neurocomputing* 133, 46–53.
- (Gal et Ghahramani, 2015) Y. Gal & Z. Ghahramani, 2015. On modern deep learning and variational inference. Dans les actes de *Advances in Approximate Bayesian Inference workshop, NIPS*, Volume 2.
- (Garnier-Rizet et al., 2008) M. Garnier-Rizet, G. Adda, F. Cailliau, J. Gauvain, S. Guillemin-Lanne, L. Lamel, S. Vanni, & C. Waast-Richard, 2008. Callsurf-automatic transcription, indexing and structuration of call center conversational speech for knowledge extraction and query by content. Dans les actes de *Proceedings of LREC*.

- (Garofolo et al., 1993) J. S. Garofolo, L. F. Lamel, W. M. Fisher, J. G. Fiscus, & D. S. Pallett, 1993. Darpa timit acoustic-phonetic continous speech corpus cd-rom. nist speech disc 1-1.1. *NASA STI/Recon technical report n* 93.
- (Gers et al., 2001) F. A. Gers, D. Eck, & J. Schmidhuber, 2001. Applying lstm to time series predictable through time-window approaches. Dans les actes de *Artificial Neural Networks at ICANN 2001*, 669–676. Springer.
- (Glorot et Bengio, 2010) X. Glorot & Y. Bengio, 2010. Understanding the difficulty of training deep feedforward neural networks. Dans les actes de *International conference on artificial intelligence and statistics*, 249–256.
- (Goodfellow et al., 2016) I. Goodfellow, Y. Bengio, & A. Courville, 2016. *Deep Learning*. MIT Press. http://www.deeplearningbook.org.
- (Goodfellow et al., 2014) I. Goodfellow, J. Pouget-Abadie, M. Mirza, B. Xu, D. Warde-Farley, S. Ozair, A. Courville, & Y. Bengio, 2014. Generative adversarial nets. Dans les actes de *Advances in neural information processing systems*, 2672–2680.
- (Graves, 2012) A. Graves, 2012. Neural networks. Dans les actes de *Supervised Sequence Labelling with Recurrent Neural Networks*, 15–35. Springer.
- (Graves et al., 2006) A. Graves, S. Fernández, F. Gomez, & J. Schmidhuber, 2006. Connectionist temporal classification: labelling unsegmented sequence data with recurrent neural networks. Dans les actes de *Proceedings of the 23rd international conference on Machine learning*, 369–376. ACM.
- (Graves et al., 2007) A. Graves, S. Fernández, & J. Schmidhuber, 2007. Multidimensional recurrent neural networks. Dans les actes de Artificial Neural Networks - ICANN 2007, 17th International Conference, Porto, Portugal, September 9-13, 2007, Proceedings, Part I, 549–558.
- (Graves et al., 2013a) A. Graves, N. Jaitly, & A.-r. Mohamed, 2013a. Hybrid speech recognition with deep bidirectional lstm. Dans les actes de *Automatic Speech Recognition and Understanding (ASRU)*, 2013 IEEE Workshop on, 273–278. IEEE.
- (Graves et al., 2013b) A. Graves, A.-r. Mohamed, & G. Hinton, 2013b. Speech recognition with deep recurrent neural networks. Dans les actes de *Acoustics, speech and signal processing (icassp), 2013 ieee international conference on,* 6645–6649. IEEE.
- (Graves et Schmidhuber, 2005) A. Graves & J. Schmidhuber, 2005. Framewise phoneme classification with bidirectional lstm networks. Dans les actes de *Neural Networks*, 2005. *IJCNN'05. Proceedings*. 2005 *IEEE International Joint Conference on*, Volume 4, 2047–2052. IEEE.
- (Graves et Schmidhuber, 2009) A. Graves & J. Schmidhuber, 2009. Offline handwriting recognition with multidimensional recurrent neural networks. Dans D. Koller, D. Schuurmans, Y. Bengio, & L. Bottou (Eds.), Advances in Neural Information Processing Systems 21, 545–552. Curran Associates, Inc.

- (Greff et al., 2017) K. Greff, R. K. Srivastava, J. Koutník, B. R. Steunebrink, & J. Schmidhuber, 2017. Lstm: A search space odyssey. *IEEE transactions on neural networks and learning systems*.
- (Gregor et al., 2015) K. Gregor, I. Danihelka, A. Graves, D. J. Rezende, & D. Wierstra, 2015. Draw: A recurrent neural network for image generation. *arXiv preprint arXiv:*1502.04623.
- (Grèzl et al., 2007) F. Grèzl, M. Karafiàt, S. Kontàr, & J. Cernocky, 2007. Probabilistic and bottle-neck features for lvcsr of meetings. Dans les actes de *International Conference on Acoustics, Speech and Signal Processing (ICASSP)*, IV–757. IEEE.
- (Haykin et Network, 2004) S. Haykin & N. Network, 2004. A comprehensive foundation. *Neural Networks* 2.
- (Hazen, 2011) T. Hazen, 2011. Topic identification. *Spoken Language Understanding: Systems for Extracting Semantic Information from Speech*, 319–356.
- (He et al., 2015) K. He, X. Zhang, S. Ren, & J. Sun, 2015. Delving deep into rectifiers: Surpassing human-level performance on imagenet classification. Dans les actes de *Proceedings of the IEEE international conference on computer vision*, 1026–1034.
- (He et al., 2016) K. He, X. Zhang, S. Ren, & J. Sun, 2016. Deep residual learning for image recognition. Dans les actes de *Proceedings of the IEEE conference on computer* vision and pattern recognition, 770–778.
- (Hinton et al., 2006) G. E. Hinton, S. Osindero, & Y.-W. Teh, 2006. A fast learning algorithm for deep belief nets. *Neural computation* 18(7), 1527–1554.
- (Hinton et Salakhutdinov, 2006) G. E. Hinton & R. R. Salakhutdinov, 2006. Reducing the dimensionality of data with neural networks. *Science* 313(5786), 504–507.
- (Hirose et Yoshida, 2012) A. Hirose & S. Yoshida, 2012. Generalization characteristics of complex-valued feedforward neural networks in relation to signal coherence. *IEEE Transactions on Neural Networks and learning systems* 23(4), 541–551.
- (Hochreiter, 1998) S. Hochreiter, 1998. The vanishing gradient problem during learning recurrent neural nets and problem solutions. *International Journal of Uncertainty, Fuzziness and Knowledge-Based Systems* 6(02), 107–116.
- (Hochreiter et Schmidhuber, 1997) S. Hochreiter & J. Schmidhuber, 1997. Long short-term memory. *Neural computation* 9(8), 1735–1780.
- (Hu et Wang, 2012) J. Hu & J. Wang, 2012. Global stability of complex-valued recurrent neural networks with time-delays. *IEEE Transactions on Neural Networks and Learning Systems* 23(6), 853–865.
- (Huang et al., 2016) M. Huang, Y. Cao, & C. Dong, 2016. Modeling rich contexts for sentiment classification with lstm. *CoRR abs/1605.01478*.

- (Isokawa et al., 2003) T. Isokawa, T. Kusakabe, N. Matsui, & F. Peper, 2003. Quaternion neural network and its application. Dans les actes de *International Conference on Knowledge-Based and Intelligent Information and Engineering Systems*, 318–324. Springer.
- (Isokawa et al., 2009) T. Isokawa, N. Matsui, & H. Nishimura, 2009. Quaternionic neural networks: Fundamental properties and applications. *Complex-Valued Neural Networks: Utilizing High-Dimensional Parameters*, 411–439.
- (Janod et al., 2016) K. Janod, M. Morchid, R. Dufour, G. Linarès, & R. De Mori, 2016. Deep stacked autoencoders for spoken language understanding. Dans les actes de *Conference of the International Speech Communication Association (INTERSPEECH)*. ISCA.
- (Janod et al., 2017) K. Janod, M. Morchid, R. Dufour, G. Linares, & R. De Mori, 2017. Denoised bottleneck features from deep autoencoders for telephone conversation analysis. *IEEE/ACM Transactions on Audio, Speech, and Language Processing* 25(9), 1809–1820.
- (Janod et al., 2016) K. Janod, M. Morchid, R. Dufour, G. Linarès, & R. D. Mori, 2016. Deep stacked autoencoders for spoken language understanding. Dans les actes de *Interspeech* 2016, 720–724.
- (Kim et al., 2017) T. Kim, M. Cha, H. Kim, J. K. Lee, & J. Kim, 2017. Learning to discover cross-domain relations with generative adversarial networks. Dans les actes de *Proceedings of the 34th International Conference on Machine Learning-Volume 70*, 1857– 1865. JMLR. org.
- (Kingma et Ba, 2014) D. Kingma & J. Ba, 2014. Adam: A method for stochastic optimization. *arXiv preprint arXiv:1412.6980*.
- (Kingma et Welling, 2013) D. P. Kingma & M. Welling, 2013. Auto-encoding variational bayes. *arXiv preprint arXiv:1312.6114*.
- (Kohonen, 1982) T. Kohonen, 1982. Self-organized formation of topologically correct feature maps. *Biological cybernetics* 43(1), 59–69.
- (Kulkarni et al., 2015) T. D. Kulkarni, W. F. Whitney, P. Kohli, & J. Tenenbaum, 2015. Deep convolutional inverse graphics network. Dans les actes de *Advances in neural information processing systems*, 2539–2547.
- (Lagus et Kuusisto, 2002) K. Lagus & J. Kuusisto, 2002. Topic identification in natural language dialogues using neural networks. Dans les actes de *Proceedings of the Third SIGdial Workshop on Discourse and Dialogue*, Philadelphia, Pennsylvania, USA, 95–102. Association for Computational Linguistics.
- (Larsen et al., 2015) A. B. L. Larsen, S. K. Sønderby, H. Larochelle, & O. Winther, 2015. Autoencoding beyond pixels using a learned similarity metric. *arXiv preprint arXiv:*1512.09300.

- (LeCun et al., 2015) Y. LeCun, Y. Bengio, & G. Hinton, 2015. Deep learning. *Nature* 521(7553), 436–444.
- (LeCun et al., 1989) Y. LeCun, B. Boser, J. S. Denker, D. Henderson, R. E. Howard, W. Hubbard, & L. D. Jackel, 1989. Backpropagation applied to handwritten zip code recognition. *Neural computation* 1(4), 541–551.
- (LeCun et al., 1998) Y. LeCun, L. Bottou, Y. Bengio, & P. Haffner, 1998. Gradient-based learning applied to document recognition. *Proceedings of the IEEE* 86(11), 2278–2324.
- (Li et al., 2017) J. Li, W. Monroe, T. Shi, S. Jean, A. Ritter, & D. Jurafsky, 2017. Adversarial learning for neural dialogue generation. *arXiv preprint arXiv:1701.06547*.
- (Ma et al., 2019) C.-Y. Ma, M.-H. Chen, Z. Kira, & G. AlRegib, 2019. Ts-lstm and temporal-inception: Exploiting spatiotemporal dynamics for activity recognition. *Signal Processing: Image Communication* 71, 76–87.
- (Makhzani et al., 2015) A. Makhzani, J. Shlens, N. Jaitly, I. Goodfellow, & B. Frey, 2015. Adversarial autoencoders. *arXiv preprint arXiv:1511.05644*.
- (Matsui et al., 2004) N. Matsui, T. Isokawa, H. Kusamichi, F. Peper, & H. Nishimura, 2004. Quaternion neural network with geometrical operators. *Journal of Intelligent & Fuzzy Systems* 15(3, 4), 149–164.
- (McClelland et al., 1986) J. L. McClelland, D. E. Rumelhart, P. R. Group, et al., 1986. Parallel distributed processing. *Explorations in the Microstructure of Cognition 2*, 216–271.
- (McCulloch et Pitts, 1943) W. S. McCulloch & W. Pitts, 1943. A logical calculus of the ideas immanent in nervous activity. *The bulletin of mathematical biophysics* 5(4), 115–133.
- (Medsker et Jain, 2001) L. R. Medsker & L. J. Jain, 2001. Recurrent neural networks. *Design and Applications 5.*
- (Melamed et Gilbert, 2011) I. Melamed & M. Gilbert, 2011. Speech analytics. *Spoken Language Understanding: Systems for Extracting Semantic Information from Speech*, 397–416.
- (Mikolov, 2012) T. Mikolov, 2012. Statistical language models based on neural networks. *Presentation at Google, Mountain View, 2nd April.*
- (Minemoto et al., 2017) T. Minemoto, T. Isokawa, H. Nishimura, & N. Matsui, 2017. Feed forward neural network with random quaternionic neurons. *Signal Processing* 136, 59–68.
- (Mirza et Osindero, 2014) M. Mirza & S. Osindero, 2014. Conditional generative adversarial nets. *arXiv preprint arXiv:1411.1784*.
- (Mohri et al., 2002) M. Mohri, F. Pereira, & M. Riley, 2002. Weighted finite-state transducers in speech recognition. *Computer Speech and Language* 16(1), 69 – 88.

- (Morchid, 2018) M. Morchid, 2018. Parsimonious memory unit for recurrent neural networks with application to natural language processing. *Neurocomputing* 314, 48–64.
- (Morchid et al., 2015a) M. Morchid, M. Bouallegue, R. Dufour, G. Linarès, D. Matrouf, & R. De Mori, 2015a. Compact multiview representation of documents based on the total variability space. *IEEE/ACM Transactions on Audio, Speech, and Language Processing* 23(8), 1295–1308.
- (Morchid et al., 2014a) M. Morchid, R. Dufour, M. Bouallegue, G. Linarès, & R. De Mori, 2014a. Theme identification in human-human conversations with features from specific speaker type hidden spaces. Dans les actes de *Conference of the International Speech Communication Association (INTERSPEECH) 2014.* ISCA.
- (Morchid et al., 2014b) M. Morchid, R. Dufour, P.-M. Bousquet, M. Bouallegue, G. Linarès, & R. De Mori, 2014b. Improving dialogue classification using a topic space representation and a gaussian classifier based on the decision rule. Dans les actes de *International Conference on Acoustics, Speech and Signal Processing (ICASSP)*, 126–130. IEEE.
- (Morchid et al., 2014c) M. Morchid, R. Dufour, P.-M. Bousquet, M. Bouallegue, G. Linarès, & R. De Mori, 2014c. Improving dialogue classification using a topic space representation and a gaussian classifier based on the decision rule. Dans les actes de *International Conference on Acoustic, Speech and Signal Processing (ICASSP)* 2014. IEEE.
- (Morchid et al., 2015b) M. Morchid, R. Dufour, & G. Linarès, 2015b. Topic-space based setup of a neural network for theme identification of highly imperfect transcriptions. Dans les actes de *Automatic Speech Recognition and Understanding (ASRU)*, 2015 IEEE Workshop on, 346–352. IEEE.
- (Morchid et al., 2016) M. Morchid, R. Dufour, & G. Linarès, 2016. Impact of word error rate on theme identification task of highly imperfect human–human conversations. *Computer Speech & Language 38*, 68–85.
- (Morchid et al., 2013) M. Morchid, G. Linarès, M. El-Beze, & R. De Mori, 2013. Theme identification in telephone service conversations using quaternions of speech features. Dans les actes de *Conference of the International Speech Communication Association* (*INTERSPEECH*) 2013. ISCA.
- (Nasr et al., 2002) G. E. Nasr, E. Badr, & C. Joun, 2002. Cross entropy error function in neural networks: Forecasting gasoline demand. Dans les actes de *FLAIRS Conference*, 381–384.
- (Nasrabadi et Feng, 1988) N. M. Nasrabadi & Y. Feng, 1988. Vector quantization of images based upon the kohonen self-organizing feature maps. Dans les actes de *Proc. IEEE Int. Conf. Neural Networks*, Volume 1, 101–105.

- (Nguyen et al., 2017) A. Nguyen, J. Clune, Y. Bengio, A. Dosovitskiy, & J. Yosinski, 2017. Plug & play generative networks: Conditional iterative generation of images in latent space. Dans les actes de *Proceedings of the IEEE Conference on Computer Vision* and Pattern Recognition, 4467–4477.
- (Nitta, 1995) T. Nitta, 1995. A quaternary version of the back-propagation algorithm. Dans les actes de *Proceedings of the IEEE International Conference on Neural Networks*, Volume 5, 2753–2756.
- (Odena, 2016) A. Odena, 2016. Semi-supervised learning with generative adversarial networks. *arXiv preprint arXiv:1606.01583*.
- (Parcollet et al., 2016) T. Parcollet, M. Morchid, P.-M. Bousquet, R. Dufour, G. Linarès, & R. De Mori, 2016. Quaternion neural networks for spoken language understanding. Dans les actes de *Spoken Language Technology Workshop (SLT)*, 2016 IEEE, 362–368. IEEE.
- (Parcollet et al., 2017) T. Parcollet, M. Morchid, & G. Linarès, 2017. Deep quaternion neural networks for spoken language understanding. Dans les actes de *Automatic Speech Recognition and Understanding Workshop (ASRU)*, 2017 IEEE, 504–511. IEEE.
- (Parcollet et al., 2019) T. Parcollet, M. Morchid, & G. Linarès, 2019. Quaternion convolutional neural networks for heterogeneous image processing. Dans les actes de *Acoustics, Speech, and Signal Processing, IEEE International Conference on ICASSP.*, Volume 44, 1–6. IEEE.
- (Parcollet et al., 2019) T. Parcollet, M. Morchid, G. Linarès, & R. De Mori, 2019. Bidirectional quaternion long-short term memory recurrent neural networks for speech recognition. Dans les actes de *Acoustics, Speech, and Signal Processing, IEEE International Conference on ICASSP.*, Volume 44, 1–6. IEEE.
- (Parcollet et al., 2018a) T. Parcollet, M. Ravanelli, M. Morchid, G. Linarès, & R. De Mori, 2018a. Speech recognition with quaternion neural networks. *IRASL@NIPS*.
- (Parcollet et al., 2018b) T. Parcollet, M. Ravanelli, M. Morchid, G. Linarès, C. Trabelsi, R. De Mori, & Y. Bengio, 2018b. Quaternion recurrent neural networks. *arXiv preprint arXiv:1806.04418*.
- (Parcollet et al., 2018c) T. Parcollet, M. Ravanelli, M. Morchid, G. Linarès, C. Trabelsi, R. D. Mori, & Y. Bengio, 2018c. Quaternion recurrent neural networks.
- (Parcollet et al., 2018d) T. Parcollet, Y. Zhang, M. Morchid, C. Trabelsi, G. Linarès, R. de Mori, & Y. Bengio, 2018d. Quaternion convolutional neural networks for endto-end automatic speech recognition. Dans les actes de *Interspeech 2018, 19th Annual Conference of the International Speech Communication Association, Hyderabad, India, 2-6 September 2018., 22–26.*
- (Pearlmutter, 1989) B. A. Pearlmutter, 1989. Learning state space trajectories in recurrent neural networks. *Neural Computation* 1(2), 263–269.

- (Pedregosa et al., 2011) F. Pedregosa, G. Varoquaux, A. Gramfort, V. Michel, B. Thirion, O. Grisel, M. Blondel, P. Prettenhofer, R. Weiss, V. Dubourg, et al., 2011. Scikit-learn: Machine learning in python. *The Journal of Machine Learning Research* 12, 2825–2830.
- (Pei et Cheng, 1999) S.-C. Pei & C.-M. Cheng, 1999. Color image processing by using binary quaternion-moment-preserving thresholding technique. *Image Processing*, *IEEE Transactions on* 8(5), 614–628.
- (Povey et al., 2011) D. Povey, A. Ghoshal, G. Boulianne, L. Burget, O. Glembek, N. Goel, M. Hannemann, P. Motlicek, Y. Qian, P. Schwarz, J. Silovsky, G. Stemmer, & K. Vesely, 2011. The kaldi speech recognition toolkit. Dans les actes de *IEEE 2011 Workshop on Automatic Speech Recognition and Understanding*. IEEE Signal Processing Society. IEEE Catalog No.: CFP11SRW-USB.
- (Purver, 2011) M. Purver, 2011. Topic segmentation. Spoken Language Understanding: Systems for Extracting Semantic Information from Speech, 291–317.
- (Radford et al., 2015) A. Radford, L. Metz, & S. Chintala, 2015. Unsupervised representation learning with deep convolutional generative adversarial networks. *arXiv preprint arXiv*:1511.06434.
- (Raina et al., 2009) R. Raina, A. Madhavan, & A. Y. Ng, 2009. Large-scale deep unsupervised learning using graphics processors. Dans les actes de *Proceedings of the 26th annual international conference on machine learning*, 873–880. ACM.
- (Rajeswar et al., 2017) S. Rajeswar, S. Subramanian, F. Dutil, C. Pal, & A. Courville, 2017. Adversarial generation of natural language. arXiv preprint arXiv:1705.10929.
- (Ravanelli et al., 2018a) M. Ravanelli, P. Brakel, M. Omologo, & Y. Bengio, 2018a. Light gated recurrent units for speech recognition. *IEEE Transactions on Emerging Topics in Computational Intelligence* 2(2), 92–102.
- (Ravanelli et al., 2018b) M. Ravanelli, T. Parcollet, & Y. Bengio, 2018b. The pytorchkaldi speech recognition toolkit. *arXiv preprint arXiv:1811.07453*.
- (Reed et al., 2016) S. Reed, Z. Akata, X. Yan, L. Logeswaran, B. Schiele, & H. Lee, 2016. Generative adversarial text to image synthesis. *arXiv preprint arXiv:1605.05396*.
- (Rosenblatt, 1958) F. Rosenblatt, 1958. The perceptron: a probabilistic model for information storage and organization in the brain. *Psychological review* 65(6), 386.
- (Rousseau et al., 2014) A. Rousseau, G. Boulianne, P. Deléglise, Y. Estève, V. Gupta, & S. Meignier, 2014. Lium and crim asr system combination for the repere evaluation campaign. Dans les actes de *International Conference on Text, Speech, and Dialogue*, 441–448. Springer.
- (Rumelhart et MacClelland, 1986) D. Rumelhart & J. MacClelland, 1986. Learning internal representations by error backpropagation. chapter 8 from parallel distributed processing. vol. 1: Foundations.

- (Rumelhart et al., 1985) D. E. Rumelhart, G. E. Hinton, & R. J. Williams, 1985. Learning internal representations by error propagation. Rapport technique, California Univ San Diego La Jolla Inst for Cognitive Science.
- (Rumelhart et al., 1986) D. E. Rumelhart, J. L. McClelland, P. R. Group, et al., 1986. Parallel distributed processing. *Explorations in the Microstructure of Cognition 1*, 216–271.
- (Sabour et al., 2017) S. Sabour, N. Frosst, & G. E. Hinton, 2017. Dynamic routing between capsules. *arXiv preprint arXiv:1710.09829v2*.
- (Salimans et al., 2016) T. Salimans, I. Goodfellow, W. Zaremba, V. Cheung, A. Radford, & X. Chen, 2016. Improved techniques for training gans. Dans les actes de *Advances in neural information processing systems*, 2234–2242.
- (Sangwine, 1996) S. J. Sangwine, 1996. Fourier transforms of colour images using quaternion or hypercomplex, numbers. *Electronics letters* 32(21), 1979–1980.
- (Schuster, 1999) M. Schuster, 1999. On supervised learning from sequential data with applications for speech recognition. *Daktaro disertacija, Nara Institute of Science and Technology*.
- (Schuster et Paliwal, 1997) M. Schuster & K. K. Paliwal, 1997. Bidirectional recurrent neural networks. *Signal Processing, IEEE Transactions on* 45(11), 2673–2681.
- (Severyn et Moschitti, 2015) A. Severyn & A. Moschitti, 2015. Twitter sentiment analysis with deep convolutional neural networks. Dans les actes de *Proceedings of the 38th International ACM SIGIR Conference on Research and Development in Information Retrieval*, 959–962. ACM.
- (Singh et al., 2016) B. Singh, T. K. Marks, M. Jones, O. Tuzel, & M. Shao, 2016. A multi-stream bi-directional recurrent neural network for fine-grained action detection. Dans les actes de *Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition*, 1961–1970.
- (Siniscalchi et al., 2013) S. M. Siniscalchi, D. Yu, L. Deng, & C.-H. Lee, 2013. Exploiting deep neural networks for detection-based speech recognition. *Neurocomputing* 106, 148–157.
- (Song et Yam, 1998) J. Song & Y. Yam, 1998. Complex recurrent neural network for computing the inverse and pseudo-inverse of the complex matrix. *Applied mathematics and computation* 93(2-3), 195–205.
- (Srivastava et al., 2014) N. Srivastava, G. Hinton, A. Krizhevsky, I. Sutskever, & R. Salakhutdinov, 2014. Dropout: A simple way to prevent neural networks from overfitting. *The Journal of Machine Learning Research* 15(1), 1929–1958.
- (Srivastava et al., 2013) N. Srivastava, R. Salakhutdinov, & G. Hinton, 2013. Modeling documents with a deep boltzmann machine. Dans les actes de *Proceedings of the Twenty-Ninth Conference on Uncertainty in Artificial Intelligence*, 616–624. AUAI Press.

- (Stolcke et al., 2002) A. Stolcke et al., 2002. Srilm-an extensible language modeling toolkit. Dans les actes de *INTERSPEECH*, Volume 2002, 2002.
- (Stollenga et al., 2015) M. F. Stollenga, W. Byeon, M. Liwicki, & J. Schmidhuber, 2015. Parallel multi-dimensional lstm, with application to fast biomedical volumetric image segmentation. Dans les actes de *Advances in neural information processing systems*, 2998–3006.
- (Sundermeyer et al., 2012) M. Sundermeyer, R. Schlüter, & H. Ney, 2012. Lstm neural networks for language modeling. Dans les actes de *INTERSPEECH*, 194–197.
- (Sutskever et al., 2014) I. Sutskever, O. Vinyals, & Q. V. Le, 2014. Sequence to sequence learning with neural networks. Dans les actes de *Advances in neural information processing systems*, 3104–3112.
- (Titouan et al., 2019a) P. Titouan, M. Morchid, X. Bost, & G. Linares, 2019a. M2h-gan: A gan-based mapping from machine to human transcripts for speech understanding. *Proc. Interspeech* 2019, 3325–3328.
- (Titouan et al., 2017) P. Titouan, M. Morchid, & G. Linarès, 2017. Quaternion denoising encoder-decoder for theme identification of telephone conversations. *Proc. Interspeech* 2017, 3325–3328.
- (Titouan et al., 2019b) P. Titouan, M. Morchid, G. Linares, & R. De Mori, 2019b. Real to h-space encoder for speech recognition. *Proc. Interspeech* 2019, 3325–3328.
- (Titouan et al., 2018) P. Titouan, Z. Ying, M. Mohamed, T. Chiheb, L. Georges, D. M. Renato, & B. Yoshua, 2018. Quaternion convolutional neural networks for end-to-end automatic speech recognition. arXiv preprint arXiv:1806.07789.
- (Tokuda et al., 2003) K. Tokuda, H. Zen, & T. Kitamura, 2003. Trajectory modeling based on hmms with the explicit relationship between static and dynamic features. Dans les actes de *Eighth European Conference on Speech Communication and Technology*.
- (Trabelsi et al., 2017) C. Trabelsi, O. Bilaniuk, D. Serdyuk, S. Subramanian, J. F. Santos, S. Mehri, N. Rostamzadeh, Y. Bengio, & C. J. Pal, 2017. Deep complex networks. *arXiv preprint arXiv:*1705.09792.
- (Tur et De Mori, 2011) G. Tur & R. De Mori, 2011. Spoken language understanding: Systems for extracting semantic information from speech. John Wiley & Sons.
- (Turian et al., 2010) J. Turian, L. Ratinov, & Y. Bengio, 2010. Word representations: a simple and general method for semi-supervised learning. Dans les actes de *Proceedings of the 48th annual meeting of the association for computational linguistics*, 384–394. Association for Computational Linguistics.
- (Tygert et al., 2016) M. Tygert, J. Bruna, S. Chintala, Y. LeCun, S. Piantino, & A. Szlam, 2016. A mathematical motivation for complex-valued convolutional networks. *Neural computation* 28(5), 815–825.

- (Ulyanov et al., 2018) D. Ulyanov, A. Vedaldi, & V. Lempitsky, 2018. It takes (only) two: Adversarial generator-encoder networks. Dans les actes de *Thirty-Second AAAI Conference on Artificial Intelligence*.
- (Van Asch, 2013) V. Van Asch, 2013. Macro-and micro-averaged evaluation measures [[basic draft]].
- (Vincent et al., 2010) P. Vincent, H. Larochelle, I. Lajoie, Y. Bengio, & P.-A. Manzagol, 2010. Stacked denoising autoencoders: Learning useful representations in a deep network with a local denoising criterion. *The Journal of Machine Learning Research* 11, 3371–3408.
- (Vinyals et al., 2015) O. Vinyals, A. Toshev, S. Bengio, & D. Erhan, 2015. Show and tell: A neural image caption generator. Dans les actes de *Proceedings of the IEEE Conference* on Computer Vision and Pattern Recognition, 3156–3164.
- (Warde-Farley et Bengio, 2016) D. Warde-Farley & Y. Bengio, 2016. Improving generative adversarial networks with denoising feature matching.
- (Werbos, 1974) P. Werbos, 1974. Beyond regression:" new tools for prediction and analysis in the behavioral sciences. *Ph. D. dissertation, Harvard University*.
- (Wisdom et al., 2016) S. Wisdom, T. Powers, J. Hershey, J. Le Roux, & L. Atlas, 2016. Full-capacity unitary recurrent neural networks. Dans les actes de *Advances in Neural Information Processing Systems*, 4880–4888.
- (Wu et al., 2017a) L. Wu, Y. Xia, L. Zhao, F. Tian, T. Qin, J. Lai, & T.-Y. Liu, 2017a. Adversarial neural machine translation. *arXiv preprint arXiv:1704.06933*.
- (Wu et al., 2017b) Y. Wu, M. Yuan, S. Dong, L. Lin, & Y. Liu, 2017b. Remaining useful life estimation of engineered systems using vanilla lstm neural networks. *Neurocomputing*.
- (Wu et al., 2016) Z. Wu, Y.-G. Jiang, X. Wang, H. Ye, & X. Xue, 2016. Multi-stream multiclass fusion of deep networks for video classification. Dans les actes de *Proceedings of the 24th ACM international conference on Multimedia*, 791–800. ACM.
- (Xu et al., 2017) D. Xu, L. Zhang, & H. Zhang, 2017. Learning alogrithms in quaternion neural networks using ghr calculus. *Neural Network World* 27(3), 271.
- (Xu et al., 2013) X. Xu, L. Lu, P. He, Z. Pan, & L. Chen, 2013. Improving constrained clustering via swarm intelligence. *Neurocomputing* 116, 317–325.
- (Yang et al., 2017) Z. Yang, W. Chen, F. Wang, & B. Xu, 2017. Improving neural machine translation with conditional sequence generative adversarial nets. *arXiv preprint arXiv:*1703.04887.
- (Yao et al., 2015) K. Yao, T. Cohn, K. Vylomova, K. Duh, & C. Dyer, 2015. Depth-gated recurrent neural networks. *arXiv preprint*.

- (Zaremba, 2015) W. Zaremba, 2015. An empirical exploration of recurrent network architectures. *Journal of Machine Learning Research*.
- (Zhang et al., 2015) X. Zhang, J. Zhao, & Y. LeCun, 2015. Character-level convolutional networks for text classification. Dans les actes de *Advances in neural information processing systems*, 649–657.
- (Zhou et al., 2016) G.-B. Zhou, J. Wu, C.-L. Zhang, & Z.-H. Zhou, 2016. Minimal gated unit for recurrent neural networks. *International Journal of Automation and Computing* 13(3), 226–234.
- (Zhu et al., 2016) J.-Y. Zhu, P. Krähenbühl, E. Shechtman, & A. A. Efros, 2016. Generative visual manipulation on the natural image manifold. Dans les actes de *European Conference on Computer Vision*, 597–613. Springer.
- (Zhu et al., 2017) J.-Y. Zhu, T. Park, P. Isola, & A. A. Efros, 2017. Unpaired imageto-image translation using cycle-consistent adversarial networks. Dans les actes de *Proceedings of the IEEE International Conference on Computer Vision*, 2223–2232.

Bibliography

## CURRICULUM VITÆ Mohamed Morchid

# État civil

- Mohamed Morchid, né le 27 décembre 1977, célibataire, sans enfant
- Maître de conferences CL Normal à l'Avignon Université (LIA), EA 4128
- Adresse postale professionnelle : CERI/LIA, Mohamed Morchid 339, chemin Meinajaries Agroparc BP 1228 84911 AVIGNON Cedex 9, France
- Tél: 0490843577 E-mail: mohamed.morchid@univ-avignon.fr Site: mohamedmorchid.free.fr

## Titres universitaires

- Thèse de doctorat en Traitement Automatique du Langage à l'Avignon Université, Avignon, France. Intitulé "Représentations robustes de documents bruités dans des espaces homogènes " soutenu le 25 Novembre 2014, devant le Jury composé de :
  - F. BéCHET, Professeur, LIF, Marseille, France (Président/Examinateur)
  - J. BELLEGARDA, Directeur, Apple Computer Inc., états-Unis (Rapporteur)
  - L. BESACIER, Professeur, LIG, Grenoble, France (Rapporteur)
  - F. YVON, Professeur, Université Paris-Sud, France (Rapporteur)
  - Y. HAMADI, Sr. Researcher, Microsoft Research, Royaume-Uni (Examinateur)
  - B. PIWOWARSKI, Chargé de Recherche CNRS, LIP6, Paris (Examinateur)
  - G. LINARèS, France Professeur, LIA, Avignon, France (Directeur)
  - R. DUFOUR, Maître de Conférences, LIA, Avignon, France (Co-Encadrant)
- Diplôme de Master en Informatique à l'Université d'Avignon ; Titre : "Prédiction du Buzz"
- Diplôme de Licence en Informatique à l'Université d'Avignon

## **Expérience professionnelle**

- Maître de conferences à l'Avignon Université (09/2015 Présent)
- Attaché Temporaire d'Enseignement et de Recherche (ATER) à l'Avignon Université (09/2014 08/2015) Thèmes de recherche : " Compréhension de la parole " dans le cadre du projet ANR contNomina (ANR-12-BS02-0009)
- Chercheur invité à *Microsoft Research Cambridge* (Royaume-Unis) (11/2013 02/2014) Thèmes de recherche : " Réalisation de l'Assistant PowerPoint pour l'aide à la réalisation de présentation oar des espaces thématiques latents. " dans le cadre du projet international *PowerPoint Assistant*
- Stage de Master recherche au Laboratoire Informatique d'Avignon (LIA) de l'Avignon Université (02/2011 - 08/2011) - Thème de recherche : " Prédiction du Buzz ", sous la direction de Professeur Georges Linarès

### Thèmes de recherche

Dans cette partie, je présente mes activités de recherche des dernières années (Post-Doctorat) en collaboration avec mes étudiants (Master, Doctorat) et autres co-encadrants et directeurs de thèse. Les numéros de citation ont été extraits de la liste des publications jointe à ce document.

Représentations robustes du contenu parlé dans des espaces cachés issus de réseaux de neurones (Travail réalisé en collaboration avec Killian Janod, Doctorant au LIA). Les méthodes de compréhension de la parole visent à extraire des éléments représentatifs du contenu informatif depuis le signal de parole. On distingue principalement deux catégories dans la compréhension du signal de parole : la compréhension de dialogues homme/machine et la compréhension de dialogues homme/homme. En fonction du type de conversation, la structure des dialogues et les objectifs de compréhension varient. Cependant, dans les deux cas, les systèmes automatiques reposent le plus souvent sur une étape de reconnaissance automatique de la parole pour réaliser une transcription textuelle du signal parlé. Les systèmes de reconnaissance automatique de la parole, même les plus avancés, produisent dans des contextes acoustiques complexes, des transcriptions erronées ou partiellement erronées. Ces erreurs s'expliquent par la présence d'informations de natures et de fonction variées, telles que celles liées aux spécificités du locuteur ou encore liées à l'environnement sonore. Celles-ci peuvent avoir un impact négatif important pour la compréhension. Dans un premier temps, je montre que l'utilisation d'autoencodeur profond permet de produire une représentation latente des transcriptions d'un plus haut niveau d'abstraction  $[C_{14}]$ . Cette représentation permet au système de compréhension de la parole d'être plus robuste aux erreurs de transcriptions automatiques. Dans un second temps, je propose deux approches pour générer des représentations robustes en combinant plusieurs vues d'un même dialogue dans le but d'améliorer les performances du système la compréhension  $[C_{17}]$ . La première approche montre que plusieurs espaces thématiques différents peuvent être combinés simplement à l'aide d'autoencodeur ou dans un espace thématique latent pour produire une représentation qui augmente l'efficacité et la robustesse du système de compréhension de la parole. La seconde approche propose d'introduire une forme d'information de supervision dans les processus de débruitages par autoencodeur  $[J_5]$ . Ces travaux montrent que l'introduction de supervision de transcription dans un autoencodeur débruitant dégrade les représentations latentes, alors que les architectures proposées permettent de rendre comparables les performances d'un système de compréhension reposant sur une transcription automatique et un système de compréhension reposant sur des transcriptions manuelles. J'ai conduit en parallèle de ces travaux sur les autoencoders, un travail en collaboration avec l'entreprise EDD spécialisée en traitement de la parole située à Paris, autour de la thèse de Mohamed Bouaziz, d'autres travaux connexes car traitant des réseaux de neurones dans le contexte ici du traitement de flux parallèles.

Classification de séquences dans des flux parallèles à l'aide de réseaux de neurones récurrents (Travail réalisé en collaboration avec Mohamed Bouaziz, Doctorant au LIA). Les flux de contenu audiovisuel peuvent être représentés sous forme de séquences d'événements (par exemple, des suites d'émissions, de scènes, etc.). Ces données séquentielles se caractérisent par des relations chronologiques pouvant exister entre des événements successifs. Dans le contexte d'une chaîne TV, la programmation des émissions suit une cohérence définie par cette même chaîne, mais peut également être influencée par les programmations des chaînes concurrentes. Dans de telles conditions, les séquences d'événement des flux parallèles pourraient ainsi fournir des connaissances supplémentaires sur les événements d'un flux considéré. La modélisation de séquences est un sujet classique qui a été largement étudié, notamment dans le domaine de l'apprentissage automatique. Les réseaux de neurones récurrents de type Long Short-Term Memory (LSTM) ont notamment fait leur preuve dans de nombreuses applications incluant le traitement de ce type de données. Néanmoins, ces approches sont conçues pour traiter uniquement une seule séquence d'entrée à la fois. J'ai élaboré des approches capables d'intégrer conjointement des données séquentielles provenant de plusieurs flux parallèles. Le contexte applicatif de ce travail réalisé en collaboration avec l'entreprise EDD, consiste en une tâche de prédiction du genre d'une émission télévisée. Cette prédiction peut s'appuyer sur les historiques de genres des émissions précédentes de la même chaîne mais également sur les historiques appartenant à des chaînes parallèles. Nous proposons une taxonomie de genres adaptée à de tels traitements automatiques ainsi qu'un corpus de données contenant les historiques parallèles pour 4 chaînes françaises. J'ai proposé deux méthodes originales permettant d'intégrer les séquences des flux parallèles. La première, à savoir, l'architecture des LSTM parallèles (PLSTM)  $[C_{13}]$  consiste en une extension du modèle LSTM. Les PLSTM traitent simultanément chaque séquence dans une couche récurrente indépendante et combinent les sorties de chacune de ces couches pour produire la sortie finale. Pour ce qui est de la seconde proposition, dénommée MSE-SVM  $[C_{15}]$ , elle permet de tirer profit des avantages des méthodes LSTM et SVM. D'abord, des vecteurs de caractéristiques latentes sont générés indépendamment, pour chaque flux en entrée, en prenant en sortie l'événement à prédire dans le flux principal. Ces nouvelles représentations sont ensuite fusionnées et données en entrée à un algorithme SVM. Les approches PLSTM et MSE-SVM ont prouvé leur efficacité dans l'intégration des séquences parallèles en surpassant respectivement les modèles LSTM et SVM prenant uniquement en compte les séquences du flux principal. Les deux approches proposées parviennent bien à tirer profit des informations contenues dans les longues séquences. En revanche, elles ont des difficultés à traiter des séquences courtes. Ces travaux entreprit sur les réseaux de neurones récurrent additionnés à mes précédent travaux sur les nombres hyper-complexes pour la compréhension de documents parlés  $[C_{37}]$ , m'a naturellement conduit à me questionner sur l'apport de telles structures (Quaternions) pour la représentation de séquences multi-flux ou multidimensionnelles.

Réseaux de neurones fondés sur les Quaternions pour le traitement automatique du langage et pour la reconnaissance de forme (Travail réalisé en collaboration avec Titouan Parcollet). Les représentations neuronales présentées précédemment (autoencoders, RNN, etc.) permettent de traiter des données provenant de plusieurs flux d'information souvent portées sur des mediums hétérogènes. Le travail accompli sur les réseaux de neurones de nombres hyper-complexes permet de faire l'interface entre les travaux entrepris pour la représentation robuste dans des encoder-decoders, le traitement de séquence multi-flux (RNN) dans des contextes applicatifs réels comme la compréhension de documents parlés, la reconnaissance automatique de la parole, ou le traitement de l'image.

Ainsi, les réseaux de neurones de nombre hyper-complexes permettent de véhiculer l'information générée sur différents flux ou temporalités inter-dépendantes mais disposant d'une information instantanée propre. Parmi les nombres hypercomplexes, les quaternions contiennent une partie réel ainsi qu'une partie imaginaire composée de trois éléments contrairement au nombre complexe "traditionnels" qui en contiennent que un seul élément imaginaire. Ces quaternions ont la propriété de pouvoir donc stocker quatre informations représentant une entité à des instants différents ou différentes vues d'un seul et même élément dans des espaces de représentation différents. Une particularité propre aux quaternions est également liée à l'aspect structurel de son algèbre lui conférant certaines propriétés permettant de visionner les operations les plus basiques comme la multiplication, à travers le spectre de déplacement ou mouvement géodésique autour d'un vecteur porteur de cette transformation. Cette transformation est matérialisée dans l'espace euclidien par une multiplication propre à cet algèbre appelée "multiplication d'Hamilton", du nom du chercheur Anglais à l'origine de cet algèbre.

Ces dernières années, je me suis donc proposé d'étendre les travaux entamés durant ma thèse  $[C_{37}]$  sur les quaternions, au réseaux de neurones  $[J_2]$  en collaboration avec Titouan Parcollet (Doctorant en CIFRE avec l'entreprise Orkis) et Georges Linarès (Co-directeur de thèse). Ces travaux ont dans un premier temps eu pour objectif d'évaluer le modèle existant de réseaux de neurones de quaternions par Arena en 1997 dans un contexte applicatif réel de compréhension de dialogues issus du corpus DECODA. Les résultats ont montré que des représentations thématiques des mots à des instants conversationnels différents dans le dialogue, doivent être considérées de manière jointe lors de la phase d'apprentissage, pour permettre une bonne classification des dialogues en fonction des thèmes abordés par l'utilisateur. Ces résultats ont été validés lors des différentes expérimentations  $[C_{12}]$  sur des topologies neuronales différentes.

Nous avons donc par la suite étendu ce travail à la compression de données bruités issues de ce même corpus de dialogues DECODA à laide d'encoder-decoder (réseaux de neurones ayant comme entrée et sortie le même vecteur de représentation). C'est encoder-decoder permettent de compresser l'information donnée en entrée dans un espace de plus faible dimension (encoder), puis de reconstituer le vecteur à partir de cette représentation réduite (decoder). J'ai proposé donc des autoencoder de quaternion permettant de représenter dans un espace de faible dimension, des données multidimensionnelles [ $C_{11}$ ]. Néanmoins, les encoder-decoder ou autoencoder, ne sont pas ou très peu robuste aux variations de représentation. En effet, si un vecteur n'appartenant pas au corpus d'apprentissage est présenté en entrée, le système aura plus ou moins de mal à le reconstitué depuis sa représentation abstraite réduite. Ce phénomène de sur-apprentissage ou " over-fitting" est souvent rencontré, et les dispositifs permettant de pallier cette difficulté existent. Parmi celles-ci, l'introduction d'un bruit lors de la phase d'apprentissage sur les données d'entrée, contraint l'autoencoder à reconstituer le bon vecteur sachant qu'en entrée celui-ci fut altéré par une

fonction de bruit. En effet, lors de l'apprentissage, l'erreur constatée est évaluée entre le vecteur en sortie bruité et le vecteur non bruité ou "propre". J'ai donc proposé dans un premier temps de transposer les bruits connus depuis l'espace des réels, vers celui des quaternions, puis d'adapter le bruit le plus couramment utilisé, le bruit Gaussien. Ce bruit est additif, dans le sens où l'on ajoute un élément aléatoire à la distribution observée pour altérer celle-ci. Pour ce faire, j'ai donc transformé ce bruit additif en un bruit adapté à l'algèbre fondé sur les rotations des quaternions, par une rotation du vecteur portant la transformation liée à un quaternion, selon un angle défini de manière Gaussienne [ $C_{11}$ ]. Les résultats ont pu montré que ce bruit qui fait sens dans cette algèbre des quaternions permet d'obtenir de meilleurs résultats de reconstruction du vecteur d'entrée que le bruit additif.

L'avènement des réseaux de neurones dans diverses applications de traitement automatique du langage, entre autres, est essentiellement dû à leur fort potentiel d'abstraction issu de la "profondeur" du réseau. Cette profondeur permet d'empiler les couches cachées représentant chacune un niveau d'abstraction dont la granularité est plus ou moins fine selon le degré de profondeur. Ces réseaux profonds ou *deep neural networks* (DNN) convergent difficilement vers une solution optimale s'ils ne sont pas correctement initialisés. Nous avons donc par la suite proposé des réseaux de neurones profonds de quaternions initialisés à l'aide des autoencoders de quaternions [ $C_9$ ]. Les résultats ont confirmé l'apport des couches successives du réseaux ainsi que de l'initialisation à l'aide des autoencoders.

La majeur difficulté avec les données actuelles, est leur nature séquentielle et multidimensionnelle. Les réseaux convolutionnels permettent de traiter des données multidimentionnels comme les images ou toutes informations pouvant être portée sur des espaces à plusieurs dimensions. Nous avons alors ensuite proposé des réseaux de neurones de quaternions convolutionnels pour la compréhension de la parole  $[C_7]$ , pour la reconnaissance de la parole  $[C_8][C_6]$  et pour le traitement de l'image  $[C_4]$ . Les expérimentations sur le corpus TIMIT ont montrées que les quaternions permettent une meilleure représentation des dépendances entre les différents instants du signal de parole lors de la production du phonème, qu'un vecteur de réels dans lequel toutes ces informations sont concaténées  $[C_8]$ . Ce travail a été réalisé en collaboration avec Yoshua Bengio et l'Institue MILA du Canada. Ce travail est la première pierre de l'édifice d'une collaboration pérenne. En effet, je poursuis ces travaux en étroite collaboration avec les chercheurs du MILA pour étendre ces travaux aux réseaux de neurones récurrents (RNN) de quaternion  $[C_5]$   $[C_3]$  dans divers domaines applicatifs. Le problème majeur dans l'utilisabilité des réseaux de quaternions, est de trouver des caractéristiques différentes d'un même descripteurs comme un mot ou une composante de signal de parole. Jusqu'ici, nous avons employé différentes vues issues de différents espaces thématiques LDA pour obtenir différentes représentation du document ou les dérivées successives d'un même signal de parole. Malgré les bons résultats observés, il n'est toutefois pas toujours possible de trouver de tels sous-descripteurs comme pour un document texte. Nous avons alors proposé un modèle permettant la transition entre l'espace de caractéristiques réels et l'espace de quaternions appelé R2H  $[C_2]$ . Les expérimentations menées sur le corpus DECODA ont montré le potentiel d'une telle approche et ouvre les perspectives sur les réseaux de quaternions à des domaines beaucoup plus variés augmentant ainsi la visibilité des méthodes issues des quaternions développées au LIA par notre équipe.

Réseaux de neurones récurrents parcimonieux pour le traitement du langage. Le problème majeur des réseaux de neurones récurrents (RNN) est la compréhension du mécanisme de mémoire et du comportement des cellules en charge de "se rappeler" ou d'"oublier" l'information. De plus, les RNNs ne révèlent pas les dépendances à long-terme et de nouveaux RNN contenant des portes ont été proposés pour remédier à cet inconvénient, tel que les Long Short-Term Memory (LSTM). Ce modèle nécessite 4 portes pour apprendre les dépendances à court et à long-terme pour une séquence donnée d'éléments de base, et requiert donc de plus, un plus long temps lié à la phase d'apprentissage. Récemment, une nouvelle famille de RNN appelée Gated Recurrent Unit (GRU a été introduite. Le GRU contient peu de portes (portes de réinitialisation et de mise à jour) mais est basée sur le regroupement de portes sans prendre en compte les relations latentes entre les dépendances à court et à long terme. La gestion des dépendances dans le GRU à travers des unités cachées est donc similaire pour tous les neurones issus de la couche cachée du RNN. De plus, l'apprentissage des RNN nécessite une grande quantité de données et, malgré l'avènement des cartes GPU qui permettent d'apprendre des modèles plus rapidement que avec les cartes CPU, le temps de traitement reste assez important. Mes travaux les plus actuels traitent de ces problématiques. Pour surmonter ces problèmes, j'ai d'abord proposé un nouveau RNN qui tire parti des technologies LSTM, GRU (dépendances à court et à long terme) et de la *Leaky Unit* (LU) (apprentissage rapide) appelée Internal Memory Gate (IMG). L'efficacité et la robustesse de l'IMG-RNN proposé ont été illustrées lors de la tâche de classification du corpus de dialogues du projet DECODA, qui nous permet d'évaluer la capacité de chaque RNN à coder des dépendances à court terme  $[C_{10}]$ . CEs expérimentations ont montré que le IMG obtient des résultats équivalent, voire meilleurs que les

(sauf pour le LU évidement). Ces expérimentations ont été conduits sur un corpus de petite taille (750 document pour l'apprentissage) et traite donc essentiellement les dépendances à court-termes. J'ai donc, ensuite, proposé de mieux comprendre comment les états cachés gèrent les dépendances à court ou long terme dans différents RNN. Durant cette étude, j'ai montré que les RNN classique ne traite pas convenablement les dépendances entre les éléments fortement éloignés dans un document et donc introduit un nouveau RNN appelé *Parsimonious Memory Unit* (PMU). Le PMU est basé sur l'hypothèse forte que les dépendances à court et à long terme sont liées et que le rôle de chacun des neurones cachés doit être différent pour mieux gérer ces dépendances. Les expériences menées à la fois sur un petit ensemble de données de dialogues parlés (à court terme) du projet DECODA, sur un corpus de documents textuels volumineux (à long terme) des 20-Newsgroups et sur une tâche de modélisation du langage, montrent que le PMU-RNN proposé obtient de meilleures performances (efficacité) avec un temps de traitement réduit (amélioration de la portabilité) avec un gain de 50%. De plus, les expériences sur l'activité des portes montrent que le PMU proposé gère mieux les dépendances que le modèle GRU-RNN [ $J_4$ ] avec un rôle différents pour chacun des neurones cachés.

réseaux RNN, LSTM, GRU et LU tout en nécessitant moins de temps lors de la phase d'apprentissage

En résumé, mes activités scientifiques comportent plusieurs axes de recherche principaux :

- Traitement Automatique du Langage (TAL)
- Intelligence Artificielle (IA)
- Réseaux de neurones récurrents (RNN)
- Représentations complexes dans les réseaux de neurones

### Projets de recherche

#### • Projet ANR - PRCE - AISSPER

- Objectif : Intelligence artificielle pour la compréhension du langage parlé contrôlée sémantiquement (LIUM, Orkis et LIA)
- ⊃ Rôle : Coordinateur du projet
- **D** Période : 2020 2023

#### • Projet ANRT - CIFRE

- Objectif : Réseaux de neurones profonds fondés sur l'algèbre des complexes et hyper-complexes pour le traitement du langage. Co-direction de la thèse de Titouan Parcollet avec Georges Linarès (Xavier Bost pour l'entreprise Orkis)
- **⊃ Rôle** : <u>Porteur</u>. 12k $\in$ /an pendant 3 ans.
- **D** Période : 2017 2020

#### • Projet LIA Inter-disciplinaire

- Objectif : Extraction de motifs récurrents dans des documents écrits et parlés. Collaboration entre les disciplines de recherche opérationnelle (Z. Ales) et traitement du langage (M.Morchid).
- ⊃ Rôle : Co-porteur avec Zacharie Ales. 11k€.
- **D** Période : 2016 2017

#### • Projet ANR - PRC - GaFes

- Objectif : Etude des usages via des données collectées sur Internet et ré-éditorialisation des contenus captés ou produits par les internautes (SYLLABS, Cabinet d'études et de sondages, Centre Norbert Elias, EURECOM et le LIA).
- **C** Rôle : Participant durant mon année d'ATER
- **D Période :** 2014 2018

#### • Projet ANR - PRC - ContNomina

- Objectif : Reconnaissance des noms propres et contextualisation des systèmes de reconnaissance de la parole (Laboratoire Lorrain de Recherche en Informatique et ses Applications (LORIA), LIA).
- **C** Rôle : Participant durant mon année d'ATER
- **D** Période : 2013 2015

#### • Projet ANR - PRCE - DECODA

- Dispectif: DEpouillement automatique de COnversations provenant de centres D'Appels (LIF, RATP, SONEAR et LIA).
- ⊃ Rôle : Participant plus de 40 publications majeurs sur cet objet de recherche
- **> Période :** 2012 2015

- Projet ANR PRCE SuMACC
  - Objectif : Apprentissage coopératif semi-Supervisé de concepts Multimédias pour l'Aide à la Catégorisation et la détection de Concepts (EURECOM, Wikio, Syllabs et LIA).
  - **C** Rôle : Participant financement de ma thèse de doctorat
  - **D** Période : 2011 2014

### Prix et distinctions scientifiques

• Bourse de l'ISCA dans le cadre de la conférence d'audience internationale INTERSPEECH 2014 à la suite de 4 papiers acceptés [C<sub>28</sub>][C<sub>29</sub>][C<sub>30</sub>] et [C<sub>31</sub>].

### Organisation de manifestations scientifiques

- Comité d'organisation de la conférence d'audience internationale pour le traitement du langage *European Chapter of the Association for Computational Linguistics (ACL)* EACL en 2012.
- Journée d'étude corpus et textes de représentation, FR Agorantic, 15 juin 2018

#### Participations à des comités scientifiques

- Examinateur de projets ANR et de projets pour la FR Agorantic
- Examinateur pour plusieurs conférences et journaux internationaux :
  - Journaux (Reviewer) : International Journal of Robust and Nonlinear Control, Mathematics (MDPI), IEEE Transactions on Cognitive and Developmental Systems, IEEE Transactions on Cybernetics, ACM Transactions on Knowledge Discovery from Data, Applied Computing and Informatics, Big Data Research, Expert Systems with Applications, IEEE Access, Knowledge-Based Systems, International Journal of Information Management, Artificial Intelligence Review, The Institution of Engineering and Technology (IET) Journal of Engineering, The Institution of Engineering and Technology (IET) Image Processing, Computer Speech and Language, JKSU Computer and Information Sciences, Pattern Recognition Letters, Applied Sciences, Signal Processing Systems.
  - Conférences (PC/TPC) : ACL, ISCA INTERSPEECH, NAACL-HLT, EMNLP, ICPMS, IEEE ICASSP, IEEE SCC, CORIA, AAAI, ACM ICMI.

### Participations à des jurys de thèse

- Examinateur & co-encadrant de la thèse de Mohamed Bouaziz (LIA) intitulée : "*Réseaux de neurones récurrents pour la classification de séquences dans des flux audiovisuels parallèles*" soutenu le 6 décembre 2017.
- Examinateur & co-encadrant de la thèse de Killian Janod (LIA) intitulée : "La représentation des documents par réseaux de neurones pour la compréhension de documents parlés: Application au cas de la classification thématique" soutenu le 27 novembre 2017.

### Thèses encadrées

\* **Titouan Parcollet** - **En cours, (Université d'Avignon)** : Réseaux de neurones de quaternions pour le traitement du langage et la reconnaissance de forme, co-dirigée avec Lianrès G. dans le cadre d'une thèse CIFRE avec Orkis; débutée en mars 2017 - ...

\* Killian Janod - Soutenue le 27/11/2017, (Université d'Avignon) : La représentation des documents par réseaux de neurones pour la compréhension de documents parlés: Application au cas de la classification thématique, co-supervisée with Dufour R. et Lianrès G. dans le cadre d'une thèse CIFRE avec Orkis, 2013 - 2017

Résumé : Les méthodes de compréhension de la parole visent à extraire des éléments de sens pertinents du signal parlé. On distingue principalement deux catégories dans la compréhension du signal parlé : la compréhension de dialogues homme/machine et la compréhension de dialogues homme/homme. En fonction du type de conversation, la structure des dialogues et les objectifs de compréhension varient. Cependant, dans les deux cas, les systèmes automatiques reposent le plus souvent sur une étape de reconnaissance automatique de la parole pour réaliser une transcription textuelle du signal parlé. Les systèmes de reconnaissance automatique de la parole, même les plus avancés, produisent dans des contextes acoustiques complexes des transcriptions erronées ou partiellement erronées. Ces erreurs s'expliquent par la présence d'informations de natures et de fonction variées, telles que celles liées aux spécificités du locuteur ou encore l'environnement sonore. Celles-ci peuvent avoir un impact négatif important pour la compréhension. Dans un premier temps, les travaux de cette thèse montrent que l'utilisation d'autoencodeur profond permet de produire une représentation latente des transcriptions d'un plus haut niveau d'abstraction. Cette représentation permet au système de compréhension de la parole d'être plus robuste aux erreurs de transcriptions automatiques. Dans un second temps, nous proposons deux approches pour générer des représentations robustes en combinant plusieurs vues d'un même dialogue dans le but d'améliorer les performances du système la compréhension. La première approche montre que plusieurs espaces thématiques déférents peuvent être combinés simplement à l'aide d'autoencodeur ou dans un espace thématique latent pour produire une représentation qui augmente l'eficacité et la robustesse du système de compréhension de la parole. La seconde approche propose d'introduire une forme d'information de supervision dans les processus de débruitages par autoencodeur. Ces travaux montrent que l'introduction de supervision de transcription dans un autoencodeur débruitant dégrade les représentations latentes, alors que les architectures proposées permettent de rendre comparables les performances d'un système de compréhension reposant sur une transcription automatique et un système de compréhension reposant sur des transcriptions manuelles.

\* Mohamed Bouaziz - Soutenue le 6/12/2017, (Université d'Avignon) : Réseaux de neurones récurrents pour la classification de séquences dans des flux audiovisuels parallèles, co-supervisée avec Dufour R. et Lianrès G. dans le cadre d'une thèse CIFRE avec EDD, 2013 - 2017

Résumé : Les flux de contenus audiovisuels peuvent être représentés sous forme de séquences d'événements

(par exemple, des suites d'émissions, de scènes, etc.). Ces données séquentielles se caractérisent par des relations chronologiques pouvant exister entre les événements successifs. Dans le contexte d'une chaîne TV, la programmation des émissions suit une cohérence définie par cette même chaîne, mais peut également être influencée par les programmations des chaînes concurrentes. Dans de telles conditions, les séquences d'événements des flux parallèles pourraient ainsi fournir des connaissances supplémentaires sur les événements d'un flux considéré. La modélisation de séguences est un sujet classique qui a été largement étudié, notamment dans le domaine de l'apprentissage automatique. Les réseaux de neurones récurrents de type Long Short-Term Memory (LSTM) ont notamment fait leur preuve dans de nombreuses applications incluant le traitement de ce type de données. Néanmoins, ces approches sont conçues pour traiter uniquement une seule séquence d'entrée à la fois. Notre contribution dans le cadre de cette thèse consiste à élaborer des approches capables d'intégrer conjointement des données séquentielles provenant de plusieurs flux parallèles.Le contexte applicatif de ce travail de thèse, réalisé en collaboration avec le Laboratoire Informatique d'Avignon et l'entreprise EDD, consiste en une tâche de prédiction du genre d'une émission télévisée. Cette prédiction peut s'appuyer sur les historiques de genres des émissions précédentes de la même chaîne mais également sur les historiques appartenant à des chaînes parallèles. Nous proposons une taxonomie de genres adaptée à de tels traitements automatiques ainsi qu'un corpus de données contenant les historiques parallèles pour 4 chaînes françaises.Deux méthodes originales sont proposées dans ce manuscrit, permettant d'intégrer les séquences des flux parallèles. La première, à savoir, l'architecture des LSTM parallèles(PLSTM) consiste en une extension du modèle LSTM. Les PLSTM traitent simultanément chaque séquence dans une couche récurrente indépendante et somment les sorties de chacune de ces couches pour produire la sortie finale. Pour ce qui est de la seconde proposition, dénommée MSE-SVM, elle permet de tirer profit des avantages des méthodes LSTM et SVM. D'abord, des vecteurs de caractéristiques latentes sont générés indépendamment, pour chaque flux en entrée, en prenant en sortie l'événement à prédire dans le flux principal. Ces nouvelles représentations sont ensuite fusionnées et données en entrée à un algorithme SVM. Les approches PLSTM et MSE-SVM ont prouvé leur efficacité dans l'intégration des séquences parallèles en surpassant respectivement les modèles LSTM et SVM prenant uniquement en compte les séquences du flux principal. Les deux approches proposées parviennent bien à tirer profit des informations contenues dans les longues séquences. En revanche, elles ont des difficultés à traiter des séquences courtes.L'approche MSE-SVM atteint globalement de meilleures performances que celles obtenues par l'approche PLSTM. Cependant, le problème rencontré avec les séquences courtes est plus prononcé pour le cas de l'approche MSE-SVM. Nous proposons enfin d'étendre cette approche en permettant d'intégrer des informations supplémentaires sur les événements des séguences en entrée (par exemple, le jour de la semaine des émissions de l'historique). Cette extension, dénommée AMSE-SVM améliore remarquablement la performance pour les séquences courtes sans les baisser lorsque des séquences longues sont présentées.

### Stagiaires encadrés

\* Maxime Bisotto, (Avignon Université) - Licence. : Développer une plateforme web permettant aux chercheurs de l'université d'Avignon de partager leurs corpus avec leurs collègues et leurs étudiants. Le stagiaire devra participer à la conception - architecture logicielle et modèle relationnelle de la base de données -, développer avec un outil de gestion des sources et procéder à des tests - unitaires et fonctionnels (Agorantic), co-direction avec Ranger G. et Volpi E., 2019, 3 mois, *Now:* M.Sc à l'AU

 \* Rida El Allami, (Avignon Université) - Licence. : Développer une plateforme web permettant aux chercheurs de l'université d'Avignon de partager leurs corpus avec leurs collègues et leurs étudiants. Le stagiaire devra participer à la conception - architecture logicielle et modèle relationnelle de la base de données -, développer avec un outil de gestion des sources et procéder à des tests - unitaires et fonctionnels (Agorantic), co-direction avec Ranger G. et Volpi E., 2019, 3 mois, *Now:* M.Sc à l'AU \* Nour Elhouda Ayari, (Sup'com, Tunisie) - Stage d'Ingénieur : Prediction de la mobilité fondée sur l'intélligence artificielle, co-direction avec Haddad M. et Elazouzi R., 2018, 6 mois, *Maintenant* : Doctorant à l'Université de Tunis

\* **Titouan Parcollet, (U of Avignon)** - **Ingénieur de Recherche**: *Réseaux de neurones de quaternions pour la compréhension du langage parlé*, co-direction avec Lianrès G., 2017, 6 mois, *Maintenant:* Doctorant à l'Université d'Avignon

\* Afef Arfoui, (INPT, Maroc) - Stage d'Ingénieur : *Contrôle de l'évaluation de l'expérience utilisateur : mesures, métriques*, co-direction avec Haddad M. et Elazouzi R., 2018, 3 mois, *Maintenant :* Doctorant à l'Université d'Avignon

\* Titouan Parcollet, (Avignon Université) - Master: Réseaux de neurones de quaternions pour la compréhension du langage parlé, co-direction avec Lianrès G., 2016, 6 mois, Maintenant: Doctorant l'Université d'Avignon

\* Amina Brahem, (ENSI of Tunis) - Stage d'Ingénieur : Compréhension de l'activité cérébrale depuis des signaux EEG via des méthodologies issues du traitement automatique du langage, co-direction avec Lianrès G., 2014, 3 mois, *Maintenant:* Doctorant à l'Université de Tunis

\* **Etienne Papegnies, (Avignon Université)** - **Master** : Compréhension de l'activité cérébrale depuis des signaux EEG via des méthodologies issues du traitement automatique du langage, co-direction Lianrès G., 2014, 3 mois, *Maintenant:* Doctorant à l'Université d'Avignon

\* **Mathias Quillot, (Avignon Université)** - **Licence** : Utilisation d'information issue du contexte pour l'extraction de nom propre depuis un signal de parole, co-direction avec Dufour R. et Lianrès G., 2014, 2 mois, *Now:* Doctorant à l'Université d'Avignon

#### Responsabilités et affiliations scientifiques

- Responsable d'un axe de recherche au sein de la FR Agorantic de l'UAPV :
  - Contexte : Je suis responsable avec Professeur Graham Ranger de l'Axe 5 de la Fédération de Recherche (FR) Agorantic de l'UAPV appelé "Structuration et Exploitation de Corpus (SEC)". La structuration de l'information contenue dans le document consiste à extraire un ensemble fini de descripteurs de plus ou moins haut niveau ou des représentations concises et caractéristiques de ce contenu. Cette structuration permet de traiter de grands ensemble de données ou Big Data. Cette tâche de structuration se situe en amont de processus de traitement de l'information de plus haut niveau tels que la catégorisation, l'extraction d'information ou l'indexation. Il est donc nécessaire de définir des procédures de traitement de l'information permettant d'homogénéiser la gestion ainsi que la diffusion de bases de connaissances provenant de disciplines scientifiques hétérogènes. Agorantic est une structure pluridisciplinaire UAPV/CNRS, regroupant 6 laboratoires de l'UAPV autour des sociétés numériques comme objet d'étude commun aux STIC et aux SHS.
  - Implication : Animation de l'axe de recherche SEC; Organisation de séminaires et colloques autour de l'axe de recherche ainsi que, plus globalement, des événements liés à la FR Agorantic (Symposium, séminaires transversaux, etc.); Examinateurs de projets et bourses soumis (projets Agorantic, bourses doctorales Agorantic et P. Bérgé, etc.)
- Membre de l'IEEE, Institute of Electrical and Electronics Engineers
- Membre de l'ISCA, International Speech Communication Association
- Membre de l'AFCP, Association Francophone de la Communication Parlée

### Collaborations internationales, nationales et industrielles

La plupart des collaborations scientifiques mentionnées ont donné lieu à des publications et comprennent des collaborations internationales et nationales et des collaborations inter-disciplinaires (Informatique, mathématiques, géographie, etc.).

- Internationales
  - Montreal Institute for Learning Algorithms (MILA), Université de Montréal, Montréal ; Thème : Réseaux de neurones convolutionels de quaternion pour le traitement de la parole ; Cadre : Séjour de recherche de T. Parcollet; Personne concernée : Y. Zhang, C. Trabelsi, M. Ravanelli et Y. Bengio; Période : 2017 - ...
  - Microsoft Research, Cambridge (GB); Thème Représentation de documents dans des espaces thématiques multiples: ; Cadre : Séjour de recherche; Personne concernée : Y. Hamadi ; Période : 2013-2014
  - Universidad Autonoma Metropolitana (Mexique) ; Thème : Résumé automatique de documents pour accélérer l'apprentissage d'espaces latents; Cadre : projets communs; Personne concernée : J. Ramirez-Rodriguez ; Période : 2015
  - Delft Université de Technologie (Pays-Bas), Université de Southampton (Grande-Bretagne) et L3S Research Center, Université de Hanover ; Thème : Etiquetage automatique; Cadre : projets communs; Personne concernée : B. Loni (Pays-Bas) Delft University, J. Hare (GB) University of Southampton, M. Georgescu (Allemagne) L3S Hannover, M. Riegler (Norvège) Simula Research Laboratory, X. Zhu (Allemagne) L3S Hannover et M. Larson (Pays-Bas) Delft University. ; Période : 2013-2014
- Nationales
  - EURECOM, Nice; Thème : Structuration de l'information dans des espaces thématiques; Cadre
    : Projet aNR SuMACC; Personne concernée : B. Meiraldo, U. Niaz. Période : 2014
  - ➔ UMR Espace de l'UAPV ; Thème : Géolocalisation de tweets à l'aide d'espaces thématiques ; Cadre : FR Agorantic ; Personne concernée : D. Joselin; Période : 2015
  - Université du Mans ; Thème : Extraction de mentions dans des dialogues parlés bruités ; Cadre : projets communs; Personne concernée : Y. Estève, Mohamed Bouallegue, C. Lailler; Période : 2015
- Industrielles
  - Microsoft Research, Cambridge (Grande-Bretagne) Thème : Assistant PowerPoint; Cadre : Séjour de recherche; Personne concernée : Y. Hamadi ; Période : 2013-2014
  - ⊃ Orkis, (France) ; Thème : Compression de l'information dans des encoder-decoder pour le débruitage; Cadre : Thèse CIFRE; Personne concernée : K. Janod ; Période : 2013-2017
  - EDD, (France) ; Thème : Combinaison de flux pour la structuration; Cadre : Thèse CIFRE; Personne concernée : M. Bouaziz Période : 2013-2017
  - ⊃ Orkis, (France); Thème : traitement du langage par les réseaux de neurones hyper-complexes ; Cadre : Thèse CIFRE; Personne concernée : T. Parcollet; Période : 2017-2020

 Wikio eBuzzing, (France) et Syllabs (France); Thème : Moteur de recherche thématique pour la structuration de données; Cadre : Projet ANR SuMACC; Personnes concernées : B. Peralta, C. de Groc et C. de Loupy ; Période : 2011-2014

### Séjour scientifique

- Titre : The PowerPoint Assitant
- Période : Novembre 2013 à Février 2014
- Organisme d'accueil : Microsoft Research, Cambridge (GB)
- Description : L'objectif du projet d'assistant PowerPoint est de fournir un contenu pertinent aux clients de Microsoft. Ainsi, j'ai proposé des méthodes originales basées sur les espaces de thème pour permettre aux clients Microsoft d'enrichir leurs présentations PowerPoint avec du contenu pertinent. Ce projet a été sélectionné pour être présenté lors de la réunion annuelle de Microsoft TechFest'14 (40 projets ont été soumis et 22 ont été sélectionnés). Au cours de cette manifestation, des scientifiques de Microsoft Research du monde entier sont rassemblés pour exposer et discuter de leurs derniers projets, qui couvrent un large éventail de domaines liés à l'informatique. Une présentation de l'assistant Powerpoint est disponible en ligne à l'adresse http://research.microsoft.com/apps/video/default.aspx?id=211308.

### Publications

	Depuis le Dont ces 10Dont ces 5		
	début	dernières	dernières
	de carrière	années	années
Nombre de publications dans des <b>revues</b> à comité de lecture	9	9	9
Nombre de publications dans des conférences internationales	39	39	35
Nombre de conférences invitées	2	2	1
Nombre de participations à des ouvrages collectifs	1	1	1
Nombre de publications dans des conférences nationales	13	13	10
Nombre de publications dans des workshops	8	8	1

#### Journaux Internationaux

- J1 Parcollet, T., Morchid, M., Bost, X., Linarès, G., and De Mori, R. "Real to H-space Autoencoders for Theme Identification in Telephone Conversations." In IEEE/ACM Transactions on Audio, Speech, and Language Processing, 2019 [In press].
- J<sub>2</sub> Parcollet, T., **Morchid, M.**, Linarès, G., "A survey of quaternion neural networks." In *Artificial Intelligence Review, Springer*, 2019 [In press].
- J<sub>3</sub> Morchid, M., Bousquet, P.-M., Ben Kheder, W. and Janod, K.,. "Latent Topic-based Subspace for Natural Language Processing." In *Journal of Signal Processing Systems, Springer*, 2019.

- J<sub>4</sub> Morchid, M., "Parsimonious Memory Unit for Recurrent Neural Networks with Application to Natural Language Processing." In *Neurocomputing, Elsevier*, 2018.
- Janod, K., Morchid, M., Dufour, R., Linarès, G., and De Mori, R. "Denoised Bottleneck Features from Deep Autoencoders for Telephone Conversation Analysis." In IEEE/ACM Transactions on Audio, Speech, and Language Processing, 2017.
- J<sub>6</sub> Morchid, M., Dufour, R. and Linarès, G. "Impact of Word Error Rate on Theme Identification Task of Highly Imperfect Human-human Conversations." In *Computer Speech and Language*, *Elsevier*, 2016.
- J<sub>7</sub> Morchid, M., Torres-Moreno, J.-M., Dufour, R., Ramirez-Rodriguez, J. and Linarès, G. "Automatic Summarization Approaches to Speed up Topic Model Learning Process." In International Journal of Computational Linguistics and Applications, Bahari, 2016.
- J<sub>8</sub> Morchid, M., Bouallegue, M., Dufour, R., Linarès, G., Matrouf, D., and De Mori, R. "Compact Multiview Representation of Documents based on the Total Variability Space." In IEEE/ACM Transactions on Audio, Speech, and Language Processing, 2015.
- J<sub>9</sub> Morchid, M., Dufour, R., Bousquet, P.-M., Linarès, G., and Torres-Moreno, J.-M. "Feature Selection using Principal Component Analysis for Massive Retweet Detection." In *Pattern Recognition Letters, Elsevier*, 2014.

#### Chapitre de livre

B<sub>1</sub> Estève, Y., Bouallegue, M., Lailler, C., Morchid, M., Dufour, R., Linarès, G., Matrouf, D., and De Mori, R. "Integration of word and semantic features for theme identification in telephone conversations." In Natural Language Dialog Systems and Intelligent Assistants, Geunbae Lee, G., Kook Kim, H., Jeong, M., Kim, J.-H. (Eds.), Springer, 2015.

#### **Conférences Internationales**

- C<sub>1</sub> Parcollet, T., Morchid, M., Bost, X., Linarès, G. "M2H-GAN: A GAN-based Mapping from Machine to Human Transcripts for Speech Understanding." In ISCA Conference of the International Speech Communication Association (ISCA INTERSPEECH), 2019.
- C<sub>2</sub> Parcollet, T., Morchid, M., Linarès, G., De Mori, R. "Real to H-space Encoder for Speech Recognition." In ISCA Conference of the International Speech Communication Association (ISCA INTERSPEECH), 2019.
- C<sub>3</sub> Parcollet, T., Morchid, M., Linarès, G. and de Mori, R. "Bidirectional Quaternion Long Short-Term Memory Recurrent Neural Networks for Speech Recognition." In IEEE International Conference on Acoustic, Speech and Signal Processing (IEEE ICASSP), 2019.
- C<sub>4</sub> Parcollet, T., Morchid, M. and Linarès, G. "Quaternion Convolutional Neural Networks for Heterogeneous Image Processing." In IEEE International Conference on Acoustic, Speech and Signal Processing (IEEE ICASSP), 2019.

- C<sub>5</sub> Parcollet, T., Ravanelli, M., Morchid, M., Linarès, G., Trabelsi, C., de Mori, R and Bengio, Y. "Quaternion Recurrent Neural Networks." In *International Conference on Learning Representations* (ICLR), 2019.
- C<sub>6</sub> Parcollet, T., Ravanelli, M., **Morchid, M.**, Linarès, G. and de Mori, R. "Speech Recognition with Quaternion Neural Networks." In *IRASL@NIPS*, 2018.
- C<sub>7</sub> Parcollet, T., Morchid, M., Linarès, G. and de Mori, R. "Quaternion Convolutional Neural Networks for Theme Identification of Telephone Conversations." In IEEE International Spoken Language Technology Workshop (IEEE SLT), 2018.
- C<sub>8</sub> Parcollet, T., Zhang, Y., Trabelsi, C., Morchid, M., de Mori, R., Linarès, G. and Bengio, Y.
  "Quaternion Convolutional Neural Networks for End-to-End Automatic Speech Recognition." In ISCA Conference of the International Speech Communication Association (ISCA INTERSPEECH), 2018.
- C<sub>9</sub> Parcollet, T., Morchid, M. and Linarès, G. "Deep Quaternion Neural Networks for Spoken Language Understanding." In IEEE International Automatic Speech Recognition and Understanding Workshop (IEEE ASRU), 2017.
- C<sub>10</sub> Morchid, M. "Internal Memory Gate for Recurrent Neural Networks with Application to Spoken Language Understanding." In ISCA Conference of the International Speech Communication Association (ISCA INTERSPEECH), 2017.
- C<sub>11</sub> Parcollet, T., Morchid, M. and Linarès, G. "Quaternion Denoising Encoder-Decoder for Theme Identification of Telephone Conversations." In ISCA Conference of the International Speech Communication Association (ISCA INTERSPEECH), 2017.
- C<sub>12</sub> Parcollet, T., Morchid, M., Bousquet, P.-M., Dufour, R., Linarès, G., and De Mori, R. "Quaternion Neural Networks for Spoken Language Understanding." In IEEE International Spoken Language Technology Workshop (IEEE SLT), 2016.
- C<sub>13</sub> Bouaziz, M., Morchid, M., Dufour, R., Linarès, G., and De Mori, R. "Parallel Long Short-Term Memory for Multi-Stream Classification." In IEEE International Spoken Language Technology Workshop (IEEE SLT), 2016.
- C<sub>14</sub> Janod, K., Morchid, M., Dufour, R., Linarès, G. "A Log-Linear Weighting Approach in the Word2vec Space for Spoken Language Understanding." In IEEE International Spoken Language Technology Workshop (IEEE SLT), 2016.
- C<sub>15</sub> Bouaziz, M., Morchid, M., , Dufour, R., Linarès, G. "Improving Multi-Stream Classification by Mapping Sequence-Embedding in a High Dimensional Space." In IEEE International Spoken Language Technology Workshop (IEEE SLT), 2016.
- C<sub>16</sub> Dufour, R., Morchid, M., Parcollet, T. "Tracking Dialog States using an Author-Topic based Representation." In IEEE International Spoken Language Technology Workshop (IEEE SLT), 2016.
- C<sub>17</sub> Morchid, M., Bouaziz, M., Ben Kheder, W., Janod, K., Bousquet, P.-M., Dufour, R., and Linarès, G. "Spoken Language Understanding in a Latent Topic-based Subspace." In ISCA Conference of the International Speech Communication Association (ISCA INTERSPEECH), 2016.

- C<sub>18</sub> Janod, K., Morchid, M., Dufour, R., Linarès, G. and De Mori, R. "Deep Stacked Autoencoders for Spoken Language Understanding." In ISCA Conference of the International Speech Communication Association (ISCA INTERSPEECH), 2016.
- C<sub>19</sub> Morchid, M., Dufour, R., and Linarès, G. "Topic-Space based Setup of a Neural Network for Theme Identification of Highly Imperfect Transcriptions." In IEEE International Automatic Speech Recognition and Understanding Workshop (IEEE ASRU), 2015.
- C<sub>20</sub> Morchid, M., Josselin, D., Dufour, R., Altman, E., and Linarès, G. A Topic Modeling based Representation to Detect Tweet Locations. " In *the International Society for Photogrammetry and Remote Sensing (ISPRS Geospatial week)*, 2015.
- C<sub>21</sub> Morchid, M., Dufour, R., and Matrouf, D. "A comparison of normalization techniques applied to latent space representations for speech analytics." In ISCA Conference of the International Speech Communication Association (ISCA INTERSPEECH), 2015.
- C<sub>22</sub> Morchid, M., Portilla, Y., Josselin, D., Dufour, R., Altman, E., El-Beze, M., Cossu, J.-V., Linarès, G., and Reiffers-Masson, A. "An author topic-based approach to cluster tweets and mine their location." In *International Conference on Spatial Statistics: Emerging Patterns (SPATIAL)*, 2015.
- C<sub>23</sub> Morchid, M., Dufour, R., Linarès, G., and Hamadi, Y. "Latent topic model based representations for a robust theme identification of highly imperfect automatic transcriptions." In International Conference on Intelligent Text Processing and Computational Linguistics (CICLing), 2015.
- C<sub>24</sub> Morchid, M., Dufour, R., Bouallegue, M., and Linarès, G. "Author-topic based representation of call-center conversations." In IEEE International Spoken Language Technology Workshop (IEE SLT), 2014.
- C<sub>25</sub> Loni, B., Hare, J., Georgescu, M., Riegler, M., Zhu, X., Morchid, M., Dufour, R., and Larson, M. "Getting by with a little help from the crowd: Practical approaches to social image labeling." In ACM International Workshop on Crowdsourcing for Multimedia (ACM CrowdMM), 2014.
- C<sub>26</sub> Morchid, M., Bouallegue, M., Dufour, R., Linarès, G., Matrouf, D., and De Mori, R. "An i-vector based approach to compact multi-granularity topic spaces representation of textual documents." In the Conference of Empirical Methods on Natural Language Processing (EMNLP), 2014.
- C<sub>27</sub> Morchid, M., Dufour, R., and Linarès, G. "A combined thematic and acoustic approach for a music recommendation service for tv commercials." In *International Society for Music Information Retrieval Conference (ISMIR)*, 2014.
- C<sub>28</sub> Morchid, M., Dufour, R., Bouallegue, M., Linarès, G., and De Mori, R. "Theme identification in human-human conversations with features from specific speaker type hidden spaces." In ISCA Conference of the International Speech Communication Association (ISCA INTERSPEECH), 2014.
- C<sub>29</sub> Morchid, M., Bouallegue, M., Dufour, R., Linarès, G., Matrouf, D., and De Mori, R. "I-vector based representation of highly imperfect automatic transcriptions." In ISCA Conference of the International Speech Communication Association (ISCA INTERSPEECH), 2014.

- C<sub>30</sub> Bouallegue, M., Morchid, M., Dufour, R., Driss, M., Linarès, G., and De Mori, R. "Subspace gaussian mixture models for dialogues classification." In ISCA *Conference of the International Speech Communication Association (ISCA INTERSPEECH)*, 2014.
- C<sub>31</sub> Bouallegue, M., Morchid, M., Dufour, R., Driss, M., Linarès, G., and De Mori, R. "Factor analysis based semantic variability compensation for automatic conversation representation." In ISCA Conference of the International Speech Communication Association (ISCA INTERSPEECH), 2014.
- C<sub>32</sub> Morchid, M., Dufour, R., Niaz, U., Bouvier, F., de Groc, C., Linarès, G., Meiraldo, B., and Peralta, B. "Sumacc project's corpus: A topic-based query extention approach to retrieve multimedia documents." In ISCA *Proceedings of the 17th International Conference on Text, Speech and Dialogue (ISCA TSD)*, 2014.
- C<sub>33</sub> Morchid, M., Dufour, R., and Linarès, G. "A Ida-based topic classification approach from highly imperfect automatic transcriptions." In *Proceedings of the 9th edition of the Language Resources* and Evaluation Conference (LREC), 2014.
- $C_{34}$  Morchid, M., Linarès, G., and Dufour, R. "Characterizing and predicting bursty events: The buzz case study on twitter." In *Proceedings of the 9th edition of the Language Resources and Evaluation Conference (LREC)*, 2014.
- C<sub>35</sub> Morchid, M., Dufour, R., Bousquet, P.-M., Bouallegue, M., Linarès, G., and De Mori, R. "Improving dialogue classification using a topic space representation and a gaussian classifier based on the decision rule." In IEEE International Conference on Acoustic, Speech and Signal Processing (IEEE ICASSP), 2014.
- C<sub>36</sub> Morchid, M., Dufour, R., and Linarès, G. "Thematic representation of short text messages with latent topics: Application in the twitter context." In *Conference of the Pacific Association for Computational Linguistics (PACLING)*, 2013.
- C<sub>37</sub> Morchid, M., Linarès, G., El-Beze, M., and De Mori, R. "Theme identification in telephone service conversations using quaternions of speech features." In ISCA Conference of the International Speech Communication Association (ISCA INTERSPEECH), 2013.
- C<sub>38</sub> Morchid, M. and Linarès, G. "Lda-based method for automatic tagging of youtube videos." In IEEE International Workshop on Image and Audio Analysis for Multimedia Interactive Services (IEEE WIAMIS), 2013.
- C<sub>39</sub> Morchid, M., Dufour, R., and Linarès, G. "Event detection from image hosting services by slightly-supervised multi-span context models." In IEEE International Workshop on Content-Based Multimedia Indexing (IEEE CBMI), 2013.

#### **Conférences nationales**

 $N_1$  Parcollet, T., **Morchid, M.**, Linarès, G. "Réseaux de neurones convolutifs de quaternions pour l'identification de thèmes de conversations téléphoniques." In *COnférence en Recherche d'Information et Applications (CORIA)*, 2019.

- N<sub>2</sub> Parcollet, T., **Morchid, M.**, Linarès, G. "Réseaux de neurones de quaternions pour le traitement du langage." In *COnférence en Recherche d'Information et Applications (CORIA)*, 2017.
- N<sub>3</sub> Bouaziz, M., Morchid, M., Bousquet, P.-M., Dufour, R., Janod, K., Ben Kheder, W., and Linarès, G. "Un Sous-espace Thématique Latent pour la Compréhension du Langage Parlé." In Proceedings Journées d'étude de la parole (JEP), 2016.
- N<sub>4</sub> Janod, K., **Morchid, M.**; Dufour, R., Linarès, G. and De Mori, R. "Auto-encodeurs pour la compréhension de documents parlés." In *Proceedings Journées d'étude de la parole (JEP)*, 2016.
- N<sub>5</sub> Bouaziz, M., Morchid, M., Dufour, R., Linarès, G. and Correa, P. "Un Corpus de Flux TV Annotés pour la Prédiction de Genres ." In *Proceedings Journées d'étude de la parole (JEP)*, 2016.
- N<sub>6</sub> Janod, K., Morchid, M.; Dufour, R., Linarès, G. "Réseaux de Neurones pour la Représentation de Contextes Continus des Mots." In *COnférence en Recherche d'Information et Applications* (CORIA), 2016.
- N<sub>7</sub> Morchid, M., Dufour, R., and Linarès, G. "Initialisation de Réseaux de Neurones à l'aide d'un Espace Thématique." In *Conférence sur le Traitement Automatique des Langues Naturelles* (*TALN*), 2015.
- N<sub>8</sub> Janod, K., Morchid, M., Dufour, R., and Linarès, G. "Apport de l'information temporelle des contextes pour la représentation vectorielle continue des mots." In *Conférence sur le Traitement Automatique des Langues Naturelles (TALN)*, 2015.
- N<sub>9</sub> Lailler, C., Estève, Y., De Mori, R., Bouallegue, M., and Morchid, M. "Utilisation d'annotations sémantiques pour la validation automatique d'hypothèses dans des conversations téléphoniques." In Conférence sur le Traitement Automatique des Langues Naturelles (TALN), 2015.
- $N_{10}$  Morchid, M., Dufour, R., Linarès, G., and De Mori, R. "Classification de transcriptions automatiques imparfaites : Doit-on adapter le calcul du taux d'erreur-mot ?" In *Proceedings Journées d'étude de la parole (JEP)*, 2014.
- N<sub>11</sub> **Morchid, M.** and Linarès, G. "Prédiction des buzz sur twitter." In *COnférence en Recherche d'Information et Applications (CORIA)*, 2013.
- N<sub>12</sub> Morchid, M., Dufour, R., and Linarès, G. "Combinaison de thèmes latents pour la contextualisation de tweets." In Atelier Contextualisation de Messages Courts (EGC), 2013.
- $N_{13}$  **Morchid, M.** and Linarès, G. "Extraction de mots-clefs dans des vidéos web par analyse latente de Dirichlet." In *Proceedings Journées d'étude de la parole (JEP)*, 2012.

#### Ateliers internationaux (sans comité de relecture)

W<sub>1</sub> Morchid, M., Huet, S., and Richard, D. "A topic-based approach for post-processing correction of automatic translations." In *International Workshop on Spoken Language Translation (IWSLT)*, 2014.

- W<sub>2</sub> Morchid, M., Dufour, R., Bouallegue, M., Linarès, G., and Matrouf, D. "Lia@Mediaeval 2013 Crowdsourcing task: Metadata or not metadata? that is a fashion question." In (MediaEval), 2013.
- W<sub>3</sub> Morchid, M., Dufour, R., Bouallegue, M., Linarès, G., and Matrouf, D. "Lia @ mediaeval 2013 musiclef task: A combined thematic and acoustic approach." In (*MediaEval*), 2013.
- W<sub>4</sub> Bouallegue, M., Senay, G., Morchid, M., Matrouf, D., Linarès, G., and Dufour, R. "Lia @ mediaeval 2013 spoken web search task: An i-vector based approach." In (*MediaEval*), 2013.
- W<sub>5</sub> Cossu, J.-V., Bigot, B., Bonnefoy, L., Morchid, M., Bost, X., Senay, G., Dufour, R., Bouvier, V., Torres-Moreno, J.-M., and El-Bèze, M. "Lia@replab 2013." In An evaluation campaign for Online Reputation Management Systems (CLEF), 2013.
- W<sub>6</sub> Bost, X., Brunetti, I., Cabrera-Diego, L. A., Cossu, J.-V., Linhares, A., Morchid, M., Torres-Moreno, J.-M., El-Bèze, M., and Richard, D. "Systèmes du lia à deft'13." In Actes du neuvième DÉfi Fouille de Textes (DEFT), 2013.
- W<sub>7</sub> Morchid, M. and Linarès, G. "Inex 2012 benchmark a semantic space for tweets contextualization." In Workshop of the Initiative for the Evaluation of XML Retrieval (CLEF), 2012.
- W<sub>8</sub> Morchid, M. and Linarès, G. "Mediaeval benchmark: Social event detection using Ida and external resources." In (*MediaEval*), 2011.

#### **Conférences invités**

- P1 Bouaziz, M., Morchid, M., Richard, D. and Lianrès G. "Structuration des Flux TV : Etat de l'Art." In Journée commune AFIA - ARIA 2015 : Intelligence artificielle et Recherche d'Information, 2015.
- P<sub>2</sub> Morchid, M. and Linarès, G. "Bursty event prediction by content-based approach." In *Third Workshop of the Brazilian Institute for Web Science Research*, 2012.

## Activités d'enseignement

Le volume horaire annuel pour un ensiegnant-chercheur est de plus de 192 heures. Les matières enseignées tournent autour de la programmation, de la conception ainsi que leurs applications. Je suis également responsable pédagogique du projet de Master 1 qui couvre les deux semestres. Le détail des enseignements donnés ainsi que des responsabilités pédagogiques sont disponibles **ici**.

.....

### Matières enseignées.

- Cours en Master "Ingénierie du Logiciel de la SociétE Numérique" à l'Université d'Avignon
  - ⊃ Projet Master
  - ⊃ Ingénierie Logicielle (Orienté recherche Traitement automatique du langage)
  - ⊃ Infrastructure Logicielle
  - ➔ Algorithme Avancée
  - ➔ Architectures Distribuées
  - ➔ Conception Orientée Objet
  - ⊃ Défis pour la société numérique (Orienté recherche Réseaux de Neurones (IA))
- Cours en Licence Informatique à l'Université d'Avignon
  - ➔ Modélisation Objet Java
  - ⊃ UML
  - ➔ Projet Réseaux
  - ⊃ Interface Homme-Machine
  - ⊃ Initiation à l'outil numérique
  - ⊃ Interface Graphique et Ergonomie

### Responsabilités pédagogiques

Je suis responsable de différents cours en informatique ainsi que des projets de Master 1 couvrant les deux semestres. Les cours dont j'ai la charge et la responsabilité, suivent un parcours éducatif naturel entre les différentes années formant le cursus de la Licence au Master en Informatique, comme par exemple, l'enseignement en Java entre la Licence 2 et 3.

- Responsable pédagogique des projets de Master Master
  - Les étudiants ont pour objectif de mener à bien un projet informatique de la conception via un cahier des charges lors ud premier semestre, à la réalisation à proprement dit au second semestre. Ainsi, les étudiants doivent respecter les bonnes pratiques liées à la conduite de projet (organisation du travail, suivi, réalisation de comptes rendus, organisation de réunions de travail, etc.). J'ai donc en charge d'assurer la bonne tenue de ces projets, l'organisation des évaluations des cahiers des charges et des présentations, l'envoi des comptes rendus en temps et en heure, l'enregistrement des sujets et des groupes, la médiation entre les étudiants et les tuteurs, etc.
- Conception et Modélisation Orientée Objet Master
  - Cet enseignement a pour objectif de fournir aux étudiants la maîtrise de la conception du logiciel à l'aide d'outils et concepts liés à la programmation orientée objet et du langage de programmation Java.
- Architectures Distribuées Master
  - Différentes notions liées aux architectures distribuées sont étudiées : Middlewrae (Ice), Services Web (SOAP, WSDL, ...) ainsi que différentes technologies : EJB, Applet/Servlet Java. Le projet met en exergue différents langages de programmation et met l'accent également sur leur interpolarité dans une application portable android.
- Interface Graphique et Ergonomie Licence
  - ⊃ Le contenu de ce cours de Licence 3 est à la lisière de l'enseignement des bonnes pratiques en termes d'ergonomie et d'"utilisabilité" des interface graphique (anciennement appelé en Licence 3 "Interface Homme-Machine"), et de la programmation à proprement dite de ces applications à travers le langage JavaFx. Il suit naturellement les enseignements de Java prodigués en Licence 2 (Modélisation objet et Java Licence).
- Modélisation objet et Java Licence
  - Ce cours a pour objectif d'initier les étudiants à la programmation orientée objet. Pour ce faire, j'ai réalisé l'ensemble des cours et des TP permettant aux étudiants d'acquérir les bonnes pratiques de la programmation objet via le langage de programmation Java.
- Interface Homme-Machine (IHM) Licence
  - ⊃ L'objectif de ce cours est de revoir tous les éléments de base de l'orienté objets, leur traduction en C# ainsi que les concepts spécifiques au langage. L'accent est mis particulièrement sur les délégués et les événements, afin d'introduire leur utilisation dans les formulaires. Enfin, la réalisation d'Interfaces Homme-Machine sera proposée en utilisant directement les classes du Framework .NET, et mise en oeuvre sur des travaux pratiques.
- Projet réseaux Licence
  - ⊃ Ce nouveau cours se trouve à l'interface entre le génie logiciel et les applications réseaux. L'objectif est de réaliser une interface Web semblable à celles des fournisseurs d'accès (Box) en PHP pour la gestion de scripts de configuration réseaux (DHCP, DNS, WebMail, Filtrage...).